CREDIT AGREEMENT

THIS AGREEMENT is made as of January 26, 2024

BETWEEN

HBC YSS1 LIMITED PARTNERSHIP (the "Borrower"),

- and -

HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI

(as "Nominee")

- and -

RIOCAN REAL ESTATE INVESTMENT TRUST (as "Guarantor"),

- and -

THE FINANCIAL INSTITUTIONS from time to time parties to this Agreement and designated as Lenders on the signature pages hereto (each, a "Lender" and collectively, the "Lenders")

-and-

ROYAL BANK OF CANADA

(the "Administrative Agent")

- and -

RBC CAPITAL MARKETS

(as "Lead Arranger" and "Sole Bookrunner")

WHEREAS the Borrower has requested the Credit Facility and the Lenders have agreed to provide the Credit Facility to the Borrower upon and subject to the terms and conditions set out in this Agreement;

AND WHEREAS Royal Bank of Canada will be the Administrative Agent as contemplated by Section 19.01 hereof;

AND WHEREAS it is a condition of providing the Credit Facility that the Guarantor guarantees the Obligations of the Borrower on the terms and conditions herein set forth;

NOW THEREFORE THIS AGREEMENT WITNESSES that in consideration of the covenants and agreements herein contained the parties hereto agree as follows:

ARTICLE 1 - INTERPRETATION

1.01 Definitions

In this Agreement, unless something in the subject matter or context is inconsistent therewith:

- "Additional Compensation" has the meaning set out in Section 13.01(3).
- "Adjusted Daily Compounded CORRA" means, for purposes of any calculation, the rate per annum equal to (a) Daily Compounded CORRA for such calculation plus (b) the Daily Compounded CORRA Adjustment; provided that if Adjusted Daily Compounded CORRA as so determined shall be less than the Floor, then Adjusted Daily Compounded CORRA shall be deemed to be the Floor.
- "Adjusted Term CORRA" means, for purposes of any calculation, the rate per annum equal to (a) Term CORRA for such calculation plus (b) the Term CORRA Adjustment; provided that if Adjusted Term CORRA as so determined shall ever be less than the Floor, then Adjusted Term CORRA shall be deemed to be the Floor.
- "Adjusted Term SOFR" means, for purposes of any calculation, the rate per annum equal to (a) Term SOFR for such calculation plus (b) the Term SOFR Adjustment; provided that, if the Adjusted Term SOFR as so determined shall ever be less than the Floor, then the Adjusted Term SOFR shall be deemed to be the Floor.
- "Administrative Questionnaire" means an administrative questionnaire in a form provided by the Administrative Agent.
- "Advance" means the advance of funds made hereunder to the Borrower by the Lenders by way of a Loan.
- "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
- "Agreement" means this credit agreement, including its recitals and schedules.
- "AMLA" means the *Proceeds of Crime (Money Laundering) and Terrorist Financing Act (Canada)* and its associated regulations in effect as of the date hereof.
- "Applicable Law" means (a) any domestic or foreign statute, law (including common and civil law), treaty, code, ordinance, rule, regulation, restriction or by-law (zoning or otherwise), (b) any judgement, order, writ, injunction, decision, ruling, decree or award, (c) any regulatory policy, practice, guideline or directive, or (d) any franchise, licence, qualification, authorization, consent, exemption, waiver, right, permit or other approval of any Governmental Authority, in each case having the force of law or otherwise being binding on or affecting the Person referred to in the context in which the term is used or binding on or affecting the property of such Person, provided that, notwithstanding the foregoing, "Applicable Law" shall include, without limitation,

published Canada Revenue Agency administrative positions and any rules, conventions or agreements with, or published by, Governmental Authorities or regulatory agencies to which a Lender is bound by, or subject to, as applicable."

"Applicable Margin" means, subject to the following sentences of this definition, the margins in basis points per annum set forth in and defined in Schedule D determined based on the then existing External Credit Rating. In respect of (i) Prime Rate Loans, the Applicable Margin shall be the applicable margin referred to in the column titled "Prime Rate Margin (bps)" in Schedule D; and (ii) CORRA Loans and Term SOFR Loans, the Applicable Margin shall be the applicable margin referred to in the column titled "CORRA / Term SOFR Margin (bps)" in Schedule D. Except as provided in Schedule D, the Applicable Margin shall be adjusted as of the date of any change in the External Credit Rating. For greater certainty, at the time of a change in an Applicable Margin applicable to a Term CORRA Loan or Term SOFR Loan (the "Margin Adjustment Date"), such adjusted Applicable Margin shall be applied from and after such date with respect to Term CORRA Loans or Term SOFR Loans, as applicable, made on or after the Margin Adjustment Date and shall not modify or otherwise affect the Applicable Margin applicable with respect to any outstanding Term CORRA Loan or Term SOFR Loan, as applicable, on such date.

"Appraisal" means, with respect to the Property, a "market value" appraisal of the Property prepared in accordance with the requirements for a Complete, Self-Contained Appraisal Report as defined by the Uniform Standards of Professional Appraisal Practice adopted by the Appraisal Institute and conducted in accordance with the standards of the Appraisal Institute by an Appraiser, provided that the appraisals for the Property provided to the Administrative Agent as of the date hereof are deemed to constitute an Appraisal.

"Appraisal Institute" means the Appraisal Institute of Canada.

"Appraiser" means a professional real estate appraiser who is a member in good standing of the Appraisal Institute and who has a minimum of five years' experience in the subject property type and market.

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (x) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (y) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark pursuant to this Agreement, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 2.11(4).

"Benchmark" means, initially, the Term CORRA Reference Rate or Daily Compounded CORRA, as the case may be; provided that if a Benchmark Transition Event has occurred with respect to the Term CORRA Reference Rate, Daily Compounded CORRA, or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that

such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 2.11(1).

"Benchmark Replacement" means, with respect to any Benchmark Transition Event,

- (a) where a Benchmark Transition Event has occurred with respect to Term CORRA Reference Rate, Daily Compounded CORRA; and
- (b) where a Benchmark Transition Event has occurred with respect to a Benchmark other than the Term CORRA Reference Rate, the sum of: (i) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower giving due consideration to (A) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (B) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Canadian Dollar-denominated syndicated credit facilities and (ii) the related Benchmark Replacement Adjustment.

If the Benchmark Replacement as determined pursuant to clause (a) or (b) above would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the thencurrent Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Canadian Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means a date and time determined by the Administrative Agent, which date shall be no later than the earliest to occur of the following events with respect to the then-current Benchmark:

- in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory

supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; provided that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Bank of Canada, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 2.11, and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 2.11.

"Borrower" means HBC YSS1 Limited Partnership (the general partner of which is Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI) and its permitted successors and assigns.

"Business Day" means (a) other than in connection with Term SOFR Loans, a day which is not (i) a Saturday or a Sunday, or (ii) a day observed as a holiday under the laws of the Province of Ontario or the federal laws of Canada applicable therein, and (b) when used in connection with Term SOFR Loans, a U.S. Government Securities Business Day.

"Canadian Dollars" and "Cdn. \$" mean the lawful money of Canada.

"Capital Lease" means a capital lease or a lease that should be treated as a capital lease under GAAP.

"Change in Applicable Laws" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any Applicable Law, (b) any change in any Applicable Law or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any Applicable Law by any Governmental Authority; provided that notwithstanding anything to the contrary, all requests, rules, guidelines, or directives promulgated by the Bank of International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Applicable Law", regardless of the date enacted, adopted or issued.

"Change of Control" means any course of action undertaken by any Person which is intended to result in, or would result (in the reasonable opinion of the Majority of the Lenders) in, a Change in Control, or the transfer of all or substantially all of the undertaking and assets of any Obligor. For the purposes hereof, "Change in Control" shall mean a change of Control of any of the Obligors.

For greater certainty, changes in the individuals comprising the board of trustees of the Guarantor or the board of directors of the Nominee shall not constitute a Change of Control unless there is a change in the majority of the board of trustees or the board of directors, as applicable, as the same exists on the date hereof (and excluding any change resulting from the death or incapacity of any trustee or the resignation or removal of any trustee or director, as applicable, for reason unrelated to an acquisition of Equity Interests in the Guarantor or the Nominee, as applicable).

"City" means City of Toronto, Ontario.

"Claims" means, in respect of any event, circumstance, matter or thing, all actions, proceedings, losses, damages, liabilities, taxes, claims, demands, judgments, rights (including set-off), remedies, costs and expenses of any nature or kind, including legal fees and disbursements on a full indemnity basis, and "Claim" means any one of them.

"Closing Date" means January 26, 2024.

"Collateral" means, collectively, the Property, all real and personal property (and the revenues, insurance proceeds (save as herein expressly excluded), issues, profits, proceeds and products therefrom) which are subject, or are intended or required to become subject, to the security or Encumbrance granted under any of the Loan Documents.

"Commitment" means, in respect of any Lender, the commitment by each Lender to provide the amount of Canadian Dollars set opposite its name in Schedule A annexed hereto (as such Schedule A may be amended from time to time pursuant to a provision hereof by which Commitments are modified, and a copy of which is delivered to the Borrower by the Administrative Agent), being the sum of the maximum aggregate amount of the Loans that such Lender is obliged to make, as such amount may be reduced from time to time by such Lender's Rateable Portion of the amount of any permanent repayments, reductions or prepayments required or made hereunder.

"Compliance Certificate" means the certificate required pursuant to Section 9.02(3), substantially in the form attached as Schedule 1.01(A), signed by any senior officer of the Guarantor.

"Conforming Changes" means, with respect to the use or administration of a Benchmark or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Prime Rate," the definition of "Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of breakage provisions and other technical, administrative or operational matters) that the Administrative Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Consolidated EBITDA" of the Guarantor for any period means Consolidated Net Income for such period increased by the sum of, without duplication (i) Consolidated Interest Expense for such period, (ii) depreciation and amortization expense for such period, and (iii) Consolidated Income Tax Expense for such period (other than income taxes, either positive or negative, attributable to extraordinary or non-recurring gains or losses or other non-cash gains or losses as

adjusted for in calculating Consolidated Net Income), determined on a consolidated basis in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Consolidated Gross Book Value" means, at any time, the book value of the assets of the Guarantor, as shown on its then most recent consolidated balance sheet, excluding goodwill and future income tax assets, determined on a consolidated basis in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Consolidated Income Tax Expense" of the Guarantor, for any period means the income tax expense of the Guarantor for such period, determined on a consolidated basis in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Consolidated Indebtedness" of the Guarantor, as at any date means the consolidated Indebtedness of the Guarantor, as at such date determined in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Consolidated Interest Expense" means, with respect to the Guarantor, for any period, the aggregate amount of interest expense of the Guarantor, adjusted in all cases for Proportionate Consolidation Adjustments in respect of Consolidated Indebtedness, and to the extent interest has been capitalized on projects that are under development or held for future development, the amount of interest so capitalized (including Proportionate Consolidation Adjustments), all as determined on a consolidated basis in accordance with GAAP.

"Consolidated Net Income" of the Guarantor, for any period means the net income (loss) of the Guarantor for such period determined on a consolidated basis in accordance with GAAP, excluding (i) any gain or loss attributable to the sale or other disposition of any asset or liability of the Guarantor and related transaction costs, (ii) any non-cash changes in fair value and other non-cash gains or losses of the Guarantor, determined on a consolidated basis in accordance with GAAP, (iii) other non-recurring items, and (iv) including Proportionate Consolidation Adjustments and including or excluding, as applicable, the related tax impact of items (i) to (iii).

"Consolidated Secured Indebtedness" of the Guarantor as at any date means the sum of Consolidated Indebtedness that is secured by any Encumbrance as at such date determined in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Consolidated Unsecured Indebtedness" of the Guarantor as at any date means the sum of Consolidated Indebtedness (excluding Subordinated Indebtedness) that is not secured in any manner by any Encumbrance as at such date determined in accordance with GAAP and including Proportionate Consolidation Adjustments.

"Contingent Obligation" means, with respect to any Person, any obligation, whether secured or unsecured, of such Person guaranteeing or indemnifying, or has the practical effect of guaranteeing or indemnifying, any indebtedness, leases, dividends, letters of credit or other monetary obligations (the "primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including any obligation of such Person as an account party in respect of a letter of credit or letter of guarantee issued to assure payment by the primary obligor of any such primary obligation and any obligations of such Person, whether or not

contingent, (i) to purchase any such primary obligation or any Property constituting direct or indirect security therefor, (ii) to advance or supply funds for the purchase or payment of any such primary obligation or to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase Property, securities or services primarily for the purpose of assuring the obligee under any such primary obligation of the ability of the primary obligor to make payment of such primary obligation, or (iv) otherwise to assure or hold harmless the obligee under such primary obligation against loss in respect of such primary obligation; provided, however, that the term "Contingent Obligation" does not include endorsements of instruments for deposit or collection in the ordinary course of business.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have corresponding meanings.

"Conversion" means a conversion of a Loan pursuant to Section 2.04.

"Conversion Date" means the Business Day being the date on which the Borrower is deemed to have elected to convert one type of Loan into another type of Loan.

"Conversion Notice" means a notice, substantially in the form set out in Schedule 1.01(C), to be given to the Administrative Agent by the Borrower pursuant to Section 2.04.

"CORRA" means the Canadian Overnight Repo Rate Average administered and published by the Bank of Canada (or any successor administrator).

"CORRA Loans" means Term CORRA Loans and Daily Compounded CORRA Loans.

"CORRA Margin" means, for any period, the Applicable Margin with respect to CORRA Loans.

"Credit Facility" means the non-revolving secured term senior facility, in the maximum aggregate principal amount equal to the Credit Facility Commitment as reduced or terminated in accordance with the terms hereof.

"Credit Facility Commitment" means the aggregate amount of all Commitments of the Lenders under the Credit Facility as set forth in Schedule A, as reduced or terminated in accordance with the provisions of this Agreement.

"Daily Compounded CORRA" means, for any day (a "Daily Compounded CORRA Rate Day"), a rate per annum equal to CORRA for the day (such day, the "Daily Compounded CORRA Determination Day"), that is five (5) Business Days prior to (i) if such Daily Compounded CORRA Rate Day is a Business Day, such Daily Compounded CORRA Rate Day or (ii) if such Daily Compounded CORRA Rate Day is not a Business Day, the Business Day immediately preceding such Daily Compounded CORRA Rate Day, in each case, as CORRA is published by the administrator; provided, however, that if as of 5:00 p.m. (Toronto time) on any Daily Compounded CORRA Determination Day, CORRA for the applicable tenor has not been

published by the administrator and a Benchmark Replacement Date with respect to Daily Compounded CORRA has not occurred, then Daily Compounded CORRA will be CORRA as published by the administrator on the first preceding Business Day for which CORRA was published by the administrator so long as such first preceding Business Day is not more than three (3) Business Days prior to such Daily Compounded CORRA Determination Day; provided, that to the extent such rate as determined above shall, at any time, be less than the Floor, such rate shall be deemed to be Floor for all purposes herein.

- "Daily Compounded CORRA Adjustment" means, with respect to Daily Compounded CORRA, a percentage equal to 0.29547% per annum (29.547 basis points).
- "Daily Compounded CORRA Determination Day" has the meaning set forth in the definition of "Daily Compounded CORRA".
- "Daily Compounded CORRA Loan" means a Loan made by the Lenders to the Borrower that bears interest at a rate based on Adjusted Daily Compounded CORRA.
- "Daily Compounded CORRA Rate Day" has the meaning set forth in the definition of "Daily Compounded CORRA".
- "DBRS" means Dominion Bond Rating Services.
- "Debenture" means the demand debenture in the principal amount of Cdn. \$100,000,000 to be granted by the Nominee in favour of the Administrative Agent constituting a first charge on the Property, which charge shall include the provisions of a general security agreement granting the Administrative Agent a first-priority security interest over all present and future personal property of the Nominee related to the Property, and which shall contain an assignment of property insurance proceeds.
- "**Debt Service**" means for any period, the sum of (without duplication) (i) Consolidated Interest Expense for such period and (ii) all regularly scheduled principal payments made with respect to Consolidated Indebtedness during such period (other than any balloon, bullet or similar principal payable at maturity or which repays such Indebtedness in full).
- "Debtor Relief Law" means any of the Companies' Creditors Arrangement Act (Canada), the Bankruptcy and Insolvency Act (Canada), the Winding-up and Restructuring Act (Canada), and any similar federal, provincial, state or foreign law for or in respect of the relief of debtors, conservatorship, bankruptcy, general assignment for the benefit of creditors, moratorium, arrangement, receivership, insolvency, reorganization or similar laws of Canada or other applicable jurisdictions from time to time in effect and any similar federal, provincial, state or foreign law for the relief of debtors affecting the rights of creditors generally.
- "Default" means any event or condition, the occurrence of which would, with the lapse of time or giving of notice, or both, becomes an Event of Default.
- "Defaulting Lender" means, subject to Section 13.04(1)(d) any Lender that (a) has failed to (i) fund all or any portion of its Loans within two Business Days of the date such Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the

Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent or any Lender any other amount required to be paid by it hereunder within two Business Days of the date when due, (b) has notified the Borrower or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within Canada, the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 13.04(1)(d)) upon delivery of written notice of such determination to the Borrower and each Lender.

"Direction and Acknowledgement" means the direction by the Borrower, as beneficial owner (on a limited recourse basis with recourse being limited to the Collateral subject to the provisions of Section 1.08(3)), to the Nominee, as nominee, to execute this Agreement, the Security and any other Loan Documents to which it is a party and an agreement that such Loan Documents shall bind the Borrower to the same extent as if the Borrower had executed them in the place of the Nominee.

"discretion" means in the sole, absolute, subjective and unreviewable discretion of the relevant Person.

"Disposition" means, with respect to a Person, any sale, assignment, transfer, conveyance, lease, licence or other disposition of any nature or kind whatsoever of any property or of any right, title or interest in or to any property, and the verb "Dispose" has a corresponding meaning.

"Distribution" means, in connection with the Property:

- (i) any payment, declaration of dividend or other distribution, whether in cash or property, (but expressly excluding any distribution by way of the payment of dividends by the issuance of equity securities of an issuer) to any holder of shares of any class of such Person; or
- (ii) any repurchase, redemption, retraction or other retirement or purchase for cancellation of shares of such Person, or of any options, warrants or other rights to acquire any of such shares; or
- (iii) the payment by any Obligor of any royalty, consulting fee, management fee, bonus or other fee to any Affiliate or to any director, officer or other management personnel of such Affiliate or of any Obligor; or
- (iv) making of any payment on account of any fees, principal, interest or otherwise on any subordinated debt,

and the verb "Distribute" has a corresponding meaning.

- "Drawdown" means the advance of the Credit Facility by way of a Prime Rate Loan, a CORRA Loan or a Term SOFR Loan.
- **"Drawdown Date"** means the date on which a Drawdown is made by the Borrower pursuant to the provisions hereof.
- "Drawdown Notice" means a notice, substantially in the form set out in Schedule 1.01(D), to be given to the Administrative Agent by the Borrower pursuant to Section 2.04.
- "Eligible Assignee" means any proposed assignee of an assignment pursuant to Article 20 that is:
 - (i) any Person, subject to Borrower consent (such consent not to be unreasonably withheld, and not required if an Event of Default has occurred and is continuing); or
 - (i) any Lender or an Affiliate of a Lender, and is a resident of Canada for the purposes of the *Income Tax Act* (Canada).
- "Eligible Hedge Agreements" means the agreements (including, without limitation, in the case of swap transactions, the ISDA Master Agreements) entered into between a Lender and the Borrower in respect of any Eligible Hedge Arrangement entered into pursuant to the terms of this Agreement, such agreements to be in the forms required by such Lender (with such changes and/or supplements as may be reasonably requested by the Borrower and as may be agreed to by such Lender, acting reasonably), as such agreements may be amended, restated, supplemented or otherwise modified from time to time.
- "Eligible Hedge Arrangement" means a rate swap transaction entered into by the Borrower with a Lender for the purpose of protecting or mitigating against risks in interest payable under

this Agreement (other than with respect to any Term CORRA Loans or Term SOFR Loans), but not for speculative purposes.

"Encumbrance" means, with respect to any Person, any mortgage, debenture, pledge, hypothec, lien, charge, assignment by way of security, hypothecation or security interest granted or permitted by such Person or arising by operation of law, in respect of any of such Person's property, or any consignment by way of security or Capital Lease of property by such Person as consignee or lessee, as the case may be, or any other security agreement, trust or arrangement having the effect of security for the payment of any debt, liability or other obligation, and "Encumbrances", "Encumbrancer", "Encumber" and "Encumbered" have corresponding meanings.

"Environmental Costs" means any and all claims (including, without limitation, personal injury, wrongful death, property damage, natural resource damage, strict liability and punitive damages), obligations, damages, losses, expenses, fines, penalties, fees, judgments, awards, amounts paid in settlement, costs of remedial or restoration work whether or not performed voluntarily (including, without limitation, any financial assurances required to be posted for completion of remediation and costs associated with administrative oversight), consultant, expert and legal fees (including, without limitation, the expenses of counsel for the Administrative Agent), investigations costs (including, without limitation, sampling, testing, analysis and monitoring of soil, water, air, building or other materials) incurred by the Administrative Agent or any Lender for any claim or cause of action as against the Administrative Agent or any Lender of any kind by any person arising directly or indirectly out of or in any way relating to any one or more of the following:

- (i) the presence, suspected presence, release or suspected release of any Hazardous Substance on, in or into the air, soil, groundwater, surface water or improvements at, on or about, under or within the Property, or any portion thereof, or elsewhere in connection with the use, disposal, treatment, storage, generation, production, manufacturing, processing, refining, control, management, abatement, removal, handling, transfer or transportation to or from the Property of any Hazardous Substances not in compliance with Environmental Law, whether or not the presence or release of such Hazardous Substance is known or unknown by the parties hereto at the time this Agreement is executed;
- (ii) any breach of any representation, warranty or covenant relating to Hazardous Substances and the Property made or given by any Obligor to the Administrative Agent and the Lenders or to any prospective or actual buyer or lessee of all or any portion of the Property;
- (iii) any act or omission of any Obligor or any other person or entity including damages or injuries to the person or property of any third party, in connection with Hazardous Substances;
- (iv) the failure of any Obligor to comply with any Environmental Law relating to any Hazardous Substance in, on, or around the Property;

- (v) the failure of any Obligor to protect public health and safety that is or may be threatened by any Hazardous Substance being present or released in, on, or around the Property; or
- (vi) the enforcement of the terms of this Agreement relating to Hazardous Substances or Environmental Laws.

"Environmental Law" means any Applicable Law relating to the environment, including those pertaining to:

- (i) reporting, licensing, permitting, investigating, remediating and cleaning up in connection with any presence or Release, or the threat of the same, of Hazardous Substances; and
- (ii) the manufacture, processing, distribution, use, treatment, storage, disposal, transport, handling and the like of Hazardous Substances, including those pertaining to occupational health and safety.

"Equity Interests" means, with respect to any Person, shares of capital stock of (or other ownership or profit interests in) such Person, warrants, options or other rights for the purchase or other acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or other acquisition from such Person of such shares (or such other interests), and other ownership or profit interests in such Person (including, without limitation, partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are authorized on any date of determination.

"Equivalent Amount" in one currency on any day means the amount of that currency into which a specified amount of another currency can be converted at the rate of exchange for such conversion as determined by the Administrative Agent, using the average rate of exchange for interbank transactions quoted by the Bank of Canada at approximately the close of business on such date (or, if not so quoted, the average rate of exchange for interbank transactions quoted by the Bank of Canada at approximately the close of business on the Business Day immediately preceding such date), and, if such average rate is for any reason unavailable, it shall mean the spot rate of exchange for wholesale transactions quoted by the Administrative Agent at approximately 4:00 p.m. (Toronto time) on such date in accordance with its usual practice, provided that if the Borrower has entered into an Eligible Hedge Agreement to hedge the amount of Term SOFR Loans against exchange rate fluctuations on terms satisfactory to the Administrative Agent, the Equivalent Amount of such Loan in Canadian Dollars will be the amount of Canadian Dollars provided for in such Eligible Hedge Agreement.

[&]quot;Erroneous Payment" has the meaning set out in Section 18.04(1).

[&]quot;Erroneous Payment Deficiency Assignment" has the meaning set out in Section 18.04(4)(a).

[&]quot;Erroneous Payment Impacted Credit Facility" has the meaning set out in Section 18.04(4)(a).

- "Erroneous Payment Return Deficiency" has the meaning set out in Section 18.04(4).
- **"Event of Default"** has the meaning set out in Section 12.01.
- "Excess Amount" has the meaning set out in Section 6.05.
- "Excluded Taxes" means, with respect to the Administrative Agent, any Lender or any other recipient of any payment to be made by or on account of any obligation of the Borrower hereunder or under any other Loan Document, (a) taxes imposed on or measured by its net income, capital gains, or capital, and franchise taxes imposed on it by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable lending office is located, (b) any branch tax, branch profits tax or any similar tax imposed by any jurisdiction or (c) any taxes imposed under FATCA.
- "External Credit Rating" means the public corporate issuer rating of the Guarantor provided by the Rating Agencies providing such rating at the relevant time.
- **"FATCA"** means Sections 1471 through 1474 of the Internal Revenue Code of 1986 of the United States of America (the "**IRC**"), as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof.
- "Fiscal Quarter" means a period of three consecutive months in each Fiscal Year ending on March 31, June 30, September 30, and December 31, as the case may be, of such year.
- "Fiscal Year" means in relation to any Person, its fiscal year commencing on January 1 of each calendar year and ending on December 31 of the same calendar year.
- "Floor" a rate of interest equal to zero percent (0%).
- "Force Majeure" means any of the following events which prevents or materially impairs the operation of the Property and is not caused by and is beyond the reasonable control of the Obligors: acts of God, floods, earthquakes, hurricanes, windstorms, severe weather conditions, lightning, fire, wars (whether declared or not), riots, insurrections, rebellions, civil commotions, sabotage, partial or entire failure of utilities, strikes, walkouts or other labour disruptions, delays in transportation, accidents, shortages of and inability to procure labour, materials and supplies (after all commercially reasonable efforts have been made by the Obligors to obtain replacement for such labour, materials and supplies) or orders, legislation, regulations and directives of any Governmental Authorities. For greater certainty, lack of funds, the state of the market or any wilful or negligent act or omission on the part of the Obligors (or any of them) does not constitute Force Majeure.
- "GAAP" means generally accepted accounting principles in Canada as applicable to public entities that are from time to time approved by the Canadian Institute of Chartered Accountants, or any successor institute, and for greater certainty includes IFRS.

"General Assignment of Leases" means the assignment of the Leases and the rent thereunder made by the Nominee in favour of the Administrative Agent.

"Governmental Authority" means any person, board commission, branch or office of any federal, provincial, regional, municipal or other governmental department or any governmental body or any agency, parliament, legislature, or any regulatory authority, board of any government, parliament or legislature, authority, bureau, commission, department, or instrumentality thereof, or any court, tribunal or arbitrator, including, without limitation, any securities commission or stock exchange or any court or (without limitation to the foregoing) any other law, regulation or rule making entity (including, without limitation, any central bank, fiscal or monetary authority or authority regulating banks), having or purporting to have jurisdiction in the relevant circumstances over an Obligor, the whole or any part of the Property, the Administrative Agent or the Lenders or the transaction as the case may be and shall include a board or association of insurance underwriters, as well as any public utility supplying utility services to the Property. Notwithstanding anything herein to the contrary, (a) all requests, rules, guidelines, requirements and directives promulgated by the Administrative Agent or the Lenders for International Settlements, and Basel Committee on Banking Supervision (or any successor or similar authority) or by United States, Canadian or foreign regulatory authorities, in each case pursuant to Basel III, and (b) the Dodd-Frank Wall Street Reform and Consumer Protection Act (United States) and all requests, rules, guidelines, requirements and directives thereunder or issued in connection therewith or in implementation thereof, shall in each case be deemed to be enactments of an Governmental Authority regardless of the date enacted, adopted, issued or implemented.

"Guarantor" means RioCan Real Estate Investment Trust and its permitted successors and assigns.

"Hazardous Substance" means any substance or material that is prohibited, controlled or regulated by any Governmental Authority pursuant to Environmental Law, including pollutants, contaminants, dangerous goods or substances, toxic or hazardous substances or materials, wastes (including solid non-hazardous wastes and subject wastes), petroleum and its derivatives and byproducts and other hydrocarbons, all as defined in or pursuant to any Environmental Law.

"Head Lease" means the head lease dated as of September 26, 2002, between the Head Landlord, as landlord, and the Store Ground Lessor, as tenant, as amended, restated, modified, supplemented and assigned from time to time.

"Head Landlord" means, collectively, OMERS Realty Corporation, OMERS Realty Holdings (Yorkdale) Inc., ARI YKD GP Inc. and ARI YKD Investments LP, and their respective successors and assigns.

"IFRS" mean the International Financial Reporting Standards for public entities as adopted by the Canadian Accounting Standards Board from time to time.

"Indebtedness" means (i) any obligation for borrowed money (including for greater certainty, the full principal amount of convertible debt), (ii) any obligation incurred in connection with the acquisition of property, assets or businesses, (iii) any obligation issued or assumed as the

deferred purchase price of property, (iv) any obligation under a Capital Lease, (v) obligations under letters of credit, guarantees and indemnities issued in respect of borrowed money and any reimbursement obligation or other obligation in connection with a bankers' acceptance or any similar instrument (excluding, for greater certainty, "performance" letters of credit issued in connection with construction or development relating to a property), (vi) all other indebtedness upon which interest charges are customarily paid by such Person and characterized as indebtedness under GAAP, (vii) the aggregate amount at which any securities of such Person that are not qualified by a prospectus and are redeemable or retractable at the option of the holder of such shares (except where the holder is such Person) may be redeemed or retracted prior to the Maturity Date for cash or obligations constituting Indebtedness or any combination thereof, and (viii) all Contingent Obligations incurred for the purpose of or having the effect of providing financial assistance to another entity.

"Indemnified Taxes" means Taxes other than the Excluded Taxes.

"Indemnitee" has the meaning set out in Section 16.01(2).

"Interest Payment Date" means (a) with respect to any Prime Rate Loan, the first Business Day of each calendar month, (b) with respect to any Term CORRA Loan, the last day of the Interest Period applicable to the Drawdown of which such Loan is a part; (c) with respect to any Daily Compounded CORRA Loan, the last day of each Interest Period applicable to the Drawdown of which such Loan is a part, and (d) with respect to any Term SOFR Loan, the last Business Day of each applicable Interest Period and, if any Interest Period is longer than three (3) months, the last Business Day of such three (3) month period during such Interest Period.

"Interest Period" means:

- (i) with respect to each Prime Rate Loan, the period commencing on the Drawdown Date, as the case may be, and ending on the date selected or deemed to be selected by the Borrower or the Administrative Agent, as the case may be, for the Conversion of such Loan into another type of Loan or for the repayment of such Loan;
- (ii) with respect to each Term CORRA Loan, the initial period (subject to availability) of one (1) month, three (3) months or such other period as the Administrative Agent and the Lenders permit commencing on and including the date on which a Conversion Notice is made, or the Rollover Date, as the case may be, applicable to such Term CORRA Loan and ending on and excluding the last day of such initial period, and thereafter, each successive period (subject to availability) of approximately one (1) month, three (3) months or such other permitted period as selected by the Borrower and notified to the Administrative Agent in writing commencing on and including the last day of the prior Interest Period;
- (iii) with respect to each Daily Compounded CORRA Loan, the initial period (subject to availability) of approximately one (1) month or such other period as the Administrative Agent and the Lenders permit commencing on and including the date on which a Conversion Notice is made, or the Rollover Date, as the case may

- be, applicable to such Daily Compounded CORRA Loan and ending on and excluding the last day of such initial period, and thereafter, each successive period (subject to availability) of approximately one (1) month or such other permitted period commencing on and including the last day of the prior Interest Period; and
- (iv) with respect to each Term SOFR Loan, a period of one (1), three (3), six (6) months' duration, or such other period as the Administrative Agent and the Lenders, commencing on and including the date on which a Conversion Notice is made, or the Rollover Date, as the case may be, applicable to such Term SOFR Loan and ending on and excluding the last day of such initial period,

provided that in any case the last day of each Interest Period will be also the first day of the next Interest Period (provided that for the purposes of calculation of interest payable by the Borrower, the last day of each Interest Period shall not be included in such Interest Period but shall be included in the calculation of interest payable for the subsequent Interest Period, without duplication). If the last day of an Interest Period selected or deemed to be selected by the Borrower is not a Business Day, the Borrower will be deemed to have selected an Interest Period the last day of which is the Business Day next following the last day of the Interest Period otherwise selected unless such next following Business Day falls in the next calendar month in which event the Borrower will be deemed to have selected an Interest Period the last day of which is the Business Day immediately preceding the last day of the Interest Period otherwise selected and further provided that the last Interest Period hereunder must expire on or prior to the Maturity Date.

- "Joint Venture Arrangements" means any real estate asset or operation in which the Guarantor participates where the Guarantor does not own 100% of the equity interests in the asset or operation.
- "Lands" means the lands and premises municipally known as part of 3401 Dufferin Street, Toronto, Ontario, identified in Schedule B attached hereto. [NTD: Borrower to confirm.]
- "Lease" means any lease, sublease, agreement to lease, offer to lease, licence or right of occupation granted from time to time by or on behalf of the Borrower and/or the Nominee entitling the lessee, sublessee or grantee thereunder to use or occupy all or any part of the Property.
- "Lender Assignment and Assumption" means an assignment and assumption entered into by a Lender and an assignee pursuant to Article 20 and accepted by the Administrative Agent in a form approved by the Administrative Agent.
- "Lenders" means the Persons from time to time party to this Agreement and designated in Schedule A (which will be amended and distributed to all parties by the Administrative Agent from time to time as other Persons become Lenders) and "Lender" means any one of the Lenders and includes each of their successors and permitted assigns.
- "Lenders' Counsel" means the firm of McCarthy Tétrault LLP, or such other firm of legal counsel as the Lenders may from time to time designate.

- "Limited Partnership Agreement" means a Second Amended and Restated Limited Partnership Agreement dated November 25, 2015, as amended by an Amendment No. 1 to Second Amended and Restated Limited Partnership Agreement dated November 24, 2018, as may be further amended, supplemented, or restated from time to time.
- **"Loan"** means any extension of credit by the Lenders under this Agreement, including by way of a Prime Rate Loan or a CORRA Loan outstanding under the Credit Facility.
- "Loan Documents" means, collectively, (i) this Agreement, (ii) the Security, (iii) the Eligible Hedge Agreements, and (iv) all present and future agreements, documents, certificates and instruments delivered by any Obligor to the Administrative Agent or any Lender pursuant to or in respect of this Agreement or the Security, in each case as the same may from time to time be amended, and "Loan Document" means any one of the Loan Documents.
- "Majority of the Lenders" means, at any time at which there are two Lenders hereunder, both Lenders; and at any time at which there are more than two Lenders hereunder, it is to be Lenders whose Commitments represent at least 66 2/3% of the dollar amount of the total Commitments and shall include at least two Lenders.
- "Material Adverse Change" means any event or occurrence which, when considered individually or together with other events or occurrences, has a material adverse effect on (i) the business, assets, liabilities, operations, results of operations, condition (financial or other) or prospects of the Obligors, taken as a whole, (ii) the operation of the Property in its entirety, or (iii) the ability of any Obligor to perform its Obligations in all material respects. For greater certainty, "Material Adverse Change" does not include a change in general economic conditions unless same in turn causes any of the foregoing events.
- "Material Licences" means, collectively, all licences, Permits or approvals issued by any Governmental Authority, or any applicable stock exchange or securities commission, to the Borrower, and which are at any time on or after the date of this Agreement:
 - (i) necessary or material to the business and operations of the Property, the breach or default of which would result in a Material Adverse Change; or
 - (ii) designated by a Lender as a Material Licence with respect to the Property, and provided that the Administrative Agent has notified the Borrower of such designation in writing,

including, without limitation, those Material Licences existing as of the date of this Agreement listed in Schedule 1.01(E).

"Material Property Agreements" means, collectively:

- (i) the Store Ground Lease;
- (ii) the Store Sublease;
- (iii) the Limited Partnership Agreement;

- (iv) each operating contract with respect to the Property having a term more than one year and which contemplates payments thereunder in excess of \$1,000,000 per annum;
- (v) any shared facilities and/or reciprocal easement agreements;
- (vi) at any time all agreements, documents and instruments (other than Leases) now existing or from time to time entered into in the future or assigned to or obtained by the applicable Obligor with rights in the Property affecting or relating to the Property and which are material to the operation, use or ownership thereof; and
- (vii) all planning approvals, permits, licences, development agreements and other material contracts with respect to the Property designated as Material Property Agreements by the Administrative Agent from time to time, provided that the Administrative Agent has notified the Borrower of such designation,

and includes, without limitation, those listed in Schedule 1.01(E).

"Maturity Date" has the meaning set out in Section 2.10.

"Nominee Agreement" means the nominee agreement for the Property made as of November 25, 2015 between the Borrower, as beneficial owner, and the Nominee, as nominee, as amended.

"Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.

"Non-Schedule I Lenders" means those Lenders that are not banks or that are banks chartered under the *Bank Act* (Canada) and named in Schedule II or III thereto, and "Non-Schedule I Lender" means each such Lender.

"Obligations" means all obligations of the Obligors to the Administrative Agent or the Lenders under or in connection with this Agreement, the other Loan Documents, including all debts and liabilities, present or future, direct or indirect, absolute or contingent, matured or not, at any time owing by the Obligors to the Administrative Agent or the Lenders, or any of them, in any currency or remaining unpaid by the Obligors to the Administrative Agent or the Lenders, or any of them, under or in connection with this Agreement or the other Loan Documents whether arising from dealings between the Administrative Agent or the Lenders, or any of them, and the Obligors or from any other dealings or proceedings by which the Administrative Agent or the Lenders, or any of them, may be or become in any manner whatever a creditor or obligee of the Obligors pursuant to this Agreement and the other Loan Documents, and wherever incurred, and whether incurred by any Obligor alone or with another or others and whether as principal or surety, and all interest, fees, legal and other costs, charges and expenses relating thereto. For greater certainty, "Obligations" include all Indebtedness and liabilities of the Borrower to the Lenders or any of them pursuant to Eligible Hedge Agreements and the parties acknowledge and agree that the amount owing by the Borrower or owed to the Borrower under any Eligible Hedge Agreement for purposes of determining the amount thereof to be included in the determination of Obligations shall be the amount determined in accordance with the applicable Eligible Hedge Agreement.

- "Obligors" means, collectively the Borrower, the Guarantor and the Nominee, and "Obligor" means the relevant one of such Persons as the context requires.
- "Obligors' Counsel" means Fogler Rubinoff LLP, Stikeman Elliott LLP or such other firm of legal counsel as one or more of the Obligors may from time to time designate and that is acceptable to the Lenders.
- "Organizational Documents" means, with respect to any Person, such Person's articles, memorandum or other charter documents, partnership agreement, joint venture agreement, declaration of trust, trust agreement, by-laws, unanimous shareholder agreement, or any and all other similar agreements, documents and instruments pursuant to which such Person is constituted, organized or governed.
- "Other Taxes" means all present or future stamp or documentary taxes or any other similar excise or property taxes, charges or levies arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of this Agreement or any other Loan Document, but does not include Excluded Taxes.
- "Owner Parties" means, collectively, the Borrower and the Nominee, and "Owner Party" means any one of them.
- "Payment Recipient" has the meaning set out in Section 18.04(1).
- "Periodic Term CORRA Determination Day" has the meaning assigned to it under the definition of Term CORRA.
- "Permits" means, collectively, all permits, consents, orders, waivers, applications, authorizations, licences, certificates, approvals, registrations, franchises, rights, privileges and exemptions or the like issued or granted by any Governmental Authority or by any third party with respect to the Property (including, without limitation, any Permits relating to Environmental Laws).

"Permitted Encumbrances" means, with respect to the Property, the following:

- (i) liens for Taxes, rates, assessments or other governmental charges or levies not yet due, or for which instalments have been paid based on reasonable estimates pending final assessments, or if due, the validity of which is being contested diligently and in good faith by appropriate proceedings by that Person, provided that if the aggregate amount being contested is in excess of Cdn. \$100,000, the Borrower shall have deposited with the Administrative Agent collateral satisfactory to the Administrative Agent, to secure the payment of such Taxes and assessments;
- (ii) undetermined or inchoate liens, rights of distress and charges incidental to construction, maintenance or current operations that have not at such time been filed or exercised and of which the Administrative Agent has not been given notice, or that relate to obligations not due or payable, or if due, the validity of which is being contested diligently and in good faith by appropriate proceedings by that Person;

- (iii) reservations, limitations, provisos and conditions expressed in any original grant from the Crown or other grants of real or immovable property, or interests therein, that do not materially affect the use of the affected land for the purpose for which it is used by that Person;
- (iv) permits, reservations, covenants, servitudes, right of access or user licenses, easements, rights-of-way and rights in the nature of easements (including, without limitation, licenses, easements, rights-of-way and rights in the nature of easements for railways, sidewalks, public ways, sewers, drains, gas and oil pipelines, steam and water mains or electric light and power, or telephone, telecommunication, television and telegraph conduits, poles, wires and cables) that do not materially impair the use of the affected land for the purpose for which it is used by that Person, or in respect of which satisfactory arrangements have been made for relocation so that such use will not, in the aggregate, be materially and adversely impaired, or which that Person is bound to enter into pursuant to any agreement with a Governmental Authority or with respect to any Material Property Agreements;
 - (v) title defects, irregularities or other matters relating to title that are of a minor nature and that in the aggregate do not materially impair the use of the affected property for the purpose for which it is used by that Person;
- (vi) the right reserved to or vested in any Governmental Authority by the terms of any lease, licence, franchise, grant or permit acquired by that Person or by any statutory provision to terminate any such lease, licence, franchise, grant or permit, or to require annual or other payments as a condition to the continuance thereof;
- (vii) the Encumbrance resulting from the deposit of cash or securities in connection with contracts, tenders or expropriation proceedings, or to secure workers' compensation, employment insurance, surety or appeal bonds, costs of litigation when required by law, liens and claims incidental to current construction, mechanics', warehousemen's, carriers' and other similar liens, and public, statutory and other like obligations incurred in the ordinary course of business;
- (viii) security given to a public utility or any Governmental Authority when required by such utility or authority in connection with the operations of that Person in the ordinary course of its business;
 - (ix) the Encumbrance created by a judgment of a court of competent jurisdiction, or a claim (including claims pursuant to the *Construction Act* (Ontario)) filed, against that Person as long as the judgment or claim is being contested diligently and in good faith by appropriate proceedings by that Person and does not result in an Event of Default, provided that if such judgment or claim is a construction lien or is in the aggregate greater than Cdn. \$500,000, the Borrower shall have either (A) in the case of any such judgment or claim that is not a construction lien, if acceptable to the Lenders, deposited with the Administrative Agent collateral satisfactory to the Lenders, to secure the payment of such judgment or claim, or (B) posted a payment

bond, or made payment into court, of such amount as is necessary to remove such Encumbrance;

- (x) the Security;
- (xi) encroachments by the Property or structures thereon over neighbouring lands (including public streets) and minor encroachments by neighbouring lands or structures thereon over the Property, so long as, in the former case, there are written agreements permitting such encroachments;
- (xii) subdivision, development, servicing and site plan agreements, undertakings and agreements made pursuant to applicable planning and development legislation, entered into with or made in favour of any Governmental Authority, or public or private utility, relating to the Property;
- (xiii) Leases relating to the Property that have been approved by the Administrative Agent or entered into in accordance with this Agreement and notices of them;
- (xiv) all municipal by-laws and regulations and other municipal land use instruments, including, without limitation, official plans, zoning and building by-laws, restrictive covenants and other land use limitations, public or private, and other restrictions as to the use of the Property;
- (xv) any Encumbrance described in Schedule 1.01(F);
- (xvi) such other Encumbrances as are agreed to in writing by the Administrative Agent; and
- (xvii) reciprocal agreements, shared facilities agreements, party wall agreements and rights of way.

"Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"Prime Rate" means, on any day, the annual rate of interest equal to the greater of

- (i) the annual rate of interest announced by Royal Bank of Canada and in effect as its prime rate at its principal office in Toronto, Ontario on such day for determining interest rates on Canadian Dollar denominated commercial loans in Canada; and
- (ii) the annual rate of interest equal to the sum of (A) Adjusted Term CORRA (or the Benchmark Replacement, as applicable) for an interest period of one month in effect on such day, plus (B) 1.00% per annum,

and provided that in no event shall the Prime Rate be less than zero for the purposes of this Agreement.

- "Prime Rate Loan" means a Loan in Canadian Dollars made by the Lenders to the Borrower with respect to which the Borrower has specified that interest is to be calculated by reference to the Prime Rate plus the Prime Rate Margin.
- "Prime Rate Loan Interest Payment Date" means, with respect to each Prime Rate Loan, the first Business Day of each calendar month.
- "Prime Rate Margin" means, for any period, the Applicable Margin with respect to Prime Rate Loans.
- **"Property"** means the Lands and the department store consisting of approximately 303,438 square feet and known as "The Bay", and all landscaping, all plants, machinery, improvements and equipment and all other property whether free-standing or otherwise, auxiliary or ancillary thereto or connected therewith or added thereto, to be constructed on, above or under the surface of the Lands, in each case in respect of which the Borrower or the Nominee has a freehold or leasehold ownership interest.
- "**Property Assets**" means any parcel of real property owned or leased (in whole or in part) by the Guarantor or any of its Subsidiaries.
- "Proportionate Consolidation Adjustments" means accounting adjustments to reflect assets, liabilities, unitholders' equity, revenues and expenses on a proportionate basis in place of the Guarantor's use of equity accounting in accordance with GAAP with respect to real estate investments or interest in which the Guarantor participates.
- "Qualifying Joint Venture Arrangements" means any Joint Venture Arrangement where the Guarantor owns and maintains at all times either (a) at least 50.1% of the equity and voting interests in such Joint Venture Arrangement, or (b) at least 50% of the equity and voting interests in such Joint Venture Arrangement and the Guarantor or a wholly-owned Subsidiary is a property and/or asset manager of such Joint Venture Arrangement.
- "Rateable Portion" means, for any Lender, with respect to the Credit Facility, at any given time, with regard to any amount of money, the product obtained by multiplying that amount of money by the quotient obtained by dividing (i) that Lender's Commitment with respect to such Credit Facility or portion thereof by (ii) the aggregate of all of the Lenders' Commitments with respect to the Credit Facility or portion thereof at such time, as adjusted in accordance with the terms hereof.
- "Rating Agencies" means DBRS and S&P.
- "Related Persons" means, with respect to any Person, such Person's Affiliates and the directors, officers, employees, agents and advisors of such Person and such Person's Affiliates, and "Related Person" means any one of them.
- "Release" means any release or discharge of any Hazardous Substance including any discharge, spray, injection, inoculation, abandonment, deposit, spillage, leakage, seepage, pouring, emission, emptying, throwing, dumping, placing, exhausting, escape, leach, migration, dispersal, dispensing or disposal.

- "Relevant Governmental Body" means the Bank of Canada, or a committee officially endorsed or convened by the Bank of Canada, or any successor thereto.
- "Relevant Jurisdiction" means, from time to time, with respect to a Person that is granting Security hereunder, any province or territory of Canada, any state of the United States or any other country or political subdivision thereof in which such Person has its chief executive office or chief place of business or has property that is subject to the Security.
- "Repayment Notice" means the notice substantially in the form set out in Schedule 1.01(I).
- "Requirements of Environmental Law" means all requirements of the common law or of statutes, regulations, by-laws, ordinances, treaties, judgments and decrees, and (to the extent that they have the force of law) rules, policies, guidelines, orders, approvals, notices, permits, directives and the like, of any Governmental Authority in Ontario relating to environmental or occupational health and safety matters (as they relate to exposure to a Hazardous Substance) and the assets and undertaking of such Person and the intended uses thereof in connection with such matters, including all such requirements relating to (i) the protection, preservation or remediation of the natural environment (namely, the air, land, surface water or groundwater), (ii) solid, gaseous or liquid waste generation, handling, treatment, storage, disposal or transportation, (iii) consumer, occupational or public safety and health (as they relate to exposure to a Hazardous Substance), and (iv) Hazardous Substances or conditions (namely, matters that are prohibited, controlled or otherwise regulated, such as contaminants, pollutants, toxic substances, dangerous goods, wastes, hazardous wastes, liquid industrial wastes, hazardous materials, petroleum and other materials such as urea formaldehyde and polyurethane foam insulation, asbestos or asbestos-containing materials, polychlorinated biphenyls (PCBs) or PCB contaminated fluids or equipment, lead based paint, explosives, radioactive substances, petroleum and associated products, above ground and underground storage tanks or surface impoundments).
- "Requirements of Law" means, with respect to any Person, the Organizational Documents of such Person and any Applicable Law or any determination of a Governmental Authority, in each case applicable to or binding upon such Person or any of its business or property or to which such Person or any of its business or property is subject.
- "Rollover" means, with respect to any CORRA Loan or Term SOFR Loan, the continuation of all or a portion of such Loan (subject to the terms and conditions hereof) for an additional Interest Period subsequent to the initial or any subsequent Interest Period applicable thereto.
- **"Rollover Date"** means the date of commencement of a new Interest Period applicable to a CORRA Loan or Term SOFR Loan that is being rolled over.
- "Rollover Notice" means a notice, substantially in the form set out in Schedule 1.01(B), to be given to the Administrative Agent by the Borrower pursuant to Section 2.04.
- "S&P" means Standard & Poor's Ratings Group, a division of the McGraw-Hill Companies and its successors.

- "Schedule I Lenders" means those banks that are chartered under the *Bank Act* (Canada) and named in Schedule I thereto, and "Schedule I Lender" means each such bank.
- "Security" means the documents creating an Encumbrance in favour of, or any collateral held from time to time by, the Administrative Agent for and on behalf of the Lenders, in each case securing or intended to secure repayment of the Obligations, including all security described in Article 10.
- "SOFR" means a rate per annum equal to the secured overnight financing rate as administered by the SOFR Administrator.
- "SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).
- "SOFR Margin" means, for any period, the Applicable Margin with respect to Term SOFR Loans.
- "Store Ground Lease" means the ground lease dated as of September 26, 2002, between the Store Ground Lessor, as landlord, and the Borrower (as successor-in-interest to the Nominee), as tenant, as amended by a letter agreement dated October 24, 2002, as further amended by a letter agreement dated June 28, 2011, as further amended by an amending agreement dated as of April 3, 2014, in respect of the Lands, as amended, restated, modified, supplemented and assigned from time to time in accordance with the terms and conditions of this Agreement.
- "Store Ground Lessor" means Yorkdale Shopping Centre Holdings Inc., its successors and assigns.
- "Store Ground Lessor Consent" means the leasehold lender agreement dated as of January 26, 2024, between the Store Ground Lessor, the Head Lease Landlord, the Borrower and the Administrative Agent, in respect of the Head Lease and the Store Ground Lease.
- "Store Sublease" means the sublease dated as of November 25, 2015, between the Borrower, as sublandlord, and the Nominee, as subtenant, as amended, restated, modified, supplemented and assigned from time to time in accordance with the terms and conditions of this Agreement.
- "Subordinated Indebtedness" means Indebtedness of the Guarantor (or its successor) (i) that is expressly subordinate in right of payment to the Obligations and other amounts payable under the Loan Documents, (ii) that matures after the Maturity Date, and (iii) the payment of principal and interest of which can be satisfied, at the Guarantor's (or its successor's) sole option, through the issuance of Equity Interests.
- "Subsidiary" means, at any time, with respect to any Person, any other Person, if at such time the first mentioned Person (i) owns, directly or indirectly, securities or other ownership interests in such other Person, having ordinary voting power to elect a majority of the board of directors or persons performing similar functions for such other Person, and (ii) directly or indirectly, through the operation of any agreement or otherwise, the ability to elect or cause the election of a majority of the board of directors or other persons performing similar functions for such other Person or otherwise exercise control over the management and policies of such other Person, and

in either case will include any other Person in like relationship to a Subsidiary of such first mentioned Person.

"Tax" and "Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto, and including any realty taxes, duties, rates, imposts, levies, assessments and other similar charges, whether general or special, ordinary or extraordinary, or foreseen or unforeseen, including municipal taxes, school taxes and local improvement charges and all related interest, penalties and fines which at any time may be levied, assessed, imposed or form an Encumbrances upon real property.

"Term CORRA" means, for any calculation with respect to a Term CORRA Loan, the Term CORRA Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term CORRA Determination Day") that is two (2) Business Days prior to the first day of such Interest Period, as such rate is published by the Term CORRA Administrator; provided, however, that if as of 1:00 p.m. (Toronto time) on any Periodic Term CORRA Determination Day the Term CORRA Reference Rate for the applicable tenor has not been published by the Term CORRA Administrator and a Benchmark Replacement Date with respect to the Term CORRA Reference Rate has not occurred, then Term CORRA will be the Term CORRA Reference Rate for such tenor as published by the Term CORRA Administrator on the first preceding Business Day for which such Term CORRA Reference Rate for such tenor was published by the Term CORRA Administrator so long as such first preceding Business Day is not more than three (3) Business Days prior to such Periodic Term CORRA Determination Day.

"Term CORRA Adjustment" means, with respect to Term CORRA, for an Interest Period of a duration of (a) one (1) month, a percentage equal to 0.29547% per annum (29.547 basis points), and (b) three (3) months, a percentage equal to 0.32138% per annum (32.138 basis points).

"Term CORRA Administrator" means Candeal Benchmark Administration Services Inc., TSX Inc., or any successor administrator.

"Term CORRA Loan" means a Loan made by the Lenders to the Borrower that bears interest at a rate based on Adjusted Term CORRA other than pursuant to clause (ii) of the definition of "Prime Rate".

"Term CORRA Reference Rate" means the forward-looking term rate based on CORRA.

"Term SOFR" means, for any Interest Period for a Term SOFR Loan, the Term SOFR Reference Rate (rounded upward to the nearest fifth decimal place, if necessary) for a tenor comparable to the applicable Interest Period on the day (the "Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Term SOFR Determination Day, the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as

published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Term SOFR Determination Day.

"Term SOFR Adjustment" means, with respect to Term SOFR, (i) 0.11448% (11.448 basis points) for an Interest Period of one month's duration, (ii) 0.26161% (26.161 basis points) for an Interest Period of three months' duration, and (iii) 0.42826% (42.826 basis points) for an Interest Period of six months' duration.

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by the Lender in its discretion, acting reasonably).

"Term SOFR Determination Day" has the meaning set out in the definition of "Term SOFR".

"Term SOFR Loan" means a Loan in U.S. Dollars that bears interest at a rate based on Adjusted Term SOFR.

"Term SOFR Reference Rate" means the forward-looking term rate based on SOFR.

"Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Unitholders' Equity" means, at any time, the aggregate amount of unitholders' equity of the Guarantor, as shown on the Guarantor's most recent consolidated balance sheet at such time, calculated in accordance with GAAP.

"Unencumbered Property Assets" means, as at any date, the Property Assets of the Guarantor other than Property Assets (a) that are Encumbered; (b) of any Subsidiary of the Guarantor that has incurred unsecured indebtedness for borrowed money (or guaranteed any such Indebtedness); or (c) the Guarantor designates in writing to the Administrative Agent shall not constitute an Unencumbered Property Asset under this Agreement (until the Guarantor designates otherwise in writing to the Administrative Agent), provided that, immediately following such designation, there is no Default under Section 9.05(e) of this Agreement.

"Unencumbered Property Assets Value" means the aggregate fair market value of the Guarantor's Unencumbered Property Assets (excluding the fair market value of all Properties located outside Canada), determined on a consolidated basis and using the valuation methodology described by the Guarantor in its most recently published annual or interim financial reporting, in each case prepared and determined in accordance with GAAP; provided that (a) Unencumbered Property Assets encompassing property under development, undeveloped land and non-income producing properties shall not in the aggregate exceed 5% of the Unencumbered Property Assets Value at any time for the purposes of the calculation in Section 9.05(e); (b) Unencumbered Property Assets encompassing Qualifying Joint Venture Arrangements shall not in aggregate exceed 15% of Unencumbered Property Assets Value at any time for the purposes of the calculation in Section 9.05(e); and (c) Unencumbered Property

Assets encompassing Qualifying Joint Venture Arrangements (as defined in clause (b) of the definition of Qualifying Joint Venture Arrangements) to which the Guarantor or any of its Subsidiaries is not the sole property and/or asset manager of such Qualifying Joint Venture Arrangement shall not in aggregate exceed 5% of Unencumbered Property Assets Value at any time for the purposes of the calculation in Section 9.05(e).

"U.S. Government Securities Business Day" means any day except for (i) a Saturday, (ii) a Sunday, or (iii) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"United States Dollars", "U.S. Dollars" and "U.S. \$" mean the lawful money of the United States of America.

1.02 <u>Extended Meanings</u>

In this Agreement words importing the singular number include the plural and vice versa, words importing any gender include all genders and words importing persons include individuals, corporations, limited and unlimited liability companies, general and limited partnerships, associations, trusts, unincorporated organizations, joint ventures and Governmental Authorities. The term "including" means "including without limiting the generality of the foregoing" and the term "third party" means any Person other than a Person who is a party to this Agreement. References to any document, instrument or agreement (i) shall include all exhibits, schedules and other attachments thereto, (ii) shall include all documents, instruments or agreements issued or executed in replacement thereof, and (iii) shall mean such document, instrument or agreement, or replacement or predecessor thereto, as amended, modified and supplemented from time to time in accordance with its terms and the terms of this Agreement and in effect at any given time. Reference in this Agreement or any Loan Document to any enactment, including any statute, law, by law or regulation, shall be deemed to include references to such enactment as re-enacted, amended or extended from time to time (save where expressly stated to the contrary), and references to any Loan Document (including schedules and exhibits thereto), Permitted Encumbrances "as amended" shall mean and be a reference to such documents, instruments or agreements as amended, modified and supplemented from time to time in accordance with its terms and the terms of this Agreement and in effect at any given time. Any reference to any Person shall include its successors and permitted assigns in the capacity indicated, and in the case of any Governmental Authority, any Person succeeding to its functions and capacities. Any reference to Requirements of Law and Applicable Laws shall include all references to such Requirements of Law and Applicable Laws as amended, supplemented or replaced. The Loan Documents are the result of negotiations between, and have been reviewed by each Obligor, each Lender and their respective counsel. Accordingly, the Loan Documents shall be deemed to be the product of all parties thereto, and no ambiguity shall be construed in favour of or against any Obligor or any Lender except as expressly provided under the Loan Documents. Upon an acceleration of the Loans following the occurrence of an Event of Default pursuant to Section 12.02, any reference in this Agreement or any Loan Document to an Event of Default that has occurred and is "continuing" shall be deemed to be an Event of Default that has occurred and is continuing for all times thereafter unless otherwise agreed in writing by

the Lenders in their sole discretion; and the terms "continued", "continuation" and "discontinuation" shall have corresponding meanings.

1.03 <u>Accounting Principles</u>

- (1) Where the character or amount of any asset or liability or item of revenue or expense is required to be determined, or any consolidation or other accounting computation is required to be made, for the purpose of this Agreement or any Loan Document, such determination or calculation will, to the extent applicable and except as otherwise specified herein or as otherwise agreed in writing by the parties, be made in accordance with GAAP applied on a consistent basis.
- (2) Each of the parties hereto acknowledges that the financial covenants and financial ratios contained in this Agreement have been established and agreed upon on the basis of the current accounting policies, practises and calculation methods or components thereof adopted by the Obligors on a consolidated basis and made on a basis consistent with GAAP. If at any time any change in GAAP applicable to the Borrower or the Guarantor would affect the computation of any financial ratio or requirement set forth in this Agreement, and the Borrower, the Guarantor or the Administrative Agent (on the instructions of the Majority of the Lenders) shall so require, the Administrative Agent, the Guarantor and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP; provided that, until so amended, (i) such ratio or replacement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the applicable Obligors shall provide to the Administrative Agent financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

1.04 Interest Calculations and Payments; Interest Rate Disclaimer

- (1) All interest payments to be made under this Agreement will be paid without allowance or deduction for deemed re-investment or otherwise, both before and after maturity and before and after default and/or judgment, if any, until payment of the amount on which such interest is accruing, and interest will accrue on overdue interest, if any.
- (2) Unless otherwise stated, wherever in this Agreement reference is made to a rate of interest "per annum" or a similar expression is used, such interest will be calculated on the basis of (a) a calendar year of 365 days or 366 days, as the case may be, for Prime Rate Loans, (b) a calendar year of 365 days for CORRA Loans, and (c) a calendar year of 360 days for Term SOFR Loans, and in each case using the nominal rate method of calculation and not the effective rate method of calculation or on any other basis that gives effect to the principle of deemed reinvestment of interest. Interest will continue to accrue after maturity and default and/or judgment, if any, until payment thereof, and interest will accrue and be compounded monthly on overdue interest, if any.
- (3) For the purposes of the *Interest Act* (Canada) and disclosure under such act, whenever interest to be paid under this Agreement is to be calculated on the basis of a year of

365 days or any other period of time that is less than a calendar year, the yearly rate of interest to which the rate determined pursuant to such calculation is equivalent is the rate so determined multiplied by the actual number of days in the calendar year in which the same is to be ascertained and divided by either 365 or such other period of time, as the case may be.

- (4) Notwithstanding anything herein to the contrary, the Administrative Agent shall calculate all fees and interest according to the terms of this Agreement and any other agreement entered into between the Borrower and the Administrative Agent and the Lenders in connection with the Credit Facility. For greater certainty all such calculations shall be without duplication of any day such that neither interest nor fees shall be calculated in respect of the same day twice.
- The Administrative Agent does not warrant or accept responsibility for, and shall not have any liability with respect to (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Prime Rate, Term CORRA, Daily Compounded CORRA, Adjusted Term CORRA, Adjusted Daily Compounded CORRA, Term SOFR, Adjusted Term SOFR or any component definition thereof or rates referred to in the definition thereof, or any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement) will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Prime Rate, Term CORRA, Daily Compounded CORRA, Adjusted Term CORRA, Adjusted Daily Compounded CORRA, Term SOFR, Adjusted Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. The Administrative Agent and its affiliates or other related entities may engage in transactions that affect the calculation of the Prime Rate, Term CORRA, Daily Compounded CORRA, Adjusted Term CORRA, Adjusted Daily Compounded CORRA, Term SOFR, Adjusted Term SOFR, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto, in each case, in a manner adverse to the Borrower. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain the Prime Rate, Term CORRA, Daily Compounded CORRA, Adjusted Term CORRA, Adjusted Daily Compounded CORRA, Term SOFR, Adjusted Term SOFR or any other Benchmark, in each case pursuant to the terms of this Agreement, and shall have no liability to any Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service

1.05 <u>Permitted Encumbrances</u>

The inclusion of reference to Permitted Encumbrances in any Loan Document is not intended to subordinate and will not subordinate any Encumbrance created by any of the Security to any Permitted Encumbrance.

1.06 <u>Currency</u>

Unless otherwise specified in this Agreement, all references to currency (without further description) are to lawful money of Canada.

1.07 Conflicts

In the event of a conflict between the provisions of this Agreement and the provisions of any other Loan Document, then, unless such Loan Document or an acknowledgement from the Obligors and the Administrative Agent relative to such Loan Document expressly states that this Section 1.07 is not applicable to such Loan Document, notwithstanding anything else contained in such other Loan Document, the provisions of this Agreement will prevail and the provisions of such other Loan Document will be deemed to be amended to the extent necessary to eliminate such conflict.

1.08 <u>Nature of Obligors' Liability</u>

- (1) Subject to Sections 1.08(2) and 1.08(3), nothing in any of the Loan Documents shall mean, nor be construed to mean, that the recourse of the Administrative Agent and the Lenders against the Obligors is anything other than full recourse with regard to its obligations hereunder, the manner and order of realization or the exercise of remedies hereunder or under the Security. Except where stated otherwise, the nature of the Obligors' obligations shall be joint and several.
- (2) Notwithstanding any other provision of the Loan Documents, the Administrative Agent and Lenders acknowledge and agree that the obligations and liabilities of the Guarantor under any Loan Document are not personally binding upon, and resort shall not be had to, nor shall recourse or satisfaction be sought from the private property of any of:
 - (a) the unit holders of the Guarantor;
 - (b) annuitants under a plan of which a unit holder of the Guarantor acts as a trustee or carrier; and
 - (c) the officers, trustees, employees or agents of the Guarantor.
- (3) Notwithstanding anything to the contrary contained herein, in respect of recourse against the Owner Parties only, the obligations of and rights and remedies against each of the Owner Parties only hereunder and the other Loan Documents to which such Owner Party is a party shall be performed, satisfied and paid out of and recourse shall be limited to, and enforcement taken against, only the Collateral and no other recourse shall be had, judgment issued or execution or other process levied against any Owner Party or against any assets of any Owner Party other than the Collateral (for greater certainty, nothing herein shall limit or restrict the right of the Administrative Agent or the Lenders to name any of the Owner Parties as a party in any action, proceeding or other remedial or enforcement proceeding so long as no monetary judgment or claim is sought or enforced against its assets, other than the Collateral).

1.09 Authorization and Direction

The Obligors represent and warrant that the Nominee is the bare trustee and nominee for holding legal title to the Property solely for the Borrower, pursuant to the Nominee Agreement. The Borrower irrevocably and unconditionally authorizes and directs the Nominee to execute and deliver to the Administrative Agent and the Lenders, and to perform all of its obligations under, this Agreement and the other Loan Documents to be executed by the Nominee and such other agreements, instruments and documents as may be necessary or desirable from time to give effect to this Agreement and the other Loan Documents.

1.10 Schedules

The following are the Schedules attached hereto and incorporated by reference and deemed to be part hereof:

Schedule A Lenders and Commitments Schedule B Legal Description of Lands Certificate re: Identity Schedule C Compliance Certificate Schedule 1.01(A) Rollover Notice Schedule 1.01(B) Conversion Notice Schedule 1.01(C) Schedule 1.01(D) Drawdown Notice Schedule 1.01(E) Material Licences and Material **Property Agreements** Additional Permitted Encumbrances Schedule 1.01(F) Schedule 1.01(I) Repayment Notice Schedule 8.01(14) Ownership Structure Schedule D **Applicable Margins** Schedule E Benchmark Replacement Provisions (SOFR)

ARTICLE 2- THE CREDIT FACILITY

2.01 Credit Facility

Subject to the terms and conditions of this Agreement, the Lenders establish the Credit Facility in favour of the Borrower, which Credit Facility is non-revolving and available by way of Prime Rate Loans, CORRA Loans and/or Term SOFR Loans.

The aggregate of all Loans outstanding under the Credit Facility shall not at any time exceed the Commitment.

2.02 **Purpose of Credit Facility**

The Credit Facility is being made available to the Borrower to provide funds for the purpose of interim refinancing the Property and the balance being distributed to the Borrower.

2.03 Manner of Borrowing

Subject to the terms and conditions hereof, the Borrower may make a single Drawdown under the Credit Facility by way of a Prime Rate Loan. Any amounts not drawn down on the Closing Date under the Credit Facility shall be cancelled.

2.04 Drawdowns, Conversions and Rollovers

- (1) The Borrower must give the Administrative Agent a Drawdown Notice, Conversion Notice or Rollover Notice, as the case may be, on or prior to 11:00 a.m. (Toronto time) on the day which is two (2) Business Days prior to the proposed Drawdown Date, Conversion Date or Rollover Date, as applicable. The Drawdown Date and each Conversion Date and Rollover Date must be a Business Day.
- (2) Subject to the provisions of this Agreement, the Borrower may (i) convert the whole or any part of any type of Loan into any other type of Loan, or (ii) roll over any CORRA Loans or Term SOFR Loans on the last day of the applicable Interest Period therefor, by giving the Administrative Agent a Conversion Notice or Rollover Notice, as the case may be.
- (3) The maximum amount outstanding under the Credit Facility at any time shall not exceed the Credit Facility Commitment.
- (4) If the Borrower fails to deliver a Conversion Notice or a Rollover Notice to the Administrative Agent for any maturing CORRA Loan or Term SOFR Loan as provided in this Section 2.04, then on the last day of the Interest Period applicable to such CORRA Loan or Term SOFR Loan, as applicable, such CORRA Loan or Term SOFR Loan shall be converted automatically into a Prime Rate Loan in the principal amount of such CORRA Loan or the Equivalent Amount of such Term SOFR Loan, determined on the date of Conversion thereof.

2.05 **Lenders' Obligations**

Upon receipt of a Drawdown Notice, the Administrative Agent shall forthwith notify the Lenders of the proposed Drawdown Date of each Lender's Rateable Portion of the Loan. Each Lender shall, on or before 11:00 a.m. (Toronto time) on the Drawdown Date specified by the Borrower in the applicable Drawdown Notice, credit the Administrative Agent's account with the amount of such Lender's Rateable Portion of such Loan. By 2:00 p.m. (Toronto time) on the same date, subject to Sections 7.02 and 18.01, the Administrative Agent shall pay to the Borrower in accordance with the irrevocable direction set out in the Drawdown Notice, the full amount of the amounts so credited.

2.06 Irrevocability

Each Drawdown Notice, Conversion Notice and Rollover Notice given by the Borrower hereunder is irrevocable and will oblige the Borrower to take the action contemplated on the date specified therein.

2.07 Account of Record

The Administrative Agent shall open and maintain books of account evidencing all Loans and all other amounts owing by the Borrower to the Lenders hereunder. The Administrative Agent shall enter in the foregoing accounts details of all amounts from time to time owing, paid or repaid by the Borrower hereunder. The information entered in the foregoing accounts shall constitute *prima facie* evidence, absent manifest error, of the Obligations of the Borrower to the Lenders hereunder with respect to all Loans and all other amounts owing by the Borrower to the Lenders hereunder. After a request by the Borrower, the Administrative Agent shall promptly advise the Borrower of such entries made in its books of account.

2.08 Interest on Excess Loans, Unpaid Costs and Expenses

Unless the payment of interest is otherwise specifically provided for herein, where the Borrower fails to pay any amount required to be paid by it hereunder when due having received notice that such amount is due (including, without limitation, the portion of any Loan made under a Credit Facility hereunder that exceeds the Commitment), the Borrower shall pay interest on such unpaid amount from the time such amount is due until paid at an annual rate equal to the Prime Rate plus the Prime Rate Margin plus 2.0% per annum (for greater clarity, such higher rate shall not apply to overdue interest on Loans).

2.09 Inability to Determine Rates

- (1) Subject to Section 2.11, if, on or prior to the first day of any Interest Period for any Term CORRA Loan or Daily Compounded CORRA Loan, as applicable:
 - (a) the Administrative Agent determines (which determination shall be conclusive and binding absent manifest error) that "Adjusted Term CORRA" or "Adjusted Daily Compounded CORRA", as applicable, cannot be determined pursuant to the definition thereof, for reasons other than a Benchmark Transition Event, or
 - (b) the Majority of the Lenders determines that for any reason in connection with any request for a Term CORRA Loan or Daily Compounded CORRA Loan, as applicable, or a conversion thereto or a continuation thereof that Term CORRA or Daily Compounded CORRA, as applicable, for any requested Interest Period with respect to a proposed Term CORRA Loan or Daily Compounded CORRA Loan, as applicable, does not adequately and fairly reflect the cost to such Lenders of making and maintaining such Loan, and the Majority of the Lenders has provided notice of such determination to the Administrative Agent,

the Administrative Agent will promptly so notify the Borrower and each Lender.

(2) Upon delivery of such notice by the Administrative Agent to the Borrower under Section 2.09(1), any obligation of the Lenders to make Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, and any right of the Borrower to continue Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, or to convert Prime Rate Loans to Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, shall be suspended (to the extent of the affected Term CORRA Loans or Daily Compounded CORRA

Loans, as applicable, or affected Interest Periods) until the Administrative Agent (with respect to Section 2.09(1)(b), at the instruction of the Majority of the Lenders) revokes such notice.

Upon receipt of such notice by the Administrative Agent to the Borrower under Section 2.09(1), (i)(x) the Borrower may revoke any pending request for a borrowing of, conversion to or continuation of Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, (to the extent of the affected Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, or affected Interest Periods); (y) in respect of Term CORRA Loans, the Borrower may elect to convert any such request into a request for a Drawdown of or conversion to Daily Compounded CORRA Loans; or, failing such revocation or election, (z) the Borrower will be deemed to have converted any such request into a request for a Drawdown of or conversion to Prime Rate Loans, in the amount specified therein and (ii) (x) in respect of Term CORRA Loans, the Borrower may elect to convert any outstanding affected Term CORRA Loans at the end of the applicable interest Period, into Daily Compounded CORRA Loans, and (y) otherwise, or failing such election, any outstanding affected Term CORRA Loans or Daily Compounded CORRA Loans, as applicable, will be deemed to have been converted, at the end of the applicable Interest Period, into Prime Rate Loans. Upon any such conversion, the Borrower shall also pay accrued interest on the amount so converted, together with any additional amounts required pursuant to Section 6.02.

2.10 Maturity Date

The Borrower will repay the outstanding principal amount of all Loans and all other related Obligations under the Credit Facility on January 26, 2027 (the "Maturity Date") provided that where such date is not a Business Day, the Maturity Date shall be the first Business Day preceding such date. Subject to the provisions of Section 12.02, the Borrower shall repay and there shall become due and payable on the Maturity Date, the Loans under the Credit Facility, and all accrued and unpaid interest thereon and pay such other amounts as are then owing to the Administrative Agent and the Lenders hereunder or under the other Loan Documents.

2.11 Benchmark Replacement Setting

(1) Benchmark Replacement Notwithstanding anything to the contrary herein or in any other Loan Document, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior any setting of the then-current Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (a) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document and (y) if a Benchmark Replacement is determined in accordance with clause (b) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of any Benchmark setting at or after 5:00 p.m. (Toronto time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this

Agreement or any other Loan Document so long as the Administrative Agent has not received, by such time, written notice of objection to such Benchmark Replacement from the Majority of the Lenders. If the Benchmark Replacement is Adjusted Daily Compounded CORRA, all interest payments will be payable on the last day of each Interest Period. No Eligible Hedge Agreement shall be deemed to be a "Loan Document" for purposes of this Section 2.11.

- (2) <u>Conforming Changes</u> In connection with the use, administration, adoption or implementation of a Benchmark Replacement, the Administrative Agent will have the right to make such Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- will promptly notify the Borrower and the Lenders of (i) the implementation of any Benchmark Replacement and (ii) the effectiveness of any such Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. The Administrative Agent will notify the Borrower of (x) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 2.11(4) and (y) the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 2.11 including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 2.11.
- <u>Unavailability of Tenor of Benchmark</u> Notwithstanding anything to the contrary **(4)** herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including Term CORRA) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (B) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

- (5) <u>Benchmark Unavailability Period</u> Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any pending request for a Drawdown of, conversion to or continuation of Loans, which are of the type that have a rate of interest determined by reference to the then-current Benchmark, to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for a Drawdown of or conversion to, (i) for a Benchmark Unavailability Period in respect of Term CORRA, Daily Compounded CORRA Loans, and (ii) for a Benchmark Unavailability Period in respect of a Benchmark other than Term CORRA, Prime Rate Loans.
- (6) <u>Amendments</u> The Administrative Agent and the Borrower may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or to enter into additional Loan Documents as the Administrative Agent deems appropriate in order to implement any Benchmark Replacement or any Conforming Change or otherwise effectuate the terms of this Section 2.11 in accordance with the terms hereof.

2.12 <u>Circumstances Requiring Canadian Dollar Loans</u>

If, in connection with any Term SOFR Loan, any Lender determines in good faith and notifies the Borrower that (i) by reason of circumstances affecting financial markets inside or outside Canada, deposits of U.S. Dollars are unavailable to such Lender; or (ii) any change in Applicable Laws has made it unlawful for such Lender to make or maintain or to give effect to its obligations in respect of Term SOFR Loans as contemplated hereby, then:

- (a) the right of the Borrower to select a Term SOFR Loan shall be suspended from the date of such notice thereof by the Administrative Agent to the Borrower until the relevant Lender determines, and has given notice of such determination to the Administrative Agent and the Borrower, that the circumstances causing the suspension no longer exist (which it shall do as soon as practicable thereafter);
- (b) if any such Term SOFR Loan is not yet outstanding, any applicable request for a Conversion to such Term SOFR Loan shall be suspended from the date of notice thereof by the Administrative Agent to the Borrower until the relevant Lender determines, and has given notice of such determination to the Administrative Agent and the Borrower, that the circumstances causing the suspension no longer exist (which it shall do as soon as practicable thereafter); and
- (c) if any Term SOFR Loan is already outstanding at any time when the right of the Borrower to select a Term SOFR Loan is suspended, it and all other Term SOFR Loans shall be converted automatically into a Prime Rate Loan on the last day of the then current Interest Period applicable thereto (or on such earlier date as may be required to comply with any Applicable Laws) in a principal amount equal to the Equivalent Amount in Canadian Dollars of such Term SOFR Loan determined on the date of Conversion thereof.

2.13 SOFR Benchmark Replacement

Notwithstanding anything to the contrary contained herein, all Term SOFR Loans shall be subject to the terms and conditions of Schedule E.

2.14 <u>CORRA Licensing</u>

- (1) Each Lender hereby (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent and its respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party:
 - (a) it has all necessary licenses and approvals from TMX Datalinx (or any successor licensor of the Term CORRA Reference Rate) required for the use of the Term CORRA Reference Rate with respect to the issue of Canadian Dollar loans and other loan products including the making or advance of Loans or other grants or extensions of credit pursuant to or as otherwise contemplated under this Agreement and the other Loan Documents;
 - (b) without limiting clause (a) above, it has all other necessary permits, approvals or licenses necessary for such Lender to perform its obligations as a Lender under this Agreement and the other Loan Documents; and
 - (c) all such licenses, permits and approvals are in good standing and full force and effect and such Lender has done and will do all things necessary to comply, preserve, renew and keep in full force and such licenses, permits and approvals; and
- (2) the Administrative Agent shall be entitled to rely on the forgoing representations of each Lender and shall have no obligation and shall not be responsible to verify or confirm the existence or validity of any such licenses, permits or approvals of any Lender or to monitor or verify such Lender's compliance with, or the status and standing of, any such license, permits or approvals and the Administrative Agent shall have no liability to the Borrower, any Lender or any other Person or entity for any damages of any kind in connection with the foregoing.

ARTICLE 3- DISBURSEMENT CONDITIONS

3.01 Conditions Precedent to Drawdown

The obligation of the Lenders to make the Drawdown is subject to and conditional upon the prior satisfaction of the following conditions precedent:

(a) the Administrative Agent will have received a Drawdown Notice by the time required under Section 2.04(1) and (1);

- (b) the representations and warranties deemed to be repeated pursuant to Section 8.02 will continue to be true and correct in all material respects as if made on and as of the Drawdown Date;
- (c) no Default or Event of Default will have occurred and be continuing on the Drawdown Date, or would result from making the requested advance;
- (d) a Material Adverse Change will not have occurred and be continuing;
- (e) the Administrative Agent will have received certified copies of the Organizational Documents of each Obligor, the resolutions authorizing the execution and delivery of, and performance of each Obligor's respective obligations under, the Loan Documents (including, without limitation, this Agreement) and the transactions contemplated herein and therein, as applicable, and a certificate as to the incumbency of the officers of the relevant Obligors executing the Loan Documents and any other documents to be provided pursuant to the provisions hereof;
- (f) except as otherwise agreed by the Lenders, certificates of status or comparable certificates for all Relevant Jurisdictions of each of the Borrower and the other Obligors will have been delivered to the Administrative Agent;
- (g) the Lenders will have completed their due diligence with respect to the Obligors and the Property, and will have received all financial, corporate and other information requested by the Lenders, including receipt and review of:
 - (i) (A) Up to date (i) operating statements for the Property for the three month period ending September 30, 2023 (ii) operating statements for the Property for the year ended December 31, 2022, and (iii) rent roll for the Property; (B) the most recent financial statements of the Guarantor (including (i) statements for the three months ending September 30, 2023 and (ii) the annual financial statements required pursuant to Section 9.02(1) for the Fiscal Year ending on December 31, 2022); and (C) 2024 operating budget in respect of the Property together with a one year projection budget and timeline of capital expenditures for the Property;
 - (ii) all Leases and Material Property Agreements (including, without limitation, satisfactory review and approval of the Limited Partnership Agreement), and all Permitted Encumbrances;
 - (iii) a certificate from the City or a copy of the relevant tax bill issued by the City and evidence of payment thereof satisfactory to the Administrative Agent, evidencing that realty Taxes levied against the Lands are current;
 - (iv) a Phase I and, if necessary, a Phase II environmental report (as required by the Phase I report) with respect to the Property, each of which must be addressed to the Administrative Agent or accompanied by a letter from the environmental consultant addressed to the Administrative Agent that permits the Administrative Agent to rely on such reports;

- (v) a property condition assessment with respect to the Property, which must be addressed to the Administrative Agent or accompanied by a letter from the consultant addressed to the Administrative Agent that permits the Administrative Agent to rely on such report;
- (vi) an Appraisal for the Property, the contents of which shall be satisfactory to the Administrative Agent in its sole discretion together with a reliance letter issued by the author thereof permitting the Administrative Agent to rely upon such Appraisal (such letter to be in a form acceptable to the Administrative Agent);
- (vii) if required by any Lender, such Lender shall have completed such site visits and investigations in respect of the Property as it deems appropriate and shall be satisfied with the results of same;
- (viii) the Administrative Agent shall have received the estoppel certificate in form and substance satisfactory to the Lenders, acting reasonably, in respect of the Store Sublease;
- (ix) if available, a current survey and reference plan of the Lands prepared by an accredited surveyor, acceptable to the Administrative Agent, showing the boundaries of the Lands, the location of the buildings and structures situate thereon and the location of encroachments, easements or rights-ofway;
- (x) evidence satisfactory to the Lenders that the Borrower's insurance is satisfactory and complies with this Agreement (and in respect of which the independent insurance consultant retained by the Administrative Agent shall have provided a written report to the Administrative Agent confirming same, together with all insurance certificates);
- (xi) any other documents related to the Property that the Lenders deem necessary; and
- (xii) such financial and other information or documents relating to the Obligors as the Lenders may reasonably require,

and the results of such due diligence will be satisfactory to the Lenders in their sole discretion;

- (h) except as otherwise agreed by the Lenders, the Administrative Agent will have received certified copies of all shareholder approvals and true copies of all regulatory, governmental and other approvals, if any, required in order for the Obligors to enter into this Agreement and the other Loan Documents and to perform their obligations hereunder;
- (i) all releases, discharges and postponements that are required in the discretion of the Lenders (in registrable form where necessary) with respect to all Encumbrances

- affecting the collateral Encumbered by the Security that are not Permitted Encumbrances, if any, will have been delivered to the Administrative Agent;
- (j) the Administrative Agent will have received payment of all fees and expenses (including the fees and disbursements of the Lender's Counsel) payable to the Lenders that are due and payable at such time (provided that invoices/notices have been submitted to the Borrower);
- (k) duly executed copies of the Security and all other Loan Documents and deliveries in connection therewith will have been delivered to the Administrative Agent and all such Loan Documents will have been duly registered, filed and recorded in all Relevant Jurisdictions where required by Applicable Law or where the Lenders considers it necessary, in its sole discretion, to do so;
- (l) the Administrative Agent shall have received one or more currently dated opinions of Obligors' Counsel as to such matters and in such form acceptable to the Administrative Agent and Lender's Counsel addressed to the Administrative Agent and Lender's Counsel, with respect to each Obligor including, without limitation, enforceability, creation and perfection of security interest and non-contravention of the Organizational Documents and Applicable Laws;
- (m) all registrations and filings shall have been made which the Administrative Agent determines to be necessary or advisable to preserve and protect the Security, including, without limitation, obtaining a title insurance policy in favour of the Administrative Agent in form which is satisfactory to the Administrative Agent;
- (n) the Administrative Agent will have received an identity certificate with respect to the Borrower and the other applicable Obligors substantially in the form attached at Schedule C and any applicable documentation required by the Lenders in respect of anti-money laundering requirements in a form acceptable to each Lender;
- (o) the Lenders will have received from the Borrower all required identification and other due diligence materials required to allow the Lenders to comply with their obligations under all applicable anti-money laundering and anti-terrorism laws and regulations to which each Lender may be subject (including, without limitation, the AMLA);
- (p) the Administrative Agent will have received a Compliance Certificate from the Guarantor acceptable to the Lenders;
- (q) the Administrative Agent shall have received all other reports and deliveries required hereunder for the period prior to such Drawdown Date;
- (r) the Drawdown must have occurred no later than January 31, 2024;
- (s) delivery of a true copy of and satisfactory review by the Administrative Agent of the Store Ground Lease and the Head Lease;

- (t) the Borrower shall have delivered the Store Ground Lessor Consent to the Administrative Agent;
- (u) each Lender (on the advice of Lenders' Counsel, as applicable) shall have confirmed its satisfaction with each of the foregoing conditions precedent to the Administrative Agent on or before 11:00 a.m. (Toronto time) of the Business Day prior to the Drawdown detailed in the Drawdown Notice required by Section 3.01(a) (whereby each Lender that has not indicated its satisfaction on or before such time shall be deemed to have been so satisfied); and
- (v) all other terms and conditions of this Agreement upon which the Borrower may obtain a Loan that have not been waived will have been fulfilled in all material respects,

provided that all documents delivered pursuant to this Section 3.01 must be in full force and effect, and in form and substance satisfactory to the Lenders.

3.02 Waiver

The conditions set forth in Section 3.01 are inserted for the sole benefit of the Lenders and may be waived by the Lenders, in whole or in part (with or without terms or conditions), in respect of the Drawdown.

ARTICLE 4 - PAYMENTS OF INTEREST AND COMMITMENT FEES

4.01 Interest on Prime Rate Loans

The Borrower will pay interest on each Prime Rate Loan during each Interest Period applicable thereto in Canadian Dollars at a rate *per annum* equal to the sum of (i) the Prime Rate in effect from time to time during such Interest Period plus (ii) the Prime Rate Margin. Each determination by the Administrative Agent of the Prime Rate applicable from time to time during an Interest Period will, in the absence of manifest error, be binding upon the Borrower. Such interest will be payable in arrears on each Prime Rate Loan Interest Payment Date for such Loan for the period from and including the Drawdown Date, Conversion Date or preceding Prime Rate Loan Interest Payment Date, as the case may be, for such Loan to but excluding such Prime Rate Loan Interest Payment Date (or, if such Prime Rate Loan Interest Payment Date follows the repayment of such Loan or the Conversion of such Loan, to but excluding the date of such repayment or Conversion) and will be calculated on the principal amount of the Prime Rate Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 days or 366 days, as the case may be. Changes in the Prime Rate will cause an immediate adjustment of the interest rate applicable to such Loan without the necessity of any notice to the Borrower.

4.02 Interest on CORRA Loans

- The Borrower will pay interest on each Term CORRA Loan during each Interest Period applicable thereto in Canadian Dollars at a rate per annum equal to the sum of (a) Adjusted Term CORRA in effect from time to time during such Interest Period, plus (b) the CORRA Margin. Any amount of principal of, or interest on, any such Term CORRA Loan which is not paid when due (whether at stated maturity, by acceleration or otherwise) shall be payable on demand and shall bear interest (both before and after judgment), from the date on which such amount is due until such amount is paid in full, at a rate per annum equal to Adjusted Term CORRA in effect from time to time plus the CORRA Margin. Accrued interest on each Term CORRA Loan will be (a) calculated daily and payable in arrears on (i) each Interest Payment Date for such Term CORRA Loan for the period from and including the Drawdown Date, Conversion Date, Rollover Date or preceding Interest Payment Date, as the case may be, for such Term CORRA Loan to but excluding such Interest Payment Date (or, if such Interest Payment Date follows the repayment of such Term CORRA Loan or the Conversion of such Term CORRA Loan, to but excluding the date of such repayment or Conversion), and (ii) on the day on which such Term CORRA Loan becomes due and payable in full pursuant to the provisions hereof; and (b) calculated on the principal amount of such Term CORRA Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 days.
- The Borrower will pay interest on each Daily Compounded CORRA Loan during (2) each Interest Period applicable thereto in Canadian Dollars at a rate per annum equal to the sum of (a) Adjusted Daily Compounded CORRA in effect from time to time during such Interest Period, plus (b) the CORRA Margin. Any amount of principal of, or interest on, any such Daily Compounded CORRA Loan which is not paid when due (whether at stated maturity, by acceleration or otherwise) shall be payable on demand and shall bear interest (both before and after judgment), from the date on which such amount is due until such amount is paid in full, at a rate per annum equal to Adjusted Daily Compounded CORRA in effect from time to time plus the CORRA Margin. Accrued interest on each Daily Compounded CORRA Loan will be (a) calculated daily and payable in arrears on (i) each Interest Payment Date for such Daily Compounded CORRA Loan for the period from and including the Drawdown Date, Conversion Date, Rollover Date or preceding Interest Payment Date, as the case may be, for such Daily Compounded CORRA Loan to but excluding such Interest Payment Date (or, if such Interest Payment Date follows the repayment of such Daily Compounded CORRA Loan or the Conversion of such Daily Compounded CORRA Loan, to but excluding the date of such repayment or Conversion), and (ii) on the day on which such Daily Compounded CORRA Loan becomes due and payable in full pursuant to the provisions hereof; and (b) calculated on the principal amount of such Daily Compounded CORRA Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 days.

4.03 Interest on Term SOFR Loans

The Borrower will pay interest on each Term SOFR Loan during each Interest Period applicable thereto in U.S. Dollars at a rate *per annum* (expressed on the basis of a 360 day year) equal to the sum of (a) the Adjusted Term SOFR plus (b) the SOFR Margin, in each case, from time to time during such Interest Period. Each determination by the Administrative Agent

of the Adjusted Term SOFR with respect to an Interest Period applicable from time to time during such Interest Period will, in the absence of manifest error, be binding upon the Borrower. Such interest will be payable in arrears on each Interest Payment Date for such Term SOFR Loan for the period from and including the Conversion Date, Rollover Date or preceding Interest Payment Date, as the case may be, for such Term SOFR Loan to but excluding such Interest Payment Date and will be calculated on the principal amount of the Term SOFR Loan outstanding during such period and on the basis of the actual number of days elapsed divided by 360. Changes in the Adjusted Term SOFR will result in an immediate adjustment of the interest rate applicable to such Term SOFR Loan without the necessity of any notice to the Borrower. The Administrative Agent shall provide notice of such adjustment to the Borrower as soon as practical thereafter.

4.04 Intentionally Omitted

4.05 <u>Maximum Rate of Interest</u>

Notwithstanding anything contained herein to the contrary, the Borrower will not be obliged to make any payment of interest or other amounts payable to the Lenders hereunder in excess of the amount or rate that would be permitted by Applicable Law or would result in the receipt by the Lenders of interest at a criminal rate (as such terms are construed under the *Criminal Code* (Canada)). If the making of any payment by the Borrower would result in a payment being made that is in excess of such amount or rate, the Lenders will determine the payment or payments that are to be reduced or refunded, as the case may be, so that such result does not occur.

4.06 Upfront and Other Fees

In consideration of the Lenders providing the Credit Facility pursuant to the terms of this Agreement, the Borrower will pay, on the date of the Drawdown, the Administrative Agent for the account of the Lenders (each as to their Rateable Portion, an upfront fee in the amount of Cdn. \$101,250 (being 0.045% per annum of the Credit Facility Commitment). The Borrower shall pay such other fees in amounts, and on the terms and conditions, agreed to in writing with the Administrative Agent and the Borrower. All such written arrangements will constitute a Loan Document hereunder.

ARTICLE 5 – INTENTIONALLY OMITTED

ARTICLE 6- REPAYMENT

6.01 Mandatory Repayment

(1) The Borrower shall repay in full all of the Loans, accrued and unpaid interest and all other Obligations in full under the Credit Facility on the earlier of (i) the Maturity Date, or (ii) subject to the provisions of Sections 6.01(2) and 9.03 below, the Disposition of the Property (or any material portion thereof), or the financing or refinancing of the Property (or any portion thereof) (including mortgages, mortgage bonds or other secured credit facility).

- (2) The Borrower shall repay Loans outstanding under the Credit Facility from the following amounts (on the express understanding and agreement that any such repayment(s) shall result in a corresponding permanent reduction in the Commitment:
 - (a) 100% of net proceeds of property insurance in respect of the Property, if not required to be applied towards replacement, restoration or rebuilding of same, on the terms and conditions set out in Section 9.01(8); and
 - (b) 100% of the net proceeds of any financing or Disposition of the Property (or any portion thereof) (which amount, together with any additional amount paid by the Borrower, shall be not less than the outstanding amount of the Credit Facility (which shall be cancelled in conjunction therewith)).

Repayment Compensation

In the event of (a) the payment of any principal of any Term CORRA Loan, Daily Compounded CORRA Loan or Term SOFR Loan, as applicable, prior to the last day of an Interest Period (including as a result of an Event of Default), (b) the conversion of any Term CORRA Loan, Daily Compounded CORRA Loan or Term SOFR Loan, as applicable, other than on the last day of the Interest Period, (c) the failure to borrow, convert, continue or prepay any Loan on the date specified in any notice delivered pursuant hereto, or (d) the assignment of any Term CORRA Loan, Daily Compounded CORRA Loan or Term SOFR Loan, as applicable, other than on the last day of the Interest Period, then, in any such event, the Borrower shall, after receipt of a written request by any Lender affected by any such event (which request shall set forth in reasonable detail the basis for requesting such amount), compensate each Lender for the loss, cost and expense attributable to such event. A certificate of any Lender setting forth any amount or amounts that such Lender is entitled to receive pursuant to this Section delivered to the Borrower shall be presumptively correct absent manifest error. The Borrower shall pay such Lender the amount shown as due on any such certificate within ten (10) days after receipt of such demand.

6.03 Intentionally Omitted

6.04 Voluntary Prepayments

The Borrower may at any time and from time to time prepay without penalty (subject to Section 6.02) to the Administrative Agent for the account of the Lenders, the whole or any part of any Loan together with accrued interest thereon to the date of such prepayment provided that:

- (a) the Borrower shall give a Repayment Notice to the Administrative Agent at least three (3) Business Days prior to the prepayment date;
- (b) such prepayment shall be a permanent reduction of the Credit Facility; and
- (c) each such prepayment shall be in a minimum amount of the lesser of (i) Cdn. \$500,000, and (ii) the principal amount of all Loans outstanding immediately prior to

such prepayment, and any such prepayment in excess of Cdn. \$500,000 shall be in increments of Cdn. \$100,000, and integral multiples thereof.

6.05 Excess Amount

Following a daily calculation, if the Administrative Agent determines that on any day as a result of currency fluctuations the aggregate of (a) Loans in Canadian Dollars then outstanding under the Credit Facility, and (b) the Equivalent Amount in Canadian Dollars of Loans in U.S. Dollars then outstanding under the Credit Facility, on such day exceeds the Credit Facility Commitment by greater than 2% (in each case, the amount of the excess being referred to herein as an "Excess Amount"), the Administrative Agent will notify the Borrower that such an event has occurred, and the Borrower will within five (5) Business Days following receipt of such notice (and provided that it has not received a subsequent notice from the Administrative Agent that as a result of further currency fluctuation within the previous five (5) Business Days such Excess Amount no longer exists) pay to the Administrative Agent the Excess Amount, which amount the Administrative Agent shall, at the Borrower's option, either (x) apply to repay Loans and other Obligations under the Credit Facility in an amount equal to the Excess Amount, or (y) hold in an interest bearing cash collateral account as security for the Obligations (and subject to the security interests created by the Security) and apply in payment of same as they mature, with the interest thereon paid to the Borrower unless an Event of Default is then continuing, in which case such interest shall be applied by the Administrative Agent in accordance with the terms hereof, provided that if at any time there is no longer an Excess Amount, as determined by the Administrative Agent, any funds held in the cash collateral account as set forth in this clause (v) shall be returned to the Borrower, unless an Event of Default is then continuing.

ARTICLE 7- PLACE AND APPLICATION OF PAYMENTS

7.01 Place of Payment of Principal, Interest and Fees

All payments of principal, interest, fees and other amounts to be made by the Borrower to the Administrative Agent for the account of the Lenders pursuant to this Agreement shall be made for value on the day such amount is due and, if such day is not a Business Day, on the Business Day next following, by deposit or transfer thereof to the account of the Administrative Agent designated by the Administrative Agent for such purpose or at such other place as the Administrative Agent and the Borrower may from time to time agree.

7.02 Funds

Each amount advanced, disbursed or paid hereunder shall be advanced, disbursed or paid, as the case may be, in such form of funds as may from time to time be customarily used for Canadian Dollars or U.S. Dollars, as applicable, in Toronto, Ontario, in the settlement of banking transactions similar to the banking transactions required to give effect to the provisions of this Agreement on the day such advance, disbursement or payment is to be made, and, in any event, subject to the provisions of Section 2.04.

ARTICLE 8 - REPRESENTATIONS AND WARRANTIES

8.01 Representations and Warranties

The Borrower represents and warrants to the Administrative Agent and each of the Lenders as follows, and acknowledges and confirms that the Administrative Agent and each of the Lenders are relying upon such representations and warranties:

- (1) Existence and Qualification Each Obligor (i) that is a corporation or company has been duly incorporated, amalgamated or continued, as the case may be, and is validly subsisting as a corporation or company under the laws of its jurisdiction of incorporation, amalgamation, or continuance, as the case may be, (ii) that is not a corporation or company has been duly created or established as a partnership or other entity and validly exists under the laws of the jurisdiction in which it has been created or established, and (iii) is duly qualified to carry on business in all jurisdictions in which it carries on its business and has all Material Licences required to conduct such business.
- (2) <u>Power and Authority</u> Each Obligor has the power, authority and right (i) to enter into and deliver, and to exercise its rights and perform its obligations under, the Loan Documents to which it is a party and all other instruments and agreements delivered by it pursuant to any of the Loan Documents, and (ii) to own its property and carry on its business as currently conducted and as currently proposed to be conducted by it. Without limiting the foregoing, each of the Obligors has all necessary power and authority to own its respective interest in the Property.
- (3) Execution, Delivery, Performance and Enforceability of Documents The execution, delivery and performance of each of the Loan Documents to which any Obligor is a party, and every other instrument or agreement delivered by it pursuant to any Loan Document, has been duly authorized by all actions, if any, required on its part and by its directors (or where applicable partners, members or managers), and each of such documents has been duly executed and delivered and, to the best of the knowledge of each Obligor, as applicable, constitutes a valid and legally binding obligation of the particular Obligor enforceable against it in accordance with its terms, subject to bankruptcy, insolvency, reorganization, arrangement, winding-up, moratorium and other similar laws of general application limiting the enforcement of creditors' rights generally and to general equitable principles.
- (4) <u>Loan Documents Comply with Applicable Laws, Organizational Documents and Contractual Obligations</u> Neither the entering into nor the delivery of, and neither the consummation of the transactions contemplated in nor compliance with the terms, conditions and provisions of, the Loan Documents by any Obligor conflicts with or will conflict with, or results or will result in any breach of, or constitutes a default under or contravention of, any Requirements of Law applicable to it or any of its, or if applicable, its general partner's or partners', Organizational Documents, or results or will result in the creation or imposition of any Encumbrance other than Permitted Encumbrances except in favour of the Administrative Agent upon any of its property, including, in the case of the Owner Parties, the Property.
- (5) <u>Consents Respecting Loan Documents</u> Each Obligor has obtained, made or taken all consents, approvals, authorizations, declarations, registrations, filings, notices and other

actions whatsoever required as to the date hereof in connection with the execution and delivery by it of each of the Loan Documents to which it is a party and the consummation of the transactions contemplated in the Loan Documents.

- Taxes Each of the Obligors has paid or made adequate provision for the payment (6)of all Taxes levied on it or on its property (including the Property) or income that are due and payable, including interest and penalties, or has accrued such amounts in its financial statements for the payment of such Taxes, except Taxes that are not material in amount or that are not delinquent (or if delinquent are being contested in good faith, and in respect of which nonpayment would not individually or in the aggregate constitute, or be reasonably likely to cause, a Material Adverse Change, and, if the aggregate amount of same is in excess of Cdn. \$100,000, in respect of which the Obligors have deposited with the Administrative Agent or the appropriate Governmental Authority collateral satisfactory to the Administrative Agent or such Governmental Authority, as the case may be, to secure the payment of such Taxes and so long as the Administrative Agent is satisfied that its Security is not in jeopardy), and there is no material action, suit, proceeding, investigation, audit or claim now pending, or to its knowledge threatened, by any Governmental Authority regarding any Taxes that is reasonably likely to cause a Material Adverse Change nor has it agreed to waive or extend any statute of limitations with respect to the payment or collection of Taxes.
- (7) <u>Judgments, Etc.</u> No Obligor is subject to any judgment, order, writ, injunction, decree or award, or to any restriction, rule or regulation (other than customary or ordinary course restrictions, rules and regulations consistent or similar with those imposed on other Persons engaged in similar businesses) that has not been stayed or of which enforcement has not been suspended and that individually or in the aggregate constitutes, or is reasonably likely to cause, a Material Adverse Change.
- (8) Absence of Litigation There are no actions, suits or proceedings pending or, to the best of the Obligors' knowledge and belief, after due inquiry and all reasonable investigation, threatened against or affecting any Obligor that are reasonably likely to cause, either separately or in the aggregate, a Material Adverse Change. None of the Obligors is in default with respect to any Applicable Law in a manner or to an extent that could reasonably be expected to cause a Material Adverse Change.
- (9) <u>Title to Lands</u> The Nominee is the registered owner and the Borrower is the beneficial owner of the leasehold interest in and to the Lands, in each case with good and marketable leasehold title thereof, and any other real and personal property of any nature which is part of the Property, in each case free and clear of all Encumbrances except Permitted Encumbrances, and no Person has any agreement or right to acquire an interest in the Property, save and except pursuant to Permitted Encumbrances.
- (10) <u>Labour Relations</u> No Owner Party is engaged in any unfair labour practice that could reasonably be expected to cause a Material Adverse Change, and there is no unfair labour practice complaint pending against any Owner Party or, to the best of their knowledge, threatened against any Owner Party, before any Governmental Authority that if adversely determined could reasonably be expected to cause a Material Adverse Change. No grievance or arbitration arising out of or under any collective bargaining agreement is pending against any

Owner Party or to the best of the Owner Parties' knowledge, threatened against either of them that are reasonably likely to cause a Material Adverse Change. To the best of the Borrower's knowledge, no strike, labour dispute, slowdown or stoppage is pending against any Owner Party or, to the best of its knowledge, threatened against either of them and no union representation proceeding is pending with respect to any employees of any Owner Party, except (with respect to any matter specified in this sentence, either individually or in the aggregate) such as could not reasonably be expected to cause a Material Adverse Change.

- Compliance with Laws To the best of the knowledge of the Borrower, none of the Obligors is in default under any Applicable Law where such default could reasonably be expected to cause a Material Adverse Change. To the best of the knowledge of the Borrower, the Property is in compliance in all material respects with all Applicable Law, including, without limitation, all Environmental Law where non-compliance could reasonably be expected to cause a Material Adverse Change. Further, to the best of the knowledge of the Obligors, there are no facts known or which ought reasonably to be known which could give rise to a notice of non-compliance with any Applicable Law where such non-compliance could reasonably be expected to cause a Material Adverse Change.
- (12) <u>Changes to Applicable Law</u> To the best of the knowledge of the Borrower, there are no pending or proposed changes to Applicable Law which would render illegal or materially restrict the operation of the Property.
- (13) No Default No Default or Event of Default has occurred and is continuing. No Obligor is in violation of, or in default under, any agreement, mortgage, franchise, licence, judgment, decree, order, statute, rule or regulation which is material to its or their interests in the Property which violation or default is reasonably likely to constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change nor will execution, delivery and performance of this Agreement, the Loan Documents to which they are a party or any of the agreements provided for or contemplated hereby result in any such violation.
- (14) <u>Ownership Structure</u> The ownership structure of the Borrower and the Property is as set out in Schedule 8.01(14).
- (15) <u>Relevant Jurisdictions</u> The Relevant Jurisdictions for each Obligor (other than the Nominee) is Ontario. The Relevant Jurisdiction for the Nominee is British Columbia.
- (16) Security To the best of the knowledge of the Borrower, the Security is effective to create in favour of the Administrative Agent as security for the Obligations described therein, a legal, valid, binding and enforceable security interest in the collateral described therein and the proceeds thereof, subject however to any applicable bankruptcy, insolvency, re-organization, arrangement, winding-up, moratorium and other similar laws of general application limiting the enforcement of creditors' rights generally and to general equitable principles.
 - (17) Material Property Agreements, Material Licences and Permitted Encumbrances
 - (a) True copies of each of the Material Property Agreements and Material Licences existing as of the date of this Agreement have been delivered to the Administrative Agent.

- (b) With respect to each of the Material Property Agreements: (A) it is in full force and effect and has not, except as has been disclosed to (and consented to, where required hereunder) the Administrative Agent, been amended; (B) to the extent an Obligor is a party thereto, it has been duly executed and delivered and constitutes a valid and binding obligation of such Obligor; and (C) no Obligor has received any notice or claim of a default or breach which is reasonably likely to constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change.
- (c) With respect to each of the Material Property Agreements and each of the Permitted Encumbrances: (A) all obligations and covenants, the non-compliance with which is reasonably likely to constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change, required to be met or complied with on the part of any Obligor have been complied with and with respect to any other party thereto have been met or complied with to the best of its knowledge; and (B) no default or event, which with the giving of notice or a lapse of time or both would constitute a default on the part of any Obligor exists which is reasonably likely to constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change.
- (d) No event has occurred and is continuing that would constitute a material breach of or a material default under any Material Property Agreement or Material Licence, and each Material Property Agreement to which any Obligor is a party is binding upon it and, to the best of the Obligors' knowledge, is a binding agreement of each other Person who is a party thereto.
- (18) <u>Work Orders</u> To the best of the knowledge of the Borrower, there are no outstanding judgments, writs of execution, seizures, injunctions or directives against any Obligor, nor any work orders or directives or notices of deficiency capable of resulting in work orders or directives, with respect to the Property which constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change or that are reasonably likely to constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change on the ability of such Obligor to perform its obligations under this Agreement or any Loan Document.
- (19) <u>Condition of Property</u> To the best of the knowledge of the Borrower, all buildings and improvements comprising part of the Property are in good physical condition, and there are no material defects or extraordinary repairs required in connection therewith except as disclosed in writing to, and approved by, the Administrative Agent.
- (20) <u>Financial Statements</u> All of the financial statements that have been furnished to the Administrative Agent and the Lenders in connection with this Agreement are complete in all material respects and such financial statements fairly present the financial position of each applicable Obligor as of the dates referred to therein and have been prepared in accordance with GAAP. The Guarantor does not have any liabilities (contingent or other) or other obligations of the type required to be disclosed in accordance with GAAP, that are not fully disclosed in its financial statements provided to the Administrative Agent.

(21) No Material Adverse Change

- (a) Since the date of each Obligor's most recent annual financial statements and the Property's most recent annual operating statements provided to the Administrative Agent in connection with this Agreement, there has been no condition (financial or otherwise), event or change in the Guarantor's business, liabilities, operations, results of operations, assets or prospects which constitutes, or could reasonably be expected to constitute, or cause, a Material Adverse Change; and
- (b) Since the date of the most recent financial statements of the Property provided to the Administrative Agent, there has been no condition (financial or otherwise), event or change, in any Obligor's business, liabilities, operations, results of operations, assets or prospects which constitutes, or could reasonably be expected to constitute, or cause, a Material Adverse Change.

(22) Environmental Matters

- (a) To the best of the knowledge of the Borrower, the Property is in full compliance in all material respects with all Environmental Law; the Borrower is not aware of, nor has it received notice of, any past, present or future condition, event, activity, practice or incident that may interfere with or prevent the material compliance or continued material compliance of the Property or the Borrower in all respects with all Environmental Law; and the Borrower has obtained all material licences, permits and approvals in connection with the Property that are currently required under all Environmental Law and are in material compliance with the provisions of such licences, permits and approvals.
- (b) Other than as disclosed in the environmental report(s) delivered by the Borrower to the Administrative Agent pursuant to Section 3.01(g)(iv), the Borrower is not aware that any Hazardous Substances exist on, about or within, or have been used, generated, stored, transported, disposed of on, or Released from the Lands other than in material accordance and compliance with all Environmental Law.
- (c) The use that the Borrower has made of the Lands will not result in the use, generation, storage, transportation, accumulation, disposal, or Release of any Hazardous Substances on, in or from the Lands except in material accordance and compliance with all Environmental Law.
- (d) There is no action, suit or proceeding, or, to its knowledge, any investigation or inquiry, before any Governmental Authority pending or, to its knowledge, threatened against the Borrower with respect to the Property relating in any way to any Environmental Law that would or could reasonably be expected to cause a Material Adverse Change.
- (e) Other than as disclosed in the environmental report(s) delivered by the Borrower to the Administrative Agent pursuant to Section 3.01(g)(iv), the Borrower has not (i) with respect to the Lands, incurred any current and outstanding liability for any clean-up or remedial action under any Environmental Law with respect to current or past operations, events, activities, practices or incidents relating thereto, (ii) received any

material outstanding written request for information (other than information to be provided in the normal course in connection with applications for licences, permits or approvals) by any Person under any Environmental Law with respect to the condition, use or operation of the Property, (iii) received any outstanding written notice or claim under any Environmental Law with respect to any material violation of or liability under any Environmental Law or relating to the presence of Hazardous Substance on or originating from the Lands, that, in any such case, would or could reasonably be expected to cause a Material Adverse Change, or (iv) ever been convicted of an offence or subjected to any judgment, injunction or other proceeding for noncompliance with any Environmental Law with respect to the Lands or been fined or otherwise sentenced or settled such prosecution or other proceeding short of conviction for non-compliance with any Environmental Law with respect to the Lands.

- (f) Copies of all material analysis and monitoring data for soil, ground water, surface water and the like and reports pertaining to any environmental assessments/audits, including without limitation any inspections, investigations and tests, relating to the Lands that were obtained, are in the possession or control of, or were carried out on behalf of, the Borrower have been delivered to the Administrative Agent.
- (g) Since acquiring its interests in the Lands, the Borrower has maintained all environmental and operating documents and records relating to the Lands substantially in the manner and for the time periods required in all material aspects by Environmental Law and Material Licences.
- (h) The Borrower has not defaulted in reporting in any material respect to any applicable Governmental Authority in relation to the Lands on the happening of an occurrence which it is or was required by any Environmental Law to report.
- (23) <u>Material Licences</u> All Material Licences from third parties and Governmental Authorities, being as of the date hereof those listed in Schedule 1.01(E), have been obtained.
 - (24) Zoning, Uses and Expropriation
 - (a) Except as disclosed in writing to and approved by the Administrative Agent, the Property is zoned to permit the current operation of the Property.
 - (b) The existing uses of the Property comply in all material respects with all Applicable Law.
 - (c) It has not received notice of any proposed rezoning of all or any part of the Property that would be reasonably likely to cause a Material Adverse Change in respect of the operation of the Property or otherwise.
 - (d) It has not received notice of any expropriation of all or any part of the Lands.
- (25) <u>Undisclosed Liabilities</u> There are no liabilities (including contingent liabilities) that, in the aggregate, are material in respect of the Property, which have not been previously

disclosed in writing to and approved by the Administrative Agent. There are no liabilities (including contingent liabilities) that, in the aggregate, are material in respect of the Obligors, or their respective businesses, which under GAAP should have been disclosed as of the date of their respective last financial statements and have not been disclosed in such financial statements.

- (26) <u>Liens</u> The Borrower has not received notice of any Encumbrances (including, without limitation, of any claims for construction liens with respect to work or services performed or materials supplied) related to the Property, other than Permitted Encumbrances.
- (27) <u>Insolvency</u> None of the Obligors (i) has committed any act of bankruptcy, (ii) is insolvent, or has proposed, or given notice of its intention to propose, a compromise or arrangement to its creditors generally, (iii) has made any petition for a receiving order in bankruptcy filed against it, made a voluntary assignment in bankruptcy, taken any proceeding with respect to any compromise or arrangement, taken any proceeding to have itself declared bankrupt or wound-up, taken any proceeding to have a receiver appointed of any part of its assets, has had any Encumbrancer take possession of its property, or (iv) has had an execution or distress become enforceable or become levied on, in the case of the Borrower, any or, in the case of any other Obligor, any material portion, of its assets and property.
- (28) <u>No Infringement</u> To the best of the knowledge of the Borrower after due enquiry, the operation of the Property does not infringe and will not infringe upon any patents, trademarks, trade names, service marks or copyrights, domestic or foreign, or any other industrial property or intellectual property of any other Person, which infringement would constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change.
- (29) <u>Setbacks</u> To the best of the knowledge of the Borrower, the location of any buildings on the Property are within the boundary lines of the Property as a whole and are in compliance with all applicable setback requirements.
- (30) <u>Full Disclosure</u> All information provided or to be provided to the Administrative Agent and the Lenders in connection with the Credit Facility is, to the Obligors' knowledge, true and correct in all material respects and none of the documentation furnished to the Administrative Agent and the Lenders by or on behalf of it, to its knowledge, omits or will omit as of such time, a material fact necessary to make the statements contained therein not misleading in any material way, and all expressions of expectation, intention, belief and opinion contained therein were honestly made on reasonable grounds after due and careful inquiry by it (and any other Person who furnished such material on behalf of it).
- (31) <u>Residency</u> Each Owner Party is not a non-resident for the purposes of Section 116 of the *Income Tax Act* (Canada).

(32) Store Ground Lease

(a) The Store Ground Lease or a notice regarding the Store Ground Lease has been duly registered on title to the Lands and, to the best of the Borrower's knowledge, there have been no amendments or modifications to the terms of the Store Ground Lease since its registration, except by written instruments, all of which have been disclosed to the Administrative Agent.

- (b) The Store Ground Lease is not subject to any Encumbrance superior to, or of equal priority with, the Debenture (including, without limitation, the rights of any fee mortgage, if any) other than the Store Ground Lessor's related sublandlord interest and any Permitted Encumbrances.
- (c) The Store Ground Lease is in full force and effect and no default by the tenant thereunder or, to the best of the Borrower's knowledge, by the Store Ground Lessor, has occurred, nor to the best of the Borrower's knowledge is there any existing condition which, but for the passage of time or giving of notice, would result in a default by the Borrower, as tenant thereunder or under the terms of the Store Ground Lease.

(33) Leases

- (a) Except as disclosed to the Administrative Agent, all Leases are in full force and effect.
- (b) The landlord under each of the Leases is not in material default or in material breach of any of its obligations thereunder, the landlord thereunder has not received from any tenant a notice of a material default or material breach by the landlord of any of its obligations thereunder and there is no event or condition existing, with or without the giving of notice or lapse of time or both, which would entitle any tenant thereunder to terminate any of its obligations in respect thereof.
- (c) Except as disclosed to the Administrative Agent in writing, there are no Leases with the federal government or agents thereof.

8.02 <u>Survival and Repetition of Representations and Warranties</u>

The representations and warranties set out in this Article shall survive the execution and delivery of this Agreement and the making of the Drawdown and each Conversion or Rollover.

ARTICLE 9 - COVENANTS

9.01 <u>Borrower Positive Covenants</u>

So long as any amount payable hereunder is outstanding or the Credit Facility is available hereunder, the Borrower covenants and agrees with the Administrative Agent and each of the Lenders that it shall:

(1) <u>Timely payment</u> Make due and timely payment of the Obligations required to be paid by it and punctually perform its other Obligations hereunder and under the other Loan Documents to which it is a party at the time and place and in the manner provided hereunder and under the other Loan Documents.

- (2) <u>Conduct of Business, Maintenance of Existence, Compliance with Laws</u> Engage in business of the same general type as now conducted by it; carry on and conduct its business and operations in a proper, efficient and businesslike manner, in accordance with good business practice; subject to Section 9.03(2), preserve, renew and keep in full force and effect its and the Borrower's existence; and take all reasonable action to maintain all rights, privileges and franchises necessary in the normal conduct of its business and to comply in all material respects with all Material Property Agreements, Material Licences and Requirements of Law, including Requirements of Environmental Law.
- (3) <u>Further Assurances</u> Use reasonable commercial efforts to provide the Administrative Agent with such other documents, opinions, consents, acknowledgments and agreements as are reasonably necessary to implement this Agreement and the other Loan Documents from time to time.
- (4) Access to Information Promptly provide the Administrative Agent with all information reasonably requested by it from time to time at reasonable intervals in connection with this Agreement concerning its financial condition and the Property (including, without limitation, the Leases, Material Property Agreements and Material Licences), and during normal business hours and from time to time at reasonable intervals upon reasonable notice, permit representatives of the Administrative Agent to inspect the Property and to examine its financial records regarding (and only regarding) the Property, and to discuss its financial condition with its senior officers and its auditors, the reasonable expense of all of which will be paid by the Borrower.
- (5) Obligations and Taxes Pay or discharge, or cause to be paid or discharged, before the same will become delinquent (i) all Taxes imposed upon it or upon its income or profits or in respect of its business or property (including the Property) and file all tax returns in respect thereof, (ii) all lawful claims for labour, materials and supplies, and (iii) all required payments under any of its Indebtedness; provided, however that it will not be required to pay or discharge or to cause to be paid or discharged any such amount so long as the validity or amount thereof is being contested in good faith by appropriate proceedings and an appropriate financial reserve in accordance with GAAP and satisfactory to the Administrative Agent has been established.
- (6) <u>Use of Credit Facility</u> Use the proceeds of the Credit Facility only for the purposes specified in Section 2.02.

(7) Insurance

For so long as any amounts are due hereunder:

- (a) It shall maintain or cause to be maintained:
 - A. all risks insurance (including the perils of flood, earthquake and sewer backup) on property of the Borrower or any other Obligor (or property for which they are obliged to insure pursuant to Leases or other agreements relating to the Property) of every description located in or on the Property on a replacement cost basis without a requirement to rebuild on the same

site and not subject to any co-insurance provisions or a stated-amount coinsurance clause shall apply;

- B. boiler and machinery insurance on a comprehensive basis on all machinery and equipment of the Borrower or any other Obligor or for which they may be liable located at the Property, on a replacement cost basis;
- C. business interruption insurance under the property and boiler and machinery insurance policies referred to in Section 9.01(7)(a)A and B, adequate to reimburse all lost revenues relating to the Property for a term of not less than 18 months;
- D. commercial general and umbrella liability insurance, including insurance against claims for personal injury, death, property damage or other loss arising out of the business of the Borrower or any other Obligor or the operation of the Property and extended to include coverage for contractual liability, contingent employer's liability, and liability in respect of collapse and explosion with a minimum limit of liability for any one occurrence of Cdn. \$25,000,000;
- E. such other insurance as may be required to meet the obligations of the Borrower or any other Obligor under any of the Material Property Agreements; and
- F. such other insurance as the Administrative Agent upon consultation with the independent insurance consultant may reasonably require from time to time;

all with insurance companies having a Best's B++ rating at time of placement and at all times thereafter with such insurance companies having comparable claims paying ability as approved by the Administrative Agent acting reasonably. Such insurance is to be in such form and amounts and with such deductibles as are customary in the case of owners of properties similar to the Property and in any event as are acceptable to the Administrative Agent.

- (b) The all-risk, boiler and machinery and business interruption insurance policies referred to in Section 9.01(7)(a) shall:
 - A. name the Borrower and the relevant other Obligors as first named insureds thereunder and as additional insureds all those required to be named as additional insureds under any of the Material Property Agreements,
 - B. name the Administrative Agent as mortgagee and loss payee and have attached the standard Insurance Bureau of Canada mortgage clause with respect to the all-risk and business interruption policies, and the boiler and machinery underwriters' standard mortgage clause with respect to the machinery insurance policy;

- C. provide that no cancellation or termination thereof or material change therein, for any reason whatsoever, shall take effect unless the insurer concerned has given the Administrative Agent not less than 30 days' prior written notice of such proposed action;
- D. contain a waiver by the insurer or insurers of all rights of subrogation or indemnity or any other claim to which such insurer or insurers might otherwise be entitled against the Administrative Agent;
- E. contain a breach of conditions clause; and
- F. otherwise be in such form as the Administrative Agent shall reasonably require;

and such insurance policies may contain reasonable deductibles per occurrence as approved by the Administrative Agent;

- (c) The third party liability insurance policies referred to in Section 9.01(7)(a) shall:
 - A. name the Administrative Agent as an additional insured to the extent that such policies relate to the Property or the operation thereof;
 - B. provide that no cancellation or termination thereof or material change therein, for any reason whatsoever, shall take effect unless the insurer concerned has given the Administrative Agent not less than 30 days' prior written notice of such proposed action;
 - C. contain a waiver by the insurer of all rights of subrogation or indemnity or any other claim to which the insurer might otherwise be entitled against the Administrative Agent;
 - D. contain a cross-liability clause and a severability of interests clause; and
 - E. otherwise be in such form as the Administrative Agent shall reasonably require.

The Borrower will provide detailed certificates of insurance for all policies required hereunder to be purchased and maintained by the Borrower in form acceptable to the Lenders, provided that the Lenders shall, pursuant and subject to the provisions of Section 9.01(4), be entitled to review at the Borrower's offices or the offices of the Borrower's insurance broker, the underlying policies of insurance relating to such certificates of insurance.

- (8) <u>Proceeds of Insurance</u> With regard to the insurance described in Section 9.01(7), the Permitted Encumbrances and Material Property Agreements having priority over the Security, the following shall apply:
 - (a) So long as no Default or Event of Default has occurred and is continuing, the proceeds of all such insurance (other than workers' compensation insurance, errors

and omissions insurance and third party liability insurance which may be remitted to the Borrower without condition or further action by the Administrative Agent or the Lenders) shall be dealt with as follows:

- (i) If the total amount of such proceeds equals or exceeds Cdn. \$5,000,000, they shall be payable directly into an escrow account of the Borrower (which account shall be held with the Administrative Agent and subject to the security interest created by the Security) to be disbursed by the Administrative Agent against receipts payable in not more than 30 days for expenses incurred in repairing the damage or destruction or replacing property in respect of which the insurance is payable, for release by the Administrative Agent to the Borrower to be applied by the Borrower in repairing the damage or destruction or replacing property in respect of which the insurance is payable upon receipt of:
 - A. an officer's certificate of the Borrower stating that the proceeds of such insurance together with other funds held or arranged by the Borrower are sufficient to fully repair, rebuild or replace the damage or destruction in respect of which the insurance proceeds are payable;
 - B. a letter of undertaking of the Borrower to fully repair, rebuild and replace the damage or destruction in respect of which the insurance proceeds are payable; and
 - C. evidence satisfactory to the Administrative Agent and the Lenders that the proceeds of insurance together with the other funds held or arranged by the Borrower will be sufficient to repair, replace or rebuild the damage or destruction in respect of which the insurance proceeds are payable.
- (ii) The proceeds of any business interruption insurance shall be payable to the Borrower, as its interest may appear, to be applied on account of ongoing obligations of the Borrower hereunder or in respect of the Property as the same fall due from time to time.
- (iii) The proceeds of all insurance held by the Administrative Agent shall, unless and until the same are applied or released to the Borrower as aforesaid, constitute continuing collateral security for the Borrower's obligations and liabilities in respect of amounts outstanding hereunder. The Administrative Agent shall place such funds in interest-bearing term deposits with the interest thereon to accrue to the benefit of the Borrower.
- (iv) If the total amount of such proceeds is less than Cdn. \$5,000,000 they shall be released to the Borrower subject to delivery of the documents set out in Section 9.01(8)(a)(i)A, B and C.
- (b) If an Event of Default has occurred and is continuing:

- (i) The proceeds of all insurance other than workers' compensation insurance, errors and omissions insurance and third party liability insurance shall be payable to the Administrative Agent and subject to the Security, to be applied by it, at its option, in reduction of the amounts outstanding hereunder or released by the Administrative Agent to the Borrower upon receipt of:
 - A. an officer's Certificate of the Borrower stating that the proceeds of such insurance together with other funds held or arranged by the Borrower are sufficient to fully repair, rebuild or replace the damage or destruction in respect of which the insurance proceeds are payable;
 - B. a letter of undertaking of the Borrower to fully repair, rebuild and replace the damage or destruction in respect of which the insurance proceeds are payable; and
 - C. evidence satisfactory to the Administrative Agent and the Lenders that the proceeds of insurance together with the other funds held or arranged by the Borrower will be sufficient to repair, replace or rebuild the damage or destruction in respect of which the insurance proceeds are payable.
- (ii) The proceeds of any business interruption insurance shall be payable to the Administrative Agent to be held by the Administrative Agent as additional security for the payment of all amounts payable hereunder, to be applied on account of ongoing obligations of the Borrower hereunder or in respect of the Property as the same fall due from time to time and, to the extent of any surplus, firstly to arrears of such payments and thereafter, if the Administrative Agent has opted to release proceeds of insurance to the Borrower pursuant to and in accordance with Section 9.01(8)(b)(i), then the balance of the proceeds of business interruption insurance shall be payable to the Borrower, failing which the balance, if any, remaining after application of such proceeds as aforesaid shall be paid to the Administrative Agent as partial prepayment of the Loans.
- (iii) The proceeds of all insurance held by the Administrative Agent shall, unless and until the same are applied or released to the Borrower as aforesaid, constitute continuing collateral security for the Borrower's obligations and liabilities in respect of amounts outstanding hereunder. The Administrative Agent shall place such funds in interest-bearing term deposits with the interest thereon to accrue to the benefit of the Borrower (but subject to the security interest created under the Security).
- (9) <u>Insurance Information</u> It shall provide or cause to be provided to the Administrative Agent and the independent insurance consultant such information relating to the Property or the Loan Documents, as may be reasonably requested and which is within its

possession or control. The reasonable fees and costs of the independent insurance consultant shall be paid for by the Borrower.

- (10) <u>Notice of Default</u> Promptly notify the Administrative Agent of any Default or Event of Default that would apply to it of which it becomes aware, using reasonable diligence, and detailing the steps it is taking, if any, to cure same.
- (11) <u>Notice of Material Adverse Change</u> Promptly notify the Administrative Agent of any Material Adverse Change or any matter which constitutes, or could reasonably be expected to constitute, or cause, a Material Adverse Change that would apply to it of which it becomes aware, using reasonable diligence.
- (12) <u>Notice of Litigation</u> Promptly notify the Administrative Agent on becoming aware of the occurrence of any litigation, dispute, arbitration or other proceeding the result of which if determined adversely would be a judgement or award against it would reasonably be expected to be in an amount in excess of the amount set out in Section 12.01(n) or otherwise result in a Material Adverse Change to it, and from time to time provide the Administrative Agent with all reasonable information requested by the Administrative Agent concerning the status of any such proceeding.
 - (13) Other Notices Promptly notify the Administrative Agent on becoming aware:
 - (a) of any Change of Control;
 - (b) of any labour controversy which would likely constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change;
 - (c) of the occurrence of an event of Force Majeure, describing in reasonable detail the effects of such event on the operation of the Property, the action which the Borrower intends to take to remedy such event and the estimated date when the event of Force Majeure will be remedied and will cease to impair operation of the Property;
 - (d) of the cessation of any event of Force Majeure;
 - (e) of any circumstance of which it has notice or is aware which could result in a material breach of or default or non-performance by any party under the Material Property Agreements or of any condition entitling any party to terminate its obligations thereunder;
 - (f) of any damage to or destruction of any Property that forms part of the Property which might give rise to an insurance claim if the cost of any repairs to or replacement of assets of the Borrower exceeds Cdn. \$1,000,000;
 - (g) of any material instrument related to the Property of which the Borrower has notice or is registered against title to the Property and provide to the Administrative Agent a true copy of such instrument;

- (h) of any threatened expropriation or notice of expropriation with respect to the Lands, such notice to be delivered forthwith upon the Borrower becoming aware of such threatened expropriation or its receipt of notice of such proceedings, and the Borrower hereby covenants and agrees that no such claim shall be compromised or settled without the prior written consent of the Administrative Agent;
- (i) of any non-compliance with Environmental Law relating to the Property, and of any notice, investigation, non-routine inspection or material inquiry by any Governmental Authority in connection with any Environmental Law relating to the Property, in each case which would likely constitute, or could reasonably be expected to constitute, or cause, a Material Adverse Change;
- (j) of the exercise of any tenant right of first offer under the Store Ground Lease or any written communication received by the Borrower alleging default under the Store Ground Lease; and
- (k) of any change in the External Credit Ratings or of any Rating Agency ceasing to provide an External Credit Rating.

(14) Environmental Compliance

- (a) Operate the Property in a manner such that no material obligation, including a clean-up or remedial obligation, will arise under any Environmental Law; provided, however, that if any such claim is made or any such obligation arises, it will immediately satisfy or contest such claim or obligation at its own cost and expense, and promptly notify the Administrative Agent upon learning of (i) the existence of Hazardous Substances located on, above or below the surface of the Lands or contained in the soil or water constituting such land (except those being stored, used, contained or otherwise handled in substantial compliance with Environmental Law), or (ii) the occurrence of any reportable Release of Hazardous Substances into the air, land surface water or ground water that has occurred on or from the Property, or (iii) any other event or occurrence relating to the Property which, in the opinion of the Borrower, is likely to give rise to a notice of non-compliance in any material respect with any Environmental Law.
- (b) Comply, and cause any other party that is acting under its authority to comply, in all material respects with all Environmental Law (including, but not limited to, obtaining any Material Licences or similar authorizations) relating to the Property.
- (c) Use its reasonable commercial efforts not to cause or permit a Release of any Hazardous Substance at, on or under the Property except in compliance with Environmental Law.
- (d) Provide the Administrative Agent with an environmental site assessment/audit report of the Property, or an update of such assessment/audit report (i) upon the written request of the Administrative Agent if in its reasonable opinion there is a concern about the Borrower's compliance (as it relates to the Property) or the Property's compliance in all material aspects with Environmental Law, all in scope, form and

content satisfactory to the Administrative Agent, (ii) if such assessment/audit report has been prepared at the request of or on behalf of any Governmental Authority, or (iii) if an Event of Default relating to an environmental matter has occurred, and the Administrative Agent has made a written request to the Borrower for such an assessment/audit report or update, within thirty (30) Business Days after such request, and all such assessments/audits reports or updates thereof shall be at the Borrower' expense and risk; an environmental site assessment/audit may include, for purposes of this Section, without limitation, any inspection, investigation, test, sampling, analysis, monitoring pertaining to air, land and water relating to the Property reasonably required under the circumstances giving rise to the request for the assessment/audit report.

- (e) Not use the Property, or permit it to be used, to generate, manufacture, refine, treat, transport, store, handle, dispose, transfer, produce or process Hazardous Substances except in compliance with all Environmental Law.
- (f) Maintain in all material respects all environmental and operating documents and records, including, without limitation, Material Licences and orders, relating to the Property in the manner and for the time periods required by Environmental Law.
- (15) Security Provide, or cause to be provided, to the Administrative Agent with the Security required from time to time pursuant to Article 10 in accordance with the provisions of such Article, accompanied by supporting resolutions, certificates and opinions in form and substance satisfactory to the Administrative Agent and Borrower, both acting reasonably, and do all such further acts and execute and deliver all such documents and instruments as may from time to time be reasonably requested by the Administrative Agent, to ensure that the Security constitutes at all times valid, enforceable, and perfected first priority Encumbrances against the Property (subject only to Permitted Encumbrances).
- (16) <u>Maintenance of Property</u> Keep all of its property necessary for the Property in good working order and condition, normal wear and tear excepted, except to the extent that the failure to do so would not individually or in the aggregate be reasonably likely to cause a Material Adverse Change.

(17) Intentionally Omitted

(18) <u>Adequate Books</u> In the case of the Property and financial statements in relation thereto, maintain adequate books, accounts and records in accordance with GAAP.

(19) <u>Material Property Agreements</u>

(a) At all times be and remain in full compliance in all material respects with all of its covenants, agreements and obligations in and diligently enforce all its material rights under all Material Property Agreements. The Borrower shall not alter, amend or waive, in any material respect, any of its rights under or permit any termination or surrender of any Material Property Agreement without the prior written consent of the Administrative Agent (which shall not be unreasonably withheld) unless such alterations, amendments, waivers, terminations or surrenders, as applicable, reflect, in

- all material respects, good business practice, are in the ordinary course of business, and such material terms as a prudent owner of a similar property would accept having regard to all relevant factors at the time.
- (b) Advise the Administrative Agent in writing of all new Material Property Agreements (or any material amendments of existing Material Property Agreements) entered into forthwith following the entering into thereof and shall deliver forthwith a copy thereof to the Administrative Agent. The Borrower shall provide written notice to the Administrative Agent of any assignment made by a contracting party to a Material Property Agreement.
- (20) Access Permit the Administrative Agent (through their agents, officers or employees), for the purposes of monitoring compliance with the covenants and obligations of the Borrower hereunder, at their risk, to visit and inspect the Property, to conduct tests, measurements and surveys in relation to the Property, provided that such tests, measurements and surveys are conducted in accordance with prudent industry practice and Applicable Law and/or are required as a result of the reasonable concerns of the Administrative Agent as to non-compliance with such covenant and obligation, and to be advised as to the same by the officers, engineers and advisers of the Borrower (or such other Persons as may be appropriate), all at such reasonable times and intervals as the Administrative Agent may desire upon reasonable prior notice and in the presence of the Borrower if it so desires. Such visits, inspections, measurements, reviews and tests shall be at the cost of the Borrower, provided such expenses are reasonably incurred. Any such visit, inspection, examination, discussion or tests shall not be deemed to be supervision, charge, management, control or occupation by the Administrative Agent for purposes of any environmental or other liabilities.
- (21) <u>Peaceable Entry</u> From and after the occurrence of an Event of Default and for so long as it is subsisting, the Administrative Agent shall, subject to Permitted Encumbrances, be lawfully entitled to peaceably and quietly enter into, have, hold, use, occupy, possess and enjoy the Property with its appurtenances without suit, hindrance, interruption or denial by it, any other Obligor or any other person whomsoever.
- (22) Remedy of Force Majeure Use its reasonable commercial efforts to remedy or cause to be remedied any event of Force Majeure or causes thereof; provided that notwithstanding the foregoing, no party shall be required to settle strikes of its employees or of employees of its contractors, sub-contractors and others on terms which it reasonably regards as unreasonable.
- (23) <u>Management and Control of Property</u> It shall ensure that the Property is diligently managed, leased, used and operated: (i) in compliance in all materials respects with Applicable Law, (ii) at the standard of that of a prudent owner of comparable property, and (iii) in a proper and efficient manner with a view to preserving and protecting: (A) the Property at the same or higher standard as when it became subject to the Security hereunder, and (B) the earnings, incomes, rents, and profits thereof.

- (24) <u>Encumbrances</u> Unless the same shall constitute a Permitted Encumbrance, it shall, and shall cause the other Obligors to discharge, in a timely manner, all encumbrances against the Property from time to time.
- (25) <u>Title</u> Defend the Borrower's and the Nominee's title to the Property and every part thereof against the claims of all Persons whomsoever and do, observe and perform all obligations and all things necessary or expedient to be done, observed or performed by virtue of any Applicable Law for the purpose of creating, maintaining and keeping maintained the Security constituted by the Loan Documents as valid and effective security with the priority required hereunder.
 - (26) Store Ground Lease Observe, as tenant under the Ground Lease, the following:
 - (a) pay (or cause to be paid) promptly when due and not allow to become delinquent any payments of rent, additional rent or any other amounts required to be paid under the terms of the Store Ground Lease;
 - (b) not to amend, alter, modify, restate or supplement the Store Ground Lease in any material respect without first obtaining the consent of the Administrative Agent, such consent not to be unreasonably withheld, provided that if the amendment, alteration, modification, restatement or supplement is as to increasing the rent, shortening the term, not exercising any renewal or changing the area and/or description of the leased premises, the Administrative Agent's consent may be arbitrarily withheld in the Administrative Agent's sole discretion;
 - (c) not to enter into any agreement to surrender or terminate the Store Ground Lease or take any steps or give any notice the effect of which would be to initiate and/or cause a surrender or termination of the Store Ground Lease;
 - (d) exercise such rights of renewal and/or extension of the term of the Store Ground Lease as and when permitted to do so under the terms of the Store Ground Lease, and give to the Administrative Agent not less than thirty (30) days' prior notice thereof when exercising such rights; and
 - (e) observe and perform in all of its material respects all obligations under the Store Ground Lease in accordance with its terms and within the time periods required under the terms of the Store Ground Lease.
- (27) Non-Disturbance Agreements. Simultaneously with the execution of any future Lease in respect of which the Administrative Agent makes a specific request, and following the Closing Date, the Borrower shall use its commercially reasonable efforts to obtain from the tenant thereunder an attornment and non-disturbance agreement in the Administrative Agent's standard form (or such other form as the Administrative Agent and such tenant may agree upon provided that in the event that the Administrative Agent and such tenant cannot agree upon a form of non-disturbance agreement, the Administrative Agent agrees that it shall enter into a priority agreement with the tenant granting priority to such tenant's Lease over the Security so long as the tenant agrees to attorn to the Administrative Agent). Where any other tenant under a

Lease requests the Borrower to obtain a non-disturbance agreement from the Administrative Agent in respect of such Lease, the foregoing sentence shall apply, *mutatis mutandis*.

- (28) <u>Up-to-date Appraisals</u> The Borrower shall, upon the request of the Administrative Agent, deliver to the Administrative Agent an up-to-date Appraisal of the Property, such Appraisal to be delivered no later than 60 days after the Administrative Agent's request therefor.
- (29) Soil Vapour Assessment Within three (3) months after the Closing Date, (a) cause to be completed a soil vapour assessment with respect to the lower level of the building comprising the Property and, if required, sample indoor air with respect to the Property, in accordance with the recommendations set out in the Phase One Environmental Assessment dated as of May 10, 2023, conducted by OHE Consultants, and deliver the results of such soil vapour assessment and, if applicable indoor air sampling, to the Administrative Agent, and (b) if required by the Administrative Agent, acting reasonably (on the basis of the results of such soil vapour assessment and, if applicable, indoor air sampling), implement any mitigation measures resulting from the findings of the soil vapour assessment and, if applicable, indoor air sampling.

9.02 Reporting Requirements

So long as this Agreement is in force, each of the Obligors as applicable will deliver to the Administrative Agent:

(1) Annual Reports

- (a) Within 120 days from the last day of each Fiscal Year, the annual unaudited financial statements of the Borrower, and within 90 days from the last day of each Fiscal Year, the annual audited financial statements of the Guarantor, including, in each case, where applicable, balance sheet, statement of income and retained earnings, statement of changes in financial position, source and application of funds for such Fiscal Year, which will be prepared in accordance with GAAP; and
- (b) Within 120 days from the last day of each Fiscal Year for the Property, an annual operating budget in respect of the Property together with a timeline of capital expenditures for the Property.

(2) Quarterly Reports

- (a) Within 60 days of the end of each applicable Fiscal Quarter of the Borrower (other than the fourth Fiscal Quarter in each Fiscal Year), quarterly unaudited financial statements of the Borrower, including, in each case, where applicable, balance sheet, statement of income and retained earnings, statement of changes in financial position and source and application of funds for such Fiscal Quarter, which will be prepared in accordance with GAAP, and be certified by an officer of the Borrower;
- (b) Within 45 days of the end of each applicable Fiscal Quarter of the Guarantor (other than the fourth Fiscal Quarter in each Fiscal Year), quarterly unaudited financial statements of the Guarantor, including, in each case, where applicable, balance sheet, statement of income and retained earnings, statement of changes in financial position

- and source and application of funds for such Fiscal Quarter, which will be prepared in accordance with GAAP, and be certified by an officer of the Guarantor; and
- (c) Within (i) 60 days of the end of each Fiscal Quarter (excluding the fourth Fiscal Quarter), and (ii) 120 days from the end of each Fiscal Year, a quarterly operating statement and an up to date rent roll, showing tenant name, unit number, square feet leased, annual net rent, and lease term.
- (3) <u>Compliance Certificate</u> A Compliance Certificate from the Guarantor concurrently with the delivery of the financial statements referred to in Sections 9.02(1) and 9.02(2) (including statements confirming compliance with the applicable provisions of Section 9.05 hereof).
- (4) <u>Insurance Reporting</u> Concurrently with the renewal or placement of any insurance required to be maintained by Section 9.01(7) or 9.01(8), delivery to the Administrative Agent of certificates of insurance relating to such insurance or make available for review at the Borrower's offices or the offices of the Borrower's insurance broker such certificates of insurance.
- (5) <u>Realty Taxes</u> Delivery of evidence (satisfactory to the Administrative Agent) of the payment of all realty taxes relating to the Property, such evidence to be delivered within 60 days of each calendar year-end.
- (6) <u>Other Information</u> Such other information as the Administrative Agent may reasonably request respecting the Obligors or the Property.
- (7) KYC Documentation and Anti-Money Laundering. The Obligors acknowledge that the Lenders have certain anti-money laundering and anti-terrorism responsibilities under various laws and regulations and that from time to time a Lender (including any prospective assignee or participant) may request information in order to comply with Applicable Laws and internal requirements (including any applicable "know your customer" or "know your client" requirements). The Obligors covenant and agree, upon request, to promptly provide the Lenders such additional information as may be reasonably requested. Each Obligor shall also provide the Administrative Agent with prompt written notice of any change in beneficial owners (except in the case of the Guarantor), key officers or directors after the date of this Agreement. The Obligors covenant and agree that the proceeds of the Drawdown under the Credit Facility shall not be required or invested in order to support domestic or international terrorism and shall not be directly or indirectly derived from activities that may contravene Applicable Laws, including anti-money laundering laws and regulations.

9.03 Negative Covenants – Obligors

So long as this Agreement is in force, each Obligor covenants and agrees that they shall not:

(1) <u>No Sale of Property</u> In respect of the Borrower only, other than (a) the Disposition of premises under Leases as permitted hereunder, (b) Dispositions otherwise contemplated in Sections 10.04 or 10.05, or (c) Dispositions of machinery or equipment that is

being replaced or is obsolete, Dispose of, or permit the Disposition of, the Property or any part thereof or interest therein unless approved unanimously by the Lenders acting in their sole discretion.

- (2) <u>No Change of Control</u> Permit the occurrence of any Change of Control of any of the Obligors.
- (3) <u>No Consolidation, Amalgamation, etc.</u> (i) consolidate, amalgamate or merge with any other Person, enter into any corporate reorganization or other transaction intended to effect or otherwise permit a change in its existing corporate structure, or (ii) liquidate, wind-up or dissolve itself, or permit any liquidation, winding-up or dissolution.
- (4) <u>No Change of Name</u> Change its name without providing the Administrative Agent with 30 days' prior written notice thereof.
- (5) <u>No Distributions/Restricted Payments</u> In respect of the Borrower only, make any Distribution during the presence and existence of an Event of Default.
- (6) <u>No Encumbrances</u> In respect of the Borrower only, Create, incur, assume or permit to exist any Encumbrance upon the Collateral or any part thereof or interest therein except Permitted Encumbrances.
- (7) <u>No Change to Year End</u> In respect of the Borrower only, make any change to its Fiscal Year end.
- (8) <u>No Continuance</u> Continue into any other jurisdiction without the consent of the Majority of the Lenders, which consent will not be unreasonably withheld.
- (9) <u>Amendments to Organizational Documents</u> Amend any of its Organizational Documents in a manner that would be materially prejudicial to the interests of any of the Lenders or the Administrative Agent under the Loan Documents.
- (10) <u>Amendments to Material Property Agreements</u> Except as otherwise contemplated in Section 9.01(19)(a), amend, vary or alter in any material way (it being acknowledged that any increase in the amounts of any payments due under the Management Agreement), consent to any assignment or transfer of, or waive or surrender any of its material rights or material entitlements under, any Material Property Agreements.
- (11) <u>Leasing</u> In respect of the Borrower only, enter into, or permit the Nominee to enter into, any Leases or renew, amend, terminate, forfeit or cancel any Leases in respect of the Property unless such Leases, amendments, renewals, terminations, forfeitures or cancellations reflect in all material respects good business practice and such material terms as a prudent owner of a similar property would accept having regard to all relevant factors and the leasing practice in the market at the relevant time. In respect to the Store Sublease, the Borrower shall obtain the prior written consent of the Administrative Agent in respect thereof (such consent not to be unreasonably withheld or delayed).

- (12) <u>Concerning Leases Generally</u> In respect of the Borrower only, accept or require, or permit the Nominee to accept or require, payment of rent or other moneys payable by a tenant under any Lease in respect of the Property that would result in more than one month of such rent or other moneys being prepaid under such Lease other than:
 - (a) prepaid rent or deposits on account of rent which represents the portion of the cost of construction of the relevant demised premises which exceeds the portion of such cost which was used as the basis for determining the basic rental otherwise payable under such Lease; or
 - (b) amounts representing a bona fide pre-calculation of any amount that is required to be paid under such Lease in addition to basic rental, including amounts payable with respect to taxes and maintenance of the Property and any applicable percentage rentals; or
 - (c) lease surrender payments and security deposits made by the tenant under such Lease.
- (13) Residency Become a non-resident of Canada within the meaning of Section 116 of the *Income Tax Act* (Canada).
 - (14) Intentionally Omitted
 - (15) Intentionally Omitted
- (16) <u>Not To Remove, Destroy</u> In respect of the Borrower only, destroy, remove, permit to be destroyed or removed any of the buildings, plant, machinery or equipment comprising part of the Property; provided that:
 - (a) nothing herein shall prevent the removal of any such property from one part of the Property to another or the temporary removal of any such property for the purposes of repair; and
 - (b) it and/or the relevant Obligor may remove, dismantle, sell, exchange or otherwise dispose of any plant, machinery or equipment which has become obsolete, worn out, replaced or unserviceable if such plant, machinery or equipment is replaced by plant, machinery or equipment of at least equal performance or if such plant, machinery or equipment so dealt with is unnecessary for use in the conduct of its business at the Property; provided that such removal or other disposal does not impair the successful operation of the Property.

9.04 <u>Intentionally Omitted</u>

9.05 **Guarantor Financial Covenants**

So long as this Agreement is in force, the Guarantor covenants with the Administrative Agent and the Lenders to:

- (a) **Total Indebtedness Ratio**. Maintain at all times a ratio of Consolidated Indebtedness to Consolidated Gross Book Value of not more than 60%, such to be tested quarterly as at the end of each Fiscal Quarter. The calculation shall reflect Proportionate Consolidation Adjustments.
- (b) **Secured Indebtedness Ratio**. Maintain at all times a ratio of Consolidated Secured Indebtedness to Consolidated Gross Book Value of not more than 40%, such to be tested quarterly as at the end of each Fiscal Quarter. The calculation shall reflect Proportionate Consolidation Adjustments.
- (c) **Debt Service Coverage Ratio**. Maintain at all times a ratio of Consolidated EBITDA to Debt Service of not less than 1.50:1.00, such to be tested quarterly as at the end of each Fiscal Quarter calculated on a rolling four Fiscal Quarter basis. The calculation shall reflect Proportionate Consolidation Adjustments.
- (d) **Minimum Unitholders' Equity**. Maintain at all times Unitholders' Equity of not less than the sum of (i) Cdn. \$5,000,000,000 and (ii) 75% of the net proceeds received in connection with any equity offerings made by the Guarantor after the date of this Agreement.
- (e) **Unencumbered Property Assets Ratio**. Maintain at all times a ratio of Unencumbered Property Assets Value to Consolidated Unsecured Indebtedness of not less than 1.50:1.00, such to be tested quarterly as at the end of each Fiscal Quarter. The calculation shall reflect Proportionate Consolidation Adjustments.

ARTICLE 10- SECURITY

10.01 Security

As general and continuing security for the payment and performance of the Obligations, the security described below will be granted to the Administrative Agent for and on behalf of the Lenders and shall be in form satisfactory to the Lenders:

- (a) the Debenture;
- (b) the General Assignment of Leases;
- (c) the Direction and Acknowledgement; and
- (d) such other security as the Lenders require, which is contemplated by this Agreement or which security more fully gives effect to the security contemplated by this Agreement.

10.02 Registration

The Borrower shall, at its expense, register, file or record, or cause the registration, filing or recordation of, the Security in all offices where such registration, filing or

recording is necessary or of advantage to the creation, perfection and preserving of the security applicable to it including, without limitation, any land registry offices as required hereunder. The Borrower shall renew, or shall cause to be renewed, such registrations, filings and recordings from time to time as and when required to keep them in full force and effect. The forms of Debenture and other Loan Documents have been prepared based upon Applicable Law in effect at the date hereof and recognize that such laws may change. The Administrative Agent shall have the right to require that any such forms be amended to reflect any changes in such laws, whether arising as a result of statutory amendments, court decisions or otherwise, in order to confer upon the Administrative Agent the security interests intended to be created thereby, except that in no event shall the Administrative Agent require that any such amendment be effected if the result thereof would be to grant the Administrative Agent greater rights than is otherwise contemplated herein. The Borrower authorizes, and shall cause each other Obligor to authorize, the Obligor's Counsel or the Lenders' Counsel to register, if required, by any land registrar, this Agreement or notice thereof on title.

10.03 After Acquired Property and Further Assurances

Each Owner Party will from time to time execute and deliver all such further deeds or other instruments of conveyance, assignment, transfer, mortgage, pledge or charge in connection with all property related to the Property acquired by such Owner Party after the date hereof, or as may be required to properly perfect the security interest of the Administrative Agent in any property related to the Property.

10.04 Non-Material Dealings with Property Generally

Unless there is an Event of Default that is continuing, the Borrower may in the ordinary course of business at any time and from time to time:

(a) without receiving any consideration therefor, Dispose of, exchange, surrender or grant any part or parts of any of the then-existing Property (not constituting a material portion thereof) or any licenses, easements, rights-of-way or rights in the nature of easements in respect of any part or parts of any of the then-existing Property or to a municipality or other Governmental Authority or transit commission or utility, or to an owner, lessee or licensee of any lands adjacent to then-existing Property or separated therefrom by a public street or to such other Person as shall be designated by a Governmental Authority or such aforementioned owner, lessee or licensee (which may or may not include the grant or Disposition to the Borrower of any land or any license or licenses, or easements, rights-of-way or rights in the nature of easements in, over, under or in respect of other lands), and the Administrative Agent shall release same from the Security or postpone the Security, as applicable, upon receipt of a written request of the Borrower or the Obligors' Counsel to the Administrative Agent stating that such grant, exchange, surrender or Disposition is necessary or desirable without regard to any consideration received by the Borrower therefor for the servicing or operation of the Property and would not cause a Material Adverse Change:

- (b) accept any ancillary lands, licenses, easements, rights-of-way or rights in the nature of easements which shall thereafter form part of the Property on such terms as the Borrower may determine (provided such additional lands or rights shall be made subject to the Security on terms and conditions satisfactory to the Lenders); and
- (c) create or permit to be created any Encumbrance of the type referred to in the definition of Permitted Encumbrances provided any applicable provisions of this Agreement are complied with.

10.05 Lot Line Adjustments/Easements, etc. relating to the Property

Unless there is an Event of Default that is continuing, the Borrower may in the ordinary course of business at any time and from time to time, without receiving any consideration therefor, Dispose of, exchange, surrender or grant any part or parts of any of the then-existing Property or any licenses, easements, rights-of-way or rights in the nature of easements in respect of any part or parts of any of the then-existing Property in order to adjust lot lines or boundaries, or grant easements or other rights-of-way, relating to the Property or any component thereof as may be necessary to the good and proper operation of each portion of the Property, and the Administrative Agent shall, if necessary, amend or postpone or partially discharge the Security, as applicable, upon receipt of a written request of the Borrower or the Obligors' Counsel to the Lenders setting forth the required action and the details of same (including an explanation as to the necessity thereof) and confirmation that same shall not result in a Material Adverse Change and the Administrative Agent is authorized to take such actions as may be necessary to effect same (provided that the Administrative Agent may require such opinions of Obligors' Counsel to be delivered in conjunction therewith as the Lenders may require).

10.06 Independent Investigation by Administrative Agent and Lenders

Notwithstanding any other provision of this Article 10, before taking any action provided for in this Article 10, the Administrative Agent shall advise the Lenders as to the details of the matter and shall seek the instruction of the Lenders in relation thereto and the Administrative Agent shall not make or take (and the Borrower agrees that the Administrative Agent shall not be obliged to make or take) respectively, any action which the Lender (and a Majority of the Lenders, in cases where there is more than one Lender) instructs the Administrative Agent not to make or take, as the case may be, without causing to be made such independent investigations and/or seeking such advice from the Lenders' Counsel as the Lender or a Majority of the Lenders, as applicable, acting reasonably, may see fit, prior to any such release being made or action being taken so as to satisfy themselves that the conditions relating to the making of such release or the taking of such action as set forth in this Article 10 have been met and the reasonable expense thereof shall be paid by the Borrower, or if paid by the Administrative Agent or the Lenders shall be repaid by the Borrower upon reasonable notice given by the Administrative Agent or the Lenders to the Borrower with interest at the rate of interest per annum equal to the Prime Rate plus the Prime Rate Margin from the date of such demand.

10.07 <u>Eligible Hedge Agreements</u>

- (1) The Security shall secure all Obligations owing under or in respect of the Eligible Hedge Agreements on a *pari passu* basis with all other Obligations. For greater certainty, if any Lender has entered into an Eligible Hedge Agreement and such Lender thereafter ceases to have any outstanding Commitments hereunder, the obligations and amounts owing under such Eligible Hedge Agreements entered into and in effect at the time such Lender ceases to have any outstanding Commitments hereunder shall continue to be secured by the Security and shall continue to rank *pari passu* with all other Obligations.
- (2) Notwithstanding payment and performance in full of all of the Obligations (other than the Obligations under the Eligible Hedge Agreements), the Security shall continue to secure the Obligations of the Borrower under all outstanding Eligible Hedge Agreements and the Borrower shall not be entitled to a discharge of the Security until all Eligible Hedge Agreements have been terminated and all amounts, if any, owing by the Borrower in respect thereof have been paid to the applicable Lender(s) thereunder or replacement security satisfactory to such Lender(s) in its or their sole discretion acting reasonably has been provided by the Borrower.

ARTICLE 11- GUARANTEE

11.01 **Guarantee and Indemnity**

- (1) The Guarantor hereby unconditionally and irrevocably guarantees payment and performance of the Obligations of the Borrower (including, without limitation, any Obligation of the Borrower to indemnify the Administrative Agent and each Lender hereunder).
- (2) If any or all of the Obligations are not duly paid and are not recoverable under Section 11.01(1) for any reason whatsoever, the Guarantor hereby, unconditionally and irrevocably, will, as a separate and distinct obligation, indemnify and save harmless the Administrative Agent and each Lender from and against any losses resulting from the failure of any Obligor to pay the Obligations.
- (3) If any or all of the Obligations are not duly paid and are not recoverable under Section 11.01(1) or the Administrative Agent and each Lender is not indemnified under Section 11.01(2), in each case, for any reason whatsoever, the Obligations will, as a separate and distinct obligation, be paid by and be recoverable from the Guarantor, as primary obligor.

11.02 Obligations Absolute

The liability of the Guarantor hereunder is absolute and unconditional and is not affected by:

- (a) any lack of validity or enforceability of this Agreement or any other Loan Document;
- (b) any impossibility, impracticability, frustration of purpose, illegality, Force Majeure or act of Governmental Authority;

- (c) the bankruptcy, winding-up, liquidation, dissolution, arrangement, insolvency or other similar proceeding affecting any Obligor or any other Person, the amalgamation of or any change in the status, function, control or ownership of any Obligor or any other Person;
- (d) any lack or limitation of power, incapacity or disability on the part of any Obligor or of the directors, partners or agents thereof or any other irregularity, defect or informality on the part of any Obligor in its Obligations; or
- (e) any other law, regulation or other circumstance that might otherwise constitute a defence available to, or a discharge of, any Obligor in respect of any or all of the Obligations.

11.03 No Release

The liability of the Guarantor hereunder is not released, discharged, limited or in any way affected by anything done, suffered or permitted by the Administrative Agent and each Lender or any other Person in connection with any duties or liabilities of any Obligor to the Administrative Agent and each Lender or any Security including any loss of or in respect of any Security. Without limiting the generality of the foregoing and without releasing, discharging, limiting or otherwise affecting in whole or in part the liability of the Guarantor hereunder, without obtaining the consent of or giving notice to the Guarantor, the Lenders may, subject to the terms of this Agreement and specifically the relationship between the Lenders:

- (a) discontinue, reduce, increase or otherwise vary the credit of any Obligor in any manner whatsoever;
- (b) make any change in the time, manner or place of payment under, or in any other term of, any Loan Document or the failure on the part of any Obligor to carry out any of its obligations under any Loan Document;
- (c) grant time, renewals, extensions, indulgences, releases and discharges to any Obligor;
- (d) take or abstain from taking or enforcing the Security or from perfecting Security;
- (e) accept compromises from any Obligor;
- (f) apply all money at any time received from any Obligor or from the Security upon such part of the Obligations as the Lenders may see fit or change any such application in whole or in part from time to time as each of them may see fit; and
- (g) otherwise deal with each Obligor and all other Persons and the Security as the Lenders may see fit.

11.04 No Exhaustion of Remedies

The Lenders are not bound or obligated to exhaust their recourse against any Obligor or other Person or any Security it or they may hold, or take any other action before being entitled to demand payment from any Obligor hereunder or any other Loan Document.

11.05 *Prima Facie* Evidence

Any account settled or stated in writing by or between the Administrative Agent and any Obligor will be, in the absence of manifest error, *prima facie* evidence that the balance or amount thereof appearing due to the Lenders is so due.

11.06 No Set-Off

In any claim by any Lender or the Administrative Agent against the Guarantor, the Guarantor may not assert any set-off or counterclaim that the Guarantor or any other Obligor may have against the Lender or the Administrative Agent.

11.07 <u>Continuing Guarantee</u>

The Obligations of the Guarantor hereunder will constitute and be continuing obligations and will apply to and secure any ultimate balance due or remaining due to the Lenders and the Administrative Agent and will not be considered as wholly or partially satisfied by the payment or liquidation at any time of any sum of money for the time being due or remaining unpaid to any such Person. The Obligations of the Guarantor hereunder will continue to be effective even if at any time any payment of any of the Obligations is rendered unenforceable or is rescinded or must otherwise be returned by the recipient of such payment upon the occurrence of any action or event, including the insolvency, bankruptcy or reorganization of any Obligor or otherwise, all as though such payment had not been made.

11.08 Waivers by the Guarantor

The Guarantor hereby irrevocably waives acceptance hereof, presentation, demand, protest and any notice of this Agreement, as well as any requirement that at any time any action be taken by any Person against the Guarantor, any other Obligor or any other Person.

11.09 Demand

The Guarantor will make payment to the Lenders of the Obligations and all other amounts payable by them under Section 11.10 forthwith after demand therefor is made to it. The Guarantor will also make payment to the Administrative Agent of 100% of all costs and expenses incurred by the Lenders and the Administrative Agent in enforcing the provisions of this Article 11 against the Guarantor.

11.10 Interest

The Guarantor will pay interest to the Lenders at the Prime Rate plus the Prime Rate Margin plus 1% per annum on the unpaid portion of all amounts payable by the Guarantor

hereunder, such interest to accrue from and including the date of demand on the Guarantor, and will be compounded monthly.

11.11 <u>Assignment and Postponement</u>

All debts and liabilities, present and future, of the Borrower to the Guarantor are hereby assigned to the Administrative Agent and postponed to the Obligations, and, following an Event of Default that remains uncured, all money received by the Guarantor in respect thereof will be held in trust for the Lenders and forthwith upon receipt will be paid over to the Administrative Agent, the whole without in any way lessening or limiting the liability of the Guarantor hereunder and this assignment and postponement is independent of the guarantee, indemnity and primary obligor obligations contained in this Article 11 and will remain in full force and effect until, in the case of the assignment, the liability of the Guarantor under this Agreement has been discharged or terminated and, in the case of the postponement, until all Obligations are performed and indefeasibly paid in full.

11.12 Subrogation; Contribution

The Guarantor will not be entitled to subrogation or to contribution from any other Person by reason of any payment hereunder until indefeasible payment in full of all Obligations and the termination of the Commitments. Thereafter, the Administrative Agent will, at the Guarantor's request and expense, execute and deliver to the Guarantor appropriate documents, without recourse and without representation and warranty, except as to the amount owing, necessary to evidence the transfer by subrogation to the Guarantor of an interest in the Obligations and any Security held therefor resulting from such payment by the Guarantor.

ARTICLE 12- DEFAULT

12.01 <u>Events of Default</u>

The occurrence of any one or more of the following events (each such event being referred to as an "Event of Default") will constitute a default under this Agreement:

- (a) if the Borrower defaults in payment of any principal of any Loan when the same is due and payable, whether acceleration or otherwise;
- (b) if the Borrower fails to pay any interest, fees or other Obligations (other than any principal amount) when due and such default continues for three (3) Business Days after notice of such default has been given by the Administrative Agent to the Borrower;
- (c) if any Obligor breaches any covenant in Sections 9.01(26) or 9.03;
- (d) if the Guarantor breaches any covenant in Section 9.05;
- (e) if any Obligor neglects to observe or perform, in any material respect, any covenant or obligation contained in this Agreement or any other Loan Document on its part to

be observed or performed (other than a covenant or condition whose breach or default in performance is specifically dealt with elsewhere in this Section 12.01 or such Loan Document) and such Obligor fails to remedy such default within 30 days from the earlier of (i) the date such Obligor becomes aware of such default, and (ii) the date the Administrative Agent delivers written notice of the default to such Obligor, or where the Majority of the Lenders (having regard to the subject matter of the default) have agreed, that such default cannot be cured within such 30 days, such longer period as is required so long as such Obligor is diligently proceeding at all times to cure such default and provided that, in any event, such cure period shall not extend longer than four months without the consent of the Majority of the Lenders, acting in their sole discretion;

- (f) if any information, representation or warranty given or made by any Obligor in this Agreement, any Loan Document or in any certificate or other document at any time delivered hereunder to the Administrative Agent or any Lender proves to have been incorrect or misleading in any material respect on and as of the date that it was made or was deemed to have been made and such Obligor fails to remedy such default within ten (10) Business Days of the occurrence of such event (or such longer period as the Majority of the Lenders may agree to having regard to the nature of such default and provided the affected Obligor is proceeding diligently to cure such default);
- (g) if any Obligor ceases or threatens to cease to carry on business generally or admits its inability or fails to pay its debts generally;
- (h) if any Obligor denies, to any material extent, its obligations under any Loan Document or claims any of the Loan Documents to be invalid or withdrawn in whole or in part;
- (i) any of the Loan Documents or any material provision of any of them becomes unlawful or is changed by virtue of legislation or by a Governmental Authority, if any Obligor does not, within 15 Business Days of receipt of notice of such Loan Document or material provision becoming unlawful or being changed, replace such Loan Document with a new agreement that is in form and substance satisfactory to the Lenders, or amend such Loan Document to the satisfaction of the Lenders;
- (j) if a decree or order of a court of competent jurisdiction is entered adjudging an Obligor a bankrupt or insolvent or approving as properly filed a petition seeking the winding-up of an Obligor under the *Companies' Creditors Arrangement Act* (Canada), the *Bankruptcy and Insolvency Act* (Canada), the *United States Bankruptcy Code* or the *Winding-up and Restructuring Act* (Canada) or any other bankruptcy, insolvency or analogous laws or issuing sequestration or process of execution against any substantial part of the assets of an Obligor or ordering the winding-up or liquidation of its affairs, and any such decree or order continues unstayed and in effect for a period of 30 days;

- (k) if any Obligor becomes insolvent, makes any assignment in bankruptcy or makes any other assignment for the benefit of creditors, makes any proposal under the *Bankruptcy and Insolvency Act* (Canada) or any comparable law, seeks relief under the *Companies' Creditors Arrangement Act* (Canada), the *United States Bankruptcy Code*, the *Winding-up and Restructuring Act* (Canada) or any other bankruptcy, insolvency or analogous law, is adjudged bankrupt, files a petition or proposal to take advantage of any act of insolvency, consents to or acquiesces in the appointment of a trustee, receiver, receiver and manager, interim receiver, custodian, sequestrator or other Person with similar powers of itself or of all or any substantial portion of its assets, or files a petition or otherwise commences any proceeding seeking any reorganization, arrangement, composition or readjustment under any applicable bankruptcy, insolvency, moratorium, reorganization or other similar law affecting creditors' rights or consents to, or acquiesces in, the filing of such a petition;
- (l) if an Encumbrancer takes possession, by appointment of a receiver, receiver and manager or otherwise, of: (i) all or any material part of the Lands, or (ii) all or any material part of the remainder of the Property;
- (m) if proceedings are commenced for the dissolution, liquidation or voluntary windingup of any Obligor, or for the suspension of the operations of any Obligor unless such proceedings are being actively and diligently contested in good faith;
- (n) if a final judgment or decree for the payment of money due has been obtained or entered against any Obligor in an amount, when combined with any other such judgment or decrees, is in an amount in excess of, in the case of the Borrower or the Nominee, Cdn. \$5,000,000, or in the case of the Guarantor, Cdn. \$35,000,000, and such judgment or decree has not been and remained vacated, discharged or stayed pending appeal within the applicable appeal period or the applicable Obligor has not demonstrated to the satisfaction of the Lenders that it has the financial ability to satisfy such judgement or decree without adversely affecting in any material way, such Obligor's ability to perform its obligations under the Loan Documents;
- (o) if (i) the Borrower fails to make any payment when such payment is due and payable to the Lenders and the Administrative Agent or defaults in the observance or performance of any other agreement or condition in relation to any Indebtedness (other than the Loan and other Obligations) to the Lenders and the Administrative Agent that is not cured within the applicable grace periods provided in connection with such Indebtedness; or (ii) any Obligor fails to make any payment when such payment is due and payable to any Person or defaults in the observance or performance of any other agreement or condition in relation to any Indebtedness (other than the Loan and other Obligations) to any Person or Persons that in the aggregate principal amount then outstanding is, in either or both cases, in excess of, in the case of the Guarantor, Cdn. \$35,000,000, or in the case of the Borrower or the Nominee, Cdn. \$5,000,000 or contained in any instrument or agreement evidencing, securing or relating thereto and such default or other condition has not been remedied within any applicable grace period and such Person to whom such payment is due and payable is entitled to accelerate the payments owing;

- (p) if any Security ceases to constitute a valid and perfected first priority security interest (subject only to Permitted Encumbrances) and, provided the Lenders are satisfied that their position will not be prejudiced, the applicable Obligor has failed to remedy such default within ten Business Days of becoming aware of such fact;
- (q) if an event of default occurs under any Material Property Agreement resulting in, or is likely to result in, a Material Adverse Change and which default is not cured or remedied within five (5) Business Days from the date the Borrower becomes aware or ought to have become aware of such default;
- (r) if, except as permitted under Section 9.03(2), there is any reorganization of any Obligor or any consolidation, merger or amalgamation of any Obligor with any other company or companies, the effect of which would likely be a Material Adverse Change, or if a Change of Control occurs;
- (s) if in the opinion of any Lender, a Material Adverse Change has occurred; or
- (t) if the Borrower defaults in payment of any Obligations under an Eligible Hedge Agreement when due and payable or is otherwise in default of any Eligible Hedge Agreement beyond any cure period provided therein.

12.02 Acceleration and Enforcement

- (1) If any Event of Default occurs:
- (a) the outstanding principal amount or face amount, as the case may be, of all Loans and all other Obligations will, upon notice in writing from the Administrative Agent (acting in accordance with the direction of the Majority of the Lenders), become immediately due and payable with interest thereon, at the rate or rates determined as herein provided, to the date of actual payment thereof, all without notice, presentment, protest, demand, notice of dishonour or any other demand or notice whatsoever, all of which are hereby expressly waived by the Borrower; provided, if any Event of Default described in Section 12.01(j) or (k) with respect to any Obligor occurs, the Commitments (if not theretofore terminated) will automatically terminate and the outstanding principal amount or face amount, as the case may be, of all Loans and all other Obligations will automatically be and become immediately due and payable; and
- (b) the Administrative Agent (acting in accordance with the direction of the Majority of the Lenders) may exercise any right or recourse and proceed by any action, suit, remedy or proceeding against any Obligor authorized or permitted by law for the recovery of all the Obligations to the Lenders and, whether or not the Administrative Agent (acting in accordance with the direction of the Majority of the Lenders) has exercised any of their respective rights under the foregoing clause (a), proceed to exercise any and all rights hereunder and under the Security.
- (2) The Administrative Agent (acting in accordance with the direction of the Majority of the Lenders) is not under any obligation to the Obligors or any other Person to realize upon

any collateral or enforce the Security or any part thereof or to allow any of the collateral to be dealt with or Disposed of. The Administrative Agent (acting in accordance with the direction of the Majority of the Lenders) is not responsible or liable to the Obligors or any other Person for any loss or damage arising from such realization or enforcement or the failure to do so or for any act or omission on their respective parts or on the part of any director, officer, employee, agent or adviser of any of them in connection with any of the foregoing, save as a result of the gross negligence or wilful misconduct of any such party.

12.03 <u>Intentionally Omitted</u>

12.04 Remedies Cumulative

For greater certainty, it is expressly understood and agreed that the respective rights and remedies of the Administrative Agent and/or the Lenders hereunder or under any other Loan Document or instrument executed pursuant to this Agreement are cumulative and are in addition to and not in substitution for any rights or remedies provided by law or by equity; and any single or partial exercise by the Administrative Agent and/or the Lenders in accordance with the provisions hereof of any right or remedy for a default or breach of any term, covenant, condition or agreement contained in this Agreement or other Loan Document or instrument executed pursuant to this Agreement shall not be deemed to be a waiver of or to alter, affect or prejudice any other right or remedy or other rights or remedies to which the Administrative Agent and/or the Lenders may be lawfully entitled for such default or breach. Any waiver by the Administrative Agent and/or the Lenders of the strict observance, performance or compliance with any term, covenant, condition or other matter contained herein and any indulgence granted, either expressly or by course of conduct, by the Administrative Agent and/or the Lenders shall be effective only in the specific instance and for the purpose for which it was given and shall be deemed not to be a waiver of any rights and remedies of the Administrative Agent and/or the Lenders under this Agreement or any other Loan Document or instrument

12.05 **Perform Obligations**

If an Event of Default has occurred and is continuing and if any Obligor has failed to perform any of its covenants or agreements in the Loan Documents, the Administrative Agent and/or the Lenders may, but will be under no obligation to, perform any such covenants or agreements in any manner deemed fit by the Lenders without thereby waiving any rights to enforce the Loan Documents. The reasonable expenses (including any legal costs) paid by the Lenders and the Administrative Agent in respect of the foregoing will be an Obligation and will be secured by the Security.

12.06 Third Parties

It is not necessary for any Person dealing with the Lenders, the Administrative Agent or any other agent of the Lenders to inquire whether the Security has become enforceable, or whether the powers that the Lenders or the Administrative Agent are purporting to exercise may be exercised, or whether any Obligations remain outstanding upon the security thereof, or as to the necessity or expediency of the stipulations and conditions subject to which any sale is to

be made, or otherwise as to the propriety or regularity of any sale or other disposition or any other dealing with the collateral charged by such Security or any part thereof.

12.07 **Application of Payments**

If any Event of Default shall occur and be continuing, all payments made by the Borrower or any Obligor hereunder shall be applied in the following order:

- (a) to amounts due hereunder as costs and expenses in respect of Loans;
- (b) to amounts due hereunder as overdue interest in respect of Loans;
- (c) to amounts due hereunder as fees on Loans;
- (d) to amounts due hereunder as interest on Loans; and
- (e) to (i) amounts due hereunder as principal of Loans, and (ii) amounts due under Eligible Hedge Agreements.

ARTICLE 13- COMPENSATION AND SET-OFF

13.01 Increased Costs

- (1) <u>Increased Costs Generally</u>. If any Change in Applicable Laws shall:
 - (a) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender;
 - (b) subject any Lender to any Tax of any kind whatsoever with respect to this Agreement any or any Loans made by it, or change the basis of taxation of payments to such Lender in respect thereof, except for (x) Indemnified Taxes or Other Taxes covered by Section 13.02 and (y) the imposition, or any change in the rate, of any Excluded Tax payable by such Lender; or
 - (c) impose on any Lender or any applicable interbank market any other condition, cost or expense affecting this Agreement or Loans made by such Lender;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Loans (or of maintaining its obligation to make any such Loans), or to reduce the amount of any sum received or receivable by such Lender hereunder (whether of principal, interest or any other amount), then upon request of such Lender and subject to the Lender providing the certificate referred to in Subsection 13.01(3), the Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender for such additional costs incurred or reduction suffered.

(2) <u>Capital Requirements</u>. If any Lender determines that any Change in Applicable Laws affecting such Lender or any lending office of such Lender or such Lender's holding

company, if any, regarding capital requirements has or would have the effect of reducing the rate of return on such Lender's capital or on the capital of such Lender's holding company, if any, as a consequence of this Agreement, the Commitments of such Lender or the Loans made by such Lender, to a level below that which such Lender or its holding company could have achieved but for such Change in Applicable Laws (taking into consideration such Lender's policies and the policies of its holding company with respect to capital adequacy), then from time to time and subject to the Lender providing the certificate referred to in Subsection 13.01(3), the Borrower will pay to such Lender such additional amount or amounts as will compensate such Lender or its holding company for any such reduction suffered.

- (3) Certificates for Reimbursement. A certificate of a Lender delivered to the Borrower setting forth the amount or amounts necessary to compensate such Lender or its holding company, as the case may be, as specified in Subsections 13.01(1) or 13.01(2) ("Additional Compensation"), including a description of the event by reason of which it believes it is entitled to such compensation, and supplying reasonable supporting evidence (including, in the event of a Change in Applicable Laws, a photocopy of the Applicable Laws evidencing such change) and reasonable detail of the basis of calculation of the amount or amounts, shall be conclusive evidence of the Lender's entitlement to such compensation and the amount thereof absent manifest error. The Borrower shall pay such Lender the amount shown as due on any such certificate within ten (10) days after receipt thereof. In the event the Lender subsequently recovers all or part of the Additional Compensation paid by the Borrower, it shall promptly repay an equal amount to the Borrower. The obligation to pay such Additional Compensation for subsequent periods will continue until the earlier of termination of the Loans or the Commitment affected by the Change in Applicable Laws, change in capital requirement or the lapse or cessation of the Change in Applicable Laws giving rise to the initial Additional Compensation. A Lender shall make reasonable efforts to limit the incidence of any such Additional Compensation and seek recovery for the account of the Borrower upon the Borrower's request at the Borrower's expense, provided such Lender in its reasonable determination suffers no appreciable economic, legal, regulatory or other disadvantage. Notwithstanding the foregoing provisions, a Lender shall only be entitled to rely upon the provisions of this Section 13.01 if and for so long as it is not treating the Borrower in any materially different or in any less favourable manner than is applicable to any other customers of any relevant Lender, where such other customers are bound by similar provisions to the foregoing provisions of this Section 13.01.
- (4) <u>Delay in Requests</u>. Failure or delay on the part of any Lender to demand compensation pursuant to this Section shall not constitute a waiver of such Lender's right to demand such compensation, except that the Borrower shall not be required to compensate a Lender pursuant to this Section for any increased costs incurred or reductions suffered more than nine (9) months prior to the date that such Lender notifies the Borrower of the Change in Applicable Laws giving rise to such increased costs or reductions and of such Lender's intention to claim compensation therefor, unless the Change in Applicable Laws giving rise to such increased costs or reductions is retroactive, in which case the nine (9) month period referred to above shall be extended to include the period of retroactive effect thereof. For greater clarity, no amount shall be payable under this Section 13.01 as

a result of a lender assignment but only to the extent such assignment occurs prior to an Event of Default that is continuing.

13.02 Taxes

- (1) Payments Subject to Taxes. If any Obligor, the Administrative Agent or any Lender is required by Applicable Laws to deduct or pay any Indemnified Taxes (including any Other Taxes) in respect of any payment by or on account of any Obligations, then (i) the sum payable shall be increased by that Obligor when payable as necessary so that after making or allowing for all required deductions and payments (including deductions and payments applicable to additional sums payable under this Section) the Administrative Agent or Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions or payments been required, (ii) the Obligor shall make any such deductions required to be made by it under Applicable Laws and (iii) the Obligor shall timely pay the full amount required to be deducted to the relevant Governmental Authority in accordance with Applicable Laws.
- (2) <u>Payment of Other Taxes by the Borrower</u>. Without limiting the provisions of Subsection 13.02(1) above, the Borrower shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with Applicable Law.
- (3) Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent and each Lender, within fifteen (15) days after written demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by the Administrative Agent or such Lender and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error. In the event the Lender subsequently recovers all or part of the payment made under this Section paid by the Borrower, it shall promptly repay an equal amount to the Borrower so long as no Event of Default has occurred which is then continuing. A Lender shall make reasonable efforts to limit the incidence of any payments under this Section and seek recovery for the account of the Borrower upon the Borrower's request at the Borrower's expense, provided such Lender in its reasonable determination suffers no appreciable economic, legal, regulatory or other disadvantage.
- (4) Evidence of Payments. As soon as practicable after any payment of Indemnified Taxes or Other Taxes by the Borrower to a Governmental Authority, the Borrower shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Administrative Agent.

(5) <u>FATCA</u>. If a payment made to any Lender under this Agreement would be subject to U.S. federal withholding tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the IRC, as applicable), such Lender shall deliver to the Borrower at the time or times prescribed by law and at such time or times reasonably requested by the Borrower such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the IRC) and such additional documentation reasonably requested by the Borrower as may be necessary for the Borrower to comply with its obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment.

13.03 Mitigation Obligations: Replacement of Lenders

- (1) Designation of a Different Lending Office. If any Lender requests compensation under Section 13.01, or requires the Borrower to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 13.02, then such Lender shall use reasonable efforts to designate a different lending office for funding or booking its Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to Section 13.01 or Section 13.02, as the case may be, in the future and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to the Lender. The Borrower hereby agrees to pay all reasonable out-of-pocket costs and expenses incurred by a Lender in connection with any such designation or assignment.
- (2) Replacement of Lenders. If any Lender requests compensation under Section 13.01, if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 13.02, if any Lender's obligations are suspended pursuant to Section 13.05 or if any Lender is a Defaulting Lender, then the Borrower may either, at its sole expense and effort, upon ten (10) days' notice to such Lender and the Administrative Agent: (i) repay all outstanding amounts and Obligations due to such affected Lenders (or such portion which has not been acquired pursuant to clause (ii) below) and thereupon the Commitments of the affected Lenders shall be permanently cancelled and the aggregate Commitments shall be permanently reduced by the same amount and the Commitment of each of the other Lenders shall remain the same; or (ii) require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Article 19), all of its interests, rights and obligations under this Agreement and the other Loan Documents to an assignee (if available) that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:
 - (a) the Borrower pays the Administrative Agent for all the reasonable costs and expenses in connection with the aforementioned assignment;

- (b) the assigning Lender receives payment of an amount equal to the outstanding principal of its outstanding Loans, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any breakage costs and amounts required to be paid under this Agreement as a result of prepayment to a Lender) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
- (c) in the case of any such assignment resulting from a claim for compensation under Section 13.01 or payments required to be made pursuant to Section 13.02, such assignment will result in a reduction in such compensation or payments thereafter; and
- (d) such assignment does not conflict with Applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply. No such cancellation or assignment and delegation shall relieve the Borrower from its obligations to pay any compensation or other amounts under Sections 13.01 and 13.02 accruing prior to the date of such cancellation or assignment and delegation.

13.04 Defaulting Lenders

- (1) <u>Defaulting Lender Adjustments</u> Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by applicable law:
 - (a) <u>Waivers and Amendments</u> Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Majority of the Lenders.
 - (b) Defaulting Lender Waterfall Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 12.01 or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to Section 14.01 shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; third, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to satisfy such Defaulting Lender's potential future funding obligations with respect to Loans under this Agreement; fourth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against

such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; fifth, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and sixth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans in respect of which such Defaulting Lender has not fully funded its appropriate share, and (y) such Loans were made at a time when the conditions set forth in Section 3.01 were satisfied or waived, such payment shall be applied solely to pay the Loans of all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of such Defaulting Lender until such time as all Loans are held by the Lenders pro rata in accordance with the Commitments under the applicable Credit Facility. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

- (c) <u>Fees</u> No Defaulting Lender shall be entitled to receive any upfront fee payable pursuant to Section 4.06 for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender).
- Defaulting Lender Cure If the Borrower, the Administrative Agent and each Non-(d) Defaulting Lender agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, that Lender will, to the extent applicable, purchase at par that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans to be held pro rata by the Lenders in accordance with the Commitments under the Credit Facility, whereupon such Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.

13.05 Illegality

If any Lender determines that any Applicable Laws has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable lending office to make or maintain any Loans, or to determine or charge interest rates based upon any particular rate, then, on notice thereof by such Lender to the Borrower through the Administrative Agent, any obligation of such Lender with respect to the activity that is unlawful shall be suspended until such Lender notifies the Administrative Agent and the Borrower that the

circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Borrower shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay any Loans in order to avoid the activity that is unlawful. Upon any such prepayment, the Borrower shall also pay accrued interest on the amount so prepaid. Each Lender agrees to designate a different lending office if such designation will avoid the need for such notice and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender.

ARTICLE 14 – RIGHT OF SETOFF

14.01 Right of Setoff

If an Event of Default has occurred and is continuing, each of the Lenders and each of their respective Affiliates is hereby authorized at any time and from time to time in its sole discretion to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by such Lender or any such Affiliate to or for the credit or the account of any Obligor against any and all of the Obligations of the Borrower now or hereafter existing under this Agreement or any other Loan Document to such Lender, irrespective of whether or not such Lender has made any demand under this Agreement or any other Loan Document and although such Obligations of the Obligors may be contingent or unmatured or are owed to a branch or office of such Lender different from the branch or office holding such deposit or obligated on such indebtedness. The rights of each the Lenders and their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff, consolidation of accounts and bankers' lien) that the Lenders or their respective Affiliates may have. Each Lender agrees to promptly notify the Borrower and the Administrative Agent after any such setoff and application, but the failure to give such notice shall not affect the validity of such setoff and application. If any Affiliate of a Lender exercises any rights under this Section 14.01, it shall share the benefit received in accordance with Section 16.01 as if the benefit had been received by the Lender of which it is an Affiliate. Notwithstanding the foregoing, and for greater clarity, no Lender may make such set off in respect of any amounts forming collateral as part of the Security, which collateral may only be applied in respect of the Obligations of the Borrower.

ARTICLE 15 – SPECIFIC ENVIRONMENTAL INDEMNITY

15.01 Specific Environmental Indemnity

Each of the Obligors, on a joint and several basis, shall indemnify the Administrative Agent and each Lender and hold the Administrative Agent and each Lender harmless at all times from and against any and all losses, damages and costs (including reasonable counsel fees and out-of-pocket expenses) resulting from any legal action commenced or claim made by a third party against the Administrative Agent or any Lender related to or as a result of actions or omissions on the part of the Borrower or any other Obligor related to or as a consequence of environmental matters or any requirements of Environmental Laws concerning the Property. The Borrower shall have the sole right, at its expense, to control any such legal

action or claim and to settle on terms and conditions approved by the Borrower and approved by the party named in such legal action or claim, provided that if, in the opinion of the Majority of the Lenders, the interests of the Lenders are different from those of the Borrower in connection with such legal action or claim, the Administrative Agent and each Lender shall have the sole right, at the Borrower's expense, to defend its own interests provided that any settlement of such legal action or claim shall be on terms and conditions approved by the Borrower. If the Lenders elect to defend such legal action or claim, the Administrative Agent shall promptly notify the Borrower of same and shall make reasonable efforts to consult with the Borrower on an ongoing basis in connection with such matter. If the Borrower does not defend the legal action or claim, the Lenders shall have the right to do so on its own behalf and on behalf of the Borrower at the expense of the Borrower.

ARTICLE 16 - EXPENSES; INDEMNITY: DAMAGE WAIVER

16.01 Expenses; Indemnity: Damage Waiver

- (1) Costs and Expenses Upon receipt of an invoice/notice, the Borrower shall pay (i) all reasonable out-of-pocket expenses incurred by the Administrative Agent or any Lender and its Affiliates, including the reasonable fees, charges and disbursements of counsel for the Administrative Agent and site inspections by the Lenders, in connection with satisfaction of the conditions precedent in Section 3.01 (including diligence), syndication of the Credit Facility provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), and (ii) all reasonable out-of-pocket expenses incurred by the Administrative Agent or any Lender including the reasonable fees, charges and disbursements of counsel, in connection with the enforcement of its rights in connection with this Agreement and the other Loan Documents, including its rights under this Section, or in connection with all Loans hereunder, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans.
- Indemnification by the Borrower Each Obligor shall indemnify the (2) Administrative Agent (and any sub-agent thereof), each Lender and each Related Person of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all Claims (including, for the purposes of paragraph 16.01(2)(iii) hereof, Environmental Costs) suffered or incurred by any Indemnitee or asserted against any Indemnitee by any third party or by any Obligor arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance or non-performance by the parties hereto of their respective obligations hereunder or thereunder or the consummation or non-consummation of the transactions contemplated hereby or thereby, (ii) any Loan or the use or proposed use of the proceeds therefrom, (iii) any environmental matters arising during or prior to any Obligor's ownership, possession, operation of control of the Property, or (iv) any actual or prospective Claim relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by any Obligor and regardless of whether any Indemnitee is a party thereto, provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such Claims (x) are determined by a court of competent

jurisdiction by final and non-appealable judgment to have resulted from the gross negligence, criminal acts or omissions or wilful misconduct of such Indemnitee or (y) result from a claim brought by any Obligor against an Indemnitee for breach of such Indemnitee's obligations hereunder or under any other Loan Document, if such Obligor has obtained a final and non-appealable judgment in its favour on such claim as determined by a court of competent jurisdiction.

- Reimbursement by Lenders. To the extent that any Obligor for any reason fails to indefeasibly pay any amount required under Subsections 16.01(1) or 16.01(2) to be paid by it to the Administrative Agent (or any sub-agent thereof) or any Related Person of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent) or such Related Person, as the case may be, such Lender's Rateable Portion of such unpaid amount, provided that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) in its capacity as such, or against any Related Person of any of the foregoing acting for the Administrative Agent (or any such sub-agent) in connection with such capacity. The obligations of the Lenders under this Subsection 16.01(3) are subject to the other provisions of this Agreement concerning several liability of the Lenders.
- Waiver of Consequential Damages, Etc. To the fullest extent permitted by Applicable Laws, each Obligor shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for indirect, consequential, punitive, aggravated or exemplary damages (as opposed to direct damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby (or any breach thereof), the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof. No Indemnitee shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby, provided such information or materials are distributed by such Indemnitee in accordance with the provisions of this Agreement or any other Loan Document. Notwithstanding any provision of this Agreement or other Loan Documents to the contrary, the rights of the Indemnitees under this Section 16.01 shall survive discharge or repayment in full of the Credit Facility. Notwithstanding the aforementioned, any indemnity in connection with environmental matters shall not survive discharge or repayment in full of the Credit Facility, unless the Administrative Agent, or any receiver or manager appointed pursuant to the terms of the Security, became a mortgagee in possession.
- (5) <u>Payments</u> Upon provision of an invoice/notice, all amounts due under this Section shall be payable promptly after demand therefor. A certificate of the Administrative Agent setting forth the amount or amounts owing to the Administrative Agent or the Lenders or a sub-agent or Related Person, as the case may be, as specified in this Section, including reasonable detail of the basis of calculation of the amount or amounts, and delivered to the Borrower shall be conclusive absent manifest error.
- (6) <u>Defence of Claims</u> Upon demand by any Indemnitee, the Obligors shall defend any investigation, action or proceeding involving any Claims and Environmental Costs which is

brought or commenced against such Indemnitee, all at the Obligor's own cost and by counsel to be approved by such Indemnitee in the exercise of its reasonable judgment. Obligors' Counsel shall control defence of the claim or proceeding, except that no settlement or compromise shall be accepted or entered into which binds an Indemnitee unless such Indemnitee has given its prior written consent thereto, which consent will not be unreasonably withheld or delayed. In the alternative, any Indemnitee may elect to conduct its own defence at the expense of the Obligors.

ARTICLE 17 – SHARING OF PAYMENTS BY LENDERS

17.01 Sharing of Payments by Lenders.

- (1) If any Lender, by exercising any right of setoff or counterclaim or otherwise, obtains any payment or other reduction that might result in such Lender receiving payment or other reduction of a proportion of the aggregate amount of its Loans and other outstanding Obligations hereunder and the other Loan Documents greater than its Rateable Portion (which, for the purposes of this Section 17.01, shall be calculated with reference to the Commitments) thereof as provided herein, then the Lender receiving such payment or other reduction shall (a) notify the Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the outstanding Loans and such other obligations of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders rateably in accordance with the aggregate amount of principal of and accrued interest on their respective outstanding Loans, other outstanding Obligations and other amounts owing to them, provided that:
 - (a) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations shall be rescinded and the purchase price restored to the extent of such recovery, without interest;
 - (b) the provisions of this Section shall not be construed to apply to (x) any payment made by any Obligor pursuant to and in accordance with the express terms of this Agreement or (y) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans to any assignee or participant, other than to an Obligor or any Affiliate of an Obligor (as to which the provisions of this Section shall apply);
 - (c) the provisions of this Section shall not be construed to apply to (w) any payment made while no Event of Default has occurred and is continuing in respect of obligations of the Borrower to such Lender that do not arise under or in connection with the Loan Documents, or (x) any payment made in respect of an obligation that is secured by a Permitted Encumbrance or that is otherwise entitled to priority over the Borrower's Obligations under or in connection with the Loan Documents;
- (2) Each Obligor consents to the foregoing and agrees, to the extent it may effectively do so under Applicable Laws, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of setoff and counterclaim and similar

rights of Lenders with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation.

ARTICLE 18 - FUNDING/ADMINISTRATIVE AGENT'S CLAWBACK

18.01 Failure of a Lender to Fund

- The Administrative Agent shall only be obliged to make available to the Borrower the Advance if the Administrative Agent has received the applicable amount required from each Lender as required pursuant to this Agreement. Without limiting the generality of the foregoing, unless the Administrative Agent has actual knowledge that a Lender has not made or will not make available to the Administrative Agent for value on a Drawdown Date the applicable amount required from such Lender, the Administrative Agent shall be entitled to assume that such amount has been or will be received from such Lender when so due and the Administrative Agent may in its sole discretion (but shall not be obliged to), in reliance upon such assumption, make available to the Borrower a corresponding amount (except that no such amount shall be made available to the Borrower in the case of a deemed Advance). If such amount is not in fact received by the Administrative Agent from such Lender on such Drawdown Date and the Administrative Agent has made available a corresponding amount to the Borrower on such Drawdown Date as aforesaid (or is deemed to have made an Advance to the Borrower in such amount), such Lender shall pay to the Administrative Agent on demand an amount equal to the aggregate of the applicable amount required from such Lender, plus an amount equal to the product of (i) the rate per annum applicable to overnight deposits made with the Administrative Agent for amounts approximately equal to the amount required from such Lender multiplied by (ii) the amount that should have been paid to the Administrative Agent by such Lender on such Drawdown Date and was not, multiplied by (iii) a fraction, the numerator of which is the number of days that have elapsed from and including such Drawdown Date to but excluding the date on which the amount is received by the Administrative Agent from such Lender and the denominator of which is 365 in the case of all Advances. A certificate of the Administrative Agent containing details of the amount owing by a Lender under this Section shall be binding and conclusive in the absence of manifest error. If any such amount is not in fact received by the Administrative Agent from such Lender on the Drawdown Date, the Administrative Agent shall be entitled to recover from the Borrower, on demand, the related amount made available by the Administrative Agent to the Borrower as aforesaid together with interest thereon at the applicable rate per annum payable by the Borrower hereunder.
- (2) In the event that a Lender fails to make available to the Administrative Agent its Rateable Portion of any Advance (which, for greater certainty, shall include a deemed Advance), and the Administrative Agent has elected not to make available such Lender's Rateable Portion of such Advance on behalf of such Lender pursuant to the provisions of Section 18.01(1) (it being acknowledged that the Administrative Agent has no obligation to make such Advance), then the Administrative Agent shall make available to the Borrower (except in the case of a deemed Advance) the amount made available by those Lenders which have made available their Rateable Portions in accordance with the terms hereof. For greater certainty, the failure of any Lender to make available to the Administrative Agent its Rateable Portion of any Advance as

required herein shall not relieve any other Lender of its obligations to make available to the Administrative Agent its Rateable Portion of any Advance as required herein.

18.02 Payments by the Borrower and the Obligors

All payments made by or on behalf of the Borrower or any other Obligor pursuant to this Agreement or the other Loan Documents shall be made to and received by the Administrative Agent on behalf of the Lenders and shall be distributed by the Administrative Agent to the Lenders as soon as possible upon receipt by the Administrative Agent. Subject to the provisions of Section 18.03 hereof, the Administrative Agent shall distribute in the following order:

- (1) unpaid fees, costs and expenses of the Administrative Agent;
- (2) payments of interest and fees in accordance with each Lender's Rateable Portion of the Credit Facility;
- (3) repayments of principal in accordance with each Lender's Rateable Portion of the Credit Facility;
- (4) amounts received by the Administrative Agent as a result of the exercise of any right of set-off, combination or consolidation of accounts, or by counterclaim or cross-action, in accordance with each Lender's Rateable Portion of the then outstanding Obligations owing to all of the Lenders at the time of such set-off, combination or consolidation of accounts or if applicable, at the time of the receipt of such amounts from any counterclaim or cross-action; and
- (5) all other payments received by the Administrative Agent under this Agreement, in accordance with what would otherwise be each Lender's Rateable Portion of the Credit Facility.

Subject to Section 18.03, if the Administrative Agent does not distribute a Rateable Portion of a payment made by the Borrower or any other Obligor to or for the benefit of that Lender for value on the day that payment is made or deemed to have been made to the Administrative Agent, the Administrative Agent shall pay to the Lender on demand an amount equal to the product of (i) the Administrative Agent's rate per annum applicable to overnight deposits for amounts approximately equal to the amount of the payment multiplied by (ii) the Lender's Rateable Portion of the amount received by the Administrative Agent and not so distributed, multiplied by (iii) a fraction, the numerator of which is the number of days that have elapsed from and including the date of receipt of the payment by the Administrative Agent to but excluding the date on which the payment is made by the Administrative Agent to such Lender and the denominator of which is 365.

18.03 Payments by Administrative Agent

- (1) For greater certainty, the following provisions shall apply to any and all payments made by the Administrative Agent to the Lenders hereunder:
 - (a) the Administrative Agent shall be under no obligation to make any payment (whether in respect of principal, interest, fees or otherwise) to any Lender until an amount in

- respect of such payment has been received by the Administrative Agent from the Borrower or another Obligor;
- (b) if the Administrative Agent receives less than the full amount of any payment of principal, interest, fees or other amount owing by the Borrower or the other Obligors under the Loan Documents, the Administrative Agent shall have no obligation to remit to each Lender any amount other than such Lender's Rateable Portion (based on the then outstanding Obligations) of that amount which is the amount actually received by the Administrative Agent;
- (c) if any Lender advances more or less than its Rateable Portion of the Credit Facility, such Lender's entitlement to such payment shall be increased or reduced, as the case may be, in proportion to the amount actually advanced by such Lender;
- (d) if a Lender's Rateable Portion of an Advance has been advanced for less than the full period to which any payment by the Borrower relates, such Lender's entitlement to such payment shall be reduced in proportion to the length of time such Lender's Rateable Portion of the applicable Advance has actually been outstanding;
- (e) the Administrative Agent acting reasonably and in good faith shall, after consultation with the Lenders in the case of any dispute, determine in all cases the amount of all payments to which each Lender is entitled and such determination shall, in the absence of manifest error, be binding and conclusive;
- (f) upon request, the Administrative Agent shall deliver a statement detailing any of the payments to the Lenders referred to herein; and
- (g) all payments by the Administrative Agent to a Lender hereunder shall be made to such Lender at the address provided by such Lender to the Administrative Agent pursuant to the Administrative Questionnaire.
- The Administrative Agent shall only be obliged to pay Lenders their (2) corresponding amount of a payment from the Borrower (or any portion thereof) if the Administrative Agent has received the payment from the Borrower. Without limiting the generality of the foregoing, unless the Administrative Agent has actual knowledge that the Borrower has not made or will not make a payment to the Administrative Agent for value on the date in respect of which the Borrower has notified the Administrative Agent that the payment will be made, the Administrative Agent shall be entitled to assume that such payment has been or will be received from the Borrower when due and the Administrative Agent may in its sole discretion (but shall not be obliged to), in reliance upon such assumption, pay the Lenders' corresponding amounts. If the payment by the Borrower is in fact not received by the Administrative Agent on the required date and the Administrative Agent has made available corresponding amounts to the Lenders, the Borrower shall, without limiting its other obligations under this Agreement, indemnify the Administrative Agent against any and all liabilities, obligations, losses (other than loss of profit), damages, penalties, and all reasonable costs, expenses or disbursements of any kind or nature whatsoever that may be imposed on or incurred by the Administrative Agent as a result of such non-payment. A certificate of the Administrative

Agent with respect to any amount owing under this Section shall be prima facie evidence of the amount owing in the absence of manifest error. If a payment is not in fact received by the Administrative Agent from the Borrower and the Administrative Agent has paid to a Lender a corresponding amount, such Lender shall pay to the Administrative Agent on demand an amount equal to the aggregate of the amount of such payment made to the Lender plus the product of (i) the Lender's rate per annum applicable to overnight deposits for amounts approximately equal to the amount paid by the Administrative Agent to such Lender and not so received from the Borrower multiplied by (ii) the amount paid by the Administrative Agent to such Lender and not so received from the Borrower, multiplied by (iii) a fraction, the numerator of which is the number of days that have elapsed from and including the date of payment by the Administrative Agent to the Lender to but excluding the date on which payment is made by such Lender to the Administrative Agent and the denominator of which is 365.

18.04 Erroneous Payments

- Each Lender hereby agrees that if the Administrative Agent notifies a Lender or (1) any other Person who has received funds on behalf of a Lender (any such Lender or Person being a "Payment Recipient") that the Administrative Agent has determined (whether or not after receipt of any notice under Section 18.04(2)) that any funds received by such Payment Recipient from the Administrative Agent or any of its Affiliates were erroneously transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient, whether or not known to such Lender or on its behalf (any such funds, whether received as a payment, prepayment, or repayment of principal, interest, fees, distribution or otherwise being, individually and collectively, an "Erroneous Payment"), and demands the return of such Erroneous Payment (or a portion thereof), such Erroneous Payment shall at all times remain the property of the Administrative Agent and shall be segregated by the Payment Recipient and held in trust for the benefit of the Administrative Agent, and such Lender shall (or, with respect to any other Payment Recipient who received such funds on its behalf, shall cause such Payment Recipient to) promptly, but in no event later than two (2) Business Days thereafter, return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made, in same day funds (in the currency so received), together with interest thereon in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent in same day funds at the greater of (i) the overnight rate for interbank compensation in Canada, and (ii) a rate determined by the Administrative Agent in accordance with Canadian banking industry rules on interbank compensation from time to time in effect. To the extent permitted by Applicable Law, no Lender shall assert any right or claim to an Erroneous Payment, and each Lender hereby waives, any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payment received by such Lender (or any other Payment Recipient who received such funds on its behalf) including, without limitation, waiver of any defense based on "discharge for value" or any similar doctrine. A notice of the Administrative Agent to any Lender or other Payment Recipient under this Section 18.04(1) shall be conclusive, absent manifest error.
- (2) Without limiting Section 18.04(1), each Lender hereby further agrees that if it or any other Payment Recipient acting on its behalf receives a payment, prepayment, or repayment

(whether received as a payment, prepayment, or repayment of principal, interest, fees, distribution or otherwise) from the Administrative Agent (or any of its Affiliates):

- (a) that is in a different amount than, or on a different date from, that specified in a notice of payment, prepayment, or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment, or repayment;
- (b) that was not preceded or accompanied by a notice of payment, prepayment, or repayment sent by the Administrative Agent (or any of its Affiliates); or
- (c) that such Lender otherwise becomes aware was transmitted, or received, in error or by mistake (in whole or in part), in each case:
 - (i) in the case of Sections 18.04(2)(a) or 18.04(2)(b), an error shall be presumed to have been made (absent written confirmation from the Administrative Agent to the contrary) or, in the case of Section 18.04(2)(c), an error has been made, with respect to such payment, prepayment, or repayment; and
 - (ii) such Lender shall (and shall cause any other Payment Recipient that receives funds on its behalf to) promptly (and, in any event, within one (1) Business Day of its knowledge of such error) notify the Administrative Agent of its receipt of such payment, prepayment, or repayment, the details thereof (in reasonable detail) and that it is so notifying the Administrative Agent pursuant to this Section 18.04(2)(c)(ii).
- (3) Each Lender hereby authorizes the Administrative Agent to set off, net, and apply any and all amounts at any time owing to such Lender under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Lender from any source, against any amount due to the Administrative Agent under Section 18.04(1) or under the indemnification provisions of this Agreement.
- (4) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason after demand therefor pursuant to Section 18.04(1) from any Lender that has received such Erroneous Payment (or portion thereof) (or from any other Payment Recipient who received such Erroneous Payment (or portion thereof) on such Lender's behalf) (such unrecovered amount being an "Erroneous Payment Return Deficiency"), upon the Administrative Agent's notice of same to such Lender:
 - (a) such Lender shall be deemed to have assigned its Loans (but not its Commitments) of the Credit Facility in respect of which such Erroneous Payment was made (being the "Erroneous Payment Impacted Credit Facility") in an amount equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Commitments) of the Erroneous Payment Impacted Credit Facility, being the "Erroneous Payment Deficiency Assignment") at par plus any accrued and unpaid interest (with any applicable assignment fee to be waived by the Administrative Agent in such instance), and is hereby (together with the Borrower) deemed to execute and deliver a

- Lender Assignment and Assumption with respect to such Erroneous Payment Deficiency Assignment;
- (b) the Administrative Agent, as the assignee Lender, shall be deemed to acquire the Erroneous Payment Deficiency Assignment; and
- (c) upon such deemed acquisition, the Administrative Agent as the assignee Lender shall become a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment and the assigning Lender shall cease to be a Lender hereunder with respect to such Erroneous Payment Deficiency Assignment, excluding, for the avoidance of doubt, its obligations under the indemnification provisions of this Agreement and its applicable Commitments which shall survive as to such assigning Lender.
- (5) The Administrative Agent may, in its discretion, sell any Loans acquired pursuant to an Erroneous Payment Deficiency Assignment and upon receipt of the proceeds of such sale, the Erroneous Payment Return Deficiency owing by the applicable Lender shall be reduced by the net proceeds of the sale of such Loan (or portion thereof), and the Administrative Agent shall retain all other rights, remedies and claims against such Lender (and/or against any recipient that receives funds on its respective behalf). For the avoidance of doubt, no Erroneous Payment Deficiency Assignment will reduce the Commitments of any Lender and such Commitments shall remain available in accordance with the terms of this Agreement. In addition, each party hereto agrees that, except to the extent that the Administrative Agent has sold a Loan (or portion thereof) acquired pursuant to an Erroneous Payment Deficiency Assignment, and irrespective of whether the Administrative Agent may be equitably subrogated, the Administrative Agent shall be contractually subrogated to all the rights and interests of the applicable Lender under the Loan Documents with respect to each Erroneous Payment Return Deficiency.
- (6) The parties hereto agree that an Erroneous Payment shall not pay, prepay, repay, discharge or otherwise satisfy any Obligations owed by the Borrower or any other Obligor, except, in each case, to the extent such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or any other Obligor for the purpose of making such Erroneous Payment.
- (7) Each party's obligations, agreements and waivers under this Section 18.04 shall survive the resignation or replacement of the Administrative Agent, any transfer of rights or obligations by, or the replacement of, a Lender, the termination of the Commitments, and the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.

ARTICLE 19 – AGENCY

19.01 Appointment and Authority.

Each of the Lenders hereby irrevocably appoints Royal Bank of Canada as the Administrative Agent to act on its behalf as the Administrative Agent hereunder and under the

other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. Without limiting the foregoing, the Administrative Agent shall act as collateral agent for the Lenders pursuant to the Loan Documents. The provisions of this Article are solely for the benefit of the Administrative Agent and the Lenders, and no Obligor shall have rights as a third party beneficiary of any of such provisions.

19.02 Rights as a Lender.

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with any Obligor or any Affiliate thereof as if such Person were not the Administrative Agent and without any duty to account to the Lenders. Except as may be otherwise provided in this Agreement, all communications between any Obligor and any Lender in connection with this Agreement and the other Loan Documents shall be directed through the Administrative Agent. All notices by a Lender to the Administrative Agent shall be through the Administrative Agent's office as set out herein (or as may be amended in accordance with the terms hereof) and all other notices by the Administrative Agent to the Lenders shall be through the Lender's office as set out herein (or as may be amended in accordance with the terms hereof).

19.03 Exculpatory Provisions.

- (1) The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:
 - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Event of Default has occurred and is continuing;
 - (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Majority of the Lenders (or such other number or percentage of the Lenders as shall be expressly provided for in the Loan Documents), but the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or Applicable Laws; and
 - (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained

by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

- (2) The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Majority of the Lenders (or such other number or percentage of the Lenders as is necessary, or as the Administrative Agent believes in good faith is necessary, under the provisions of this Agreement or any other Loan Document) or (ii) in the absence of its own gross negligence or wilful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default or Event of Default unless and until written notice describing such Default or Event of Default is given to the Administrative Agent by the Borrower or a Lender.
- (3) Except as otherwise expressly specified in this Agreement, the Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition specified in this Agreement, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

19.04 Reliance by Administrative Agent.

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, internet or intranet posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of the Drawdown that by its terms must be fulfilled to the satisfaction of a Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent shall have received notice to the contrary from such Lender prior to the making of the Drawdown. The Administrative Agent may consult with legal counsel, independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

19.05 Indemnification of Administrative Agent.

Each Lender agrees to indemnify the Administrative Agent and hold it harmless (to the extent not reimbursed by the Borrower), severally (and not jointly nor jointly and severally) in accordance with its Rateable Portion from and against any and all losses, claims, damages, liabilities and related expenses, including the fees, charges and disbursements of any counsel, which may be incurred by or asserted against the Administrative Agent in any way

relating to or arising out of the Loan Documents or the transactions therein contemplated. However, no Lender shall be liable for any portion of such losses, claims, damages, liabilities and related expenses resulting from the Administrative Agent's gross negligence or wilful misconduct.

19.06 Delegation of Duties.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent from among the Lenders (including the Person serving as Administrative Agent) and their respective Affiliates. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Persons. The provisions of this Article and other provisions of this Agreement for the benefit of the Administrative Agent shall apply to any such sub-agent and to the Related Persons of the Administrative Agent and any such sub-agent, and shall apply to their activities herein as Administrative Agent.

19.07 Replacement of Administrative Agent.

- (1) The Administrative Agent may at any time give thirty (30) days' prior notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Majority of the Lenders shall have the right, in consultation with the Borrower, to appoint a successor, which shall be a Lender having a Commitment and having an office in Toronto, Ontario or an Affiliate of any such Lender with an office in Toronto, Ontario. The Administrative Agent may also be removed at any time by the Majority of the Lenders upon 30 days' prior notice to the Administrative Agent and the Borrower as long as the Majority of the Lenders in consultation with the Borrower, appoint and obtain the acceptance of a successor within such thirty (30) days, which shall be a Lender having an office in Toronto, or an Affiliate of any such Lender with an office in Toronto.
- If no such successor shall have been so appointed by the Majority of the Lenders (2) and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders, appoint a successor Administrative Agent meeting the qualifications specified in Section 19.07, provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of the Security held by the Administrative Agent on behalf of the Lenders under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such Security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender directly, until such time as the Majority of the Lenders appoint a successor Administrative Agent as provided for above in the preceding paragraph.

(3) Upon a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the former Administrative Agent, and the former Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided in the preceding paragraph). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the termination of the service of the former Administrative Agent, the provisions of this Article 19 and of Article 16 one shall continue in effect for the benefit of such former Administrative Agent, its sub-agents and their respective Related Persons in respect of any actions taken or omitted to be taken by any of them while the former Administrative Agent was acting as Administrative Agent.

19.08 Non-Reliance on Administrative Agent and Other Lenders.

Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent, any other Lender or any other Person and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent, any other Lender or any other Person and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

19.09 Collective Action of the Lenders.

Each of the Lenders hereby acknowledges that to the extent permitted by Applicable Laws, the Security and all rights and remedies provided under the Loan Documents to the Lenders are for the benefit of the Lenders collectively and acting together and not severally and further acknowledges that its rights hereunder and under any Security are to be exercised not severally, but by the Administrative Agent upon the decision of the Majority of the Lenders (or other such number or percentage of the Lenders as shall be expressly provided for in the Loan Documents). Accordingly, notwithstanding any of the provisions contained herein or in any Security, each of the Lenders hereby covenants and agrees that it shall not be entitled to take any action hereunder or thereunder including any declaration of default or Event of Default hereunder or thereunder but that any such action shall be taken only by the Administrative Agent with the prior written agreement of the Majority of the Lenders (or other such number or percentage of the Lenders as shall be expressly provided for in the Loan Documents). Each of the Lenders hereby further covenants and agrees that upon any such written agreement being given, it shall co-operate fully with the Administrative Agent to the extent requested by the Administrative Agent. Notwithstanding the foregoing, in the absence of instructions from the Majority of the Lenders and where in the sole opinion of the Administrative Agent, acting reasonably and in good faith, the exigencies of the situation warrant such action, the Administrative Agent may without notice to or consent of the Lenders take such action on behalf of the Lenders as it deems appropriate or desirable in the interest of the Lenders. Subject to the foregoing, if the Borrower fails to perform any covenants on its part herein or under any Loan Document, the Administrative Agent may, in its discretion but need not, perform any covenant capable of being performed by the Administrative Agent and if the covenant requires the payment or expenditure of money, the Administrative Agent on behalf of the Lenders may make such payment or expenditure and all sums so expended shall be forthwith payable by the Borrower to the Administrative Agent on behalf of the Lenders, will be Obligations secured by the Security and shall bear interest at the Prime Rate plus the Prime Rate Margin.

19.10 No Other Duties, etc.

Anything herein to the contrary notwithstanding, none of the bookrunners, arrangers or holders of similar titles, if any, specified in this Agreement shall have any powers, duties or responsibilities under this Agreement or any of the Loan Documents, except in its capacity, as applicable, as the Administrative Agent or a Lender hereunder.

<u>ARTICLE 20 – SUCCESSORS AND ASSIGNS</u>

20.01 Successors and Assigns.

- (1) Successors and Assigns Generally. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that no Obligor may assign or otherwise transfer any of its rights or obligations hereunder (other than as permitted in this Agreement) without the prior written consent of the Administrative Agent and all Lenders and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except in accordance with this Article 20 (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in Subsection 20.01(4) and, to the extent expressly contemplated hereby, the Related Persons of each of the Administrative Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (2) <u>Assignments by Lenders</u>. Upon notice to the Borrower, any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and all outstanding Loans at the time owing to it), provided that:
 - (a) except if an Event of Default has occurred and is continuing or in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the outstanding Loans at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender, the aggregate amount of the Commitment being assigned (which for this purpose includes all outstanding Loans owing to it thereunder) or, if the applicable Commitment is not then in effect, the principal outstanding balance of all outstanding Loans of the assigning Lender subject to each such assignment (determined as of the date the Lender Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Lender Assignment and Assumption, as of the Trade Date) shall not be less than Cdn. \$5,000,000 and in increments of Cdn. \$1,000,000 thereof, unless each of the Administrative Agent and, so long as no Event of Default has

occurred and is continuing, the Borrower otherwise consent to a lower amount (each such consent not to be unreasonably withheld or delayed);

- (b) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the outstanding Loans or the Commitment assigned, except that this clause (b) shall not prohibit any Lender from assigning all or a portion of its rights and obligations among separate credits on a non-pro rata basis;
- (c) any assignment must be approved by the Administrative Agent (such approval to be in the Administrative Agent's sole discretion);
- (d) any assignment must be to an Eligible Assignee;
- (e) the parties to each assignment shall execute and deliver to the Administrative Agent a Lender Assignment and Assumption, together with a processing fee in the following sentence and the Eligible Assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. Each assignment shall be made upon payment by the assignor to the Administrative Agent of a processing fee of \$5,000 (provided, in each case, that no such fee shall be payable in respect of any assignment (i) to any Affiliate of such Lender; or (ii) by the Administrative Agent or any of its Affiliates), provided that, in the case of contemporaneous assignments by a Lender to more than one fund managed by the same investment advisor or an Affiliate of such investment advisor (which funds are not then Lenders hereunder), only a single such fee shall be payable for all such contemporaneous assignments.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 20.01(3), from and after the effective date specified in each Lender Assignment and Assumption, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Lender Assignment and Assumption, have the rights and obligations of a Lender under this Agreement and the other Loan Documents, including any collateral security, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Lender Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of a Lender Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Article 13 and Article 16, and shall continue to be liable for any breach of this Agreement by such Lender, with respect to facts and circumstances occurring prior to the effective date of such assignment. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 20.01(4). Any payment by an assignee to an assigning Lender in connection with an assignment or transfer shall not be or be deemed to be a repayment by the Borrower or a new Loan to the Borrower.

- (3) Register. The Administrative Agent shall maintain at one of its offices in Toronto, Ontario a copy of each Lender Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the outstanding Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, absent manifest error, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (4) <u>Participations</u>. Any Lender may at any time sell participations to any Person (other than a natural person, an Obligor or any Affiliate of an Obligor) (each, a "**Participant**") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the outstanding Loans owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, and (iii) the Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any payment by a Participant to a Lender in connection with a sale of a participation shall not be or be deemed to be a repayment by the Borrower or a new Loan to the Borrower.

Subject to Subsection 20.01(5), the Borrower agrees that each Participant shall be entitled to the benefits of Article 13 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to paragraph (2) of this Section, provided such Participants agree to be subject to Article 17 as though they were Lenders. To the extent permitted by Applicable Laws, each Participant also shall be entitled to the benefits of Section 14.01 as though it were a Lender, provided such Participant agrees to be subject to Article 17 as though it were a Lender.

- (5) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under Section 13.01 and Section 13.02 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent.
- (6) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, but no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

ARTICLE 21 - AMENDMENTS AND WAIVERS

21.01 Amendments and Waivers

(1) Subject to Subsections 21.01(1), 21.01(2) and 21.01(3), (a) no amendment shall be effective unless in writing between the Administrative Agent and the Borrower and approved by the Majority of the Lenders and (b) no waiver of any provision of any of the

Loan Documents nor consent to any departure by the Borrower or any other Person from such provisions shall be effective unless in writing and approved by the Majority of the Lenders. Any amendment, waiver or consent shall be effective only in the specific instance and for the specific purpose for which it was given.

- Only written amendments, waivers or consents signed by all the Lenders shall (i) increase a Lender's Commitment or subject any Lender to any additional obligation; (ii) reduce the principal or amount of, or interest on, directly or indirectly, any outstanding Loans or any fees; (iii) postpone any date fixed for any payment of principal of, or interest on, any outstanding Loans or any fees; (iv) change the percentage of the Commitments or the number or percentage of Lenders required for the Lenders, or any of them, or the Administrative Agent to take any action; (v) permit any termination or release of any of the Obligations or Obligors hereunder or the Security or the collateral subject to the Security not otherwise permitted to be released hereunder; (vi) change the definition of "Majority of the Lenders"; (vii) change any other provision of this Agreement which specifically requires the consent of all of the Lenders; and (viii) amend this Section 21.01(2).
- (3) Only written amendments, waivers or consents signed by the Administrative Agent and approved in writing by the Majority of the Lenders, shall affect the rights or duties of the Administrative Agent under the Loan Documents.

21.02 Judgment Currency.

- (1) If, for the purposes of obtaining judgment in any court, it is necessary to convert a sum due to a Lender in any currency (the "Original Currency") into another currency (the "Other Currency"), the parties agree, to the fullest extent that they may effectively do so, that the rate of exchange used shall be that at which, in accordance with normal banking procedures, such Lender could purchase the Original Currency with the Other Currency on the Business Day preceding the day on which final judgment is given or, if permitted by Applicable Laws, on the day on which the judgment is paid or satisfied.
- (2) The obligations of the Borrower in respect of any sum due in the Original Currency from it to any Lender under any of the Loan Documents shall, notwithstanding any judgment in any Other Currency, be discharged only to the extent that on the Business Day following receipt by the Lender of any sum adjudged to be so due in the Other Currency, the Lender may, in accordance with normal banking procedures, purchase the Original Currency with such Other Currency. If the amount of the Original Currency so purchased is less than the sum originally due to the Lender in the Original Currency, the Borrower agrees, as a separate obligation and notwithstanding the judgment, to indemnify the Lender, against any loss, and, if the amount of the Original Currency so purchased exceeds the sum originally due to the Lender in the Original Currency, the Lender shall remit such excess to the Borrower.

ARTICLE 22-GOVERNING LAW; JURISDICTION; ETC.

22.01 Governing Law and Submission to Jurisdiction

- (1) <u>Governing Law</u> This Agreement shall be governed by, and construed in accordance with, the laws of the Province of Ontario and the laws of Canada applicable in that Province.
- Submission to Jurisdiction Each Obligor irrevocably and unconditionally submits, for itself and its property, to the nonexclusive jurisdiction of the courts of the Province of Ontario, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement or any other Loan Document, or for recognition or enforcement of any judgment, and each of the parties hereto irrevocably and unconditionally agrees that all claims in respect of any such action or proceeding may be heard and determined in such court. Each of the parties hereto agrees that a final, non-appealable judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement or in any other Loan Document shall affect any right that the Administrative Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against each Obligor or its properties in the courts of any jurisdiction.
- (3) <u>Waiver of Venue</u> Each Obligor irrevocably and unconditionally waives, to the fullest extent permitted by Applicable Laws, any objection that it may now or hereafter have to the laying of venue of any action or proceeding arising out of or relating to this Agreement or any other Loan Document in any court of the Province of Ontario. Each of the parties hereto hereby irrevocably waives, to the fullest extent permitted by Applicable Laws, the defence of an inconvenient forum to the maintenance of such action or proceeding in any such court.

ARTICLE 23 - WAIVER OF JURY TRIAL

23.01 Waiver of Jury Trial

EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAWS, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

<u>ARTICLE 24 - NOTICES: EFFECTIVENESS;</u> <u>ELECTRONIC COMMUNICATION</u>

24.01 Notices, etc.

- (1) <u>Notices Generally</u> Except in the case of notices and other communications expressly permitted to be given by telephone or email (and except as provided in Sections 24.01(3) or (4)), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopier to the addresses or telecopier numbers specified elsewhere in this Agreement or, if to a Lender, to it at its address or telecopier number specified in this Agreement.
- (2) <u>Delivery</u> Notices sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices sent by telecopier shall be deemed to have been given when sent (except that, if not given on a Business Day between 9:00 a.m. and 5:00 p.m. local time where the recipient is located, shall be deemed to have been given at 9:00 a.m. on the next Business Day for the recipient). Notices delivered through electronic communications to the extent provided in Sections 24.01(3) or (4) below, shall be effective as provided therein.
- (3) <u>Electronic Communications</u> Notices and other communications to the Administrative Agent hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Lenders. The Lenders or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, provided that approval of such procedures may be limited to particular notices or communications.
- (4) <u>Delivery by Electronic Communication</u> Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), provided that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next Business Day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.
- (5) <u>Notice</u> The mailing addresses and address for electronic communications for the purposes of notices and other communications to the Obligors, the Administrative Agent and the Lenders are set out on the signature pages of this Agreement.
- (6) <u>Change of Address, Etc.</u> Any party hereto may change its address or telecopier number for notices and other communications hereunder by notice to the other parties hereto in accordance with the terms of this Agreement.

<u>ARTICLE 25 - COUNTERPARTS; INTEGRATION;</u> EFFECTIVENESS; ELECTRONIC EXECUTION

25.01 Counterparts; Integration; Effectiveness; Electronic Execution

- (1) <u>Counterparts; Integration; Effectiveness</u> This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement shall become effective when it has been executed by the Administrative Agent and when the Administrative Agent has received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by telecopy or by sending a scanned copy by electronic mail shall be effective as delivery of a manually executed counterpart of this Agreement.
- (2) <u>Electronic Execution of Assignments</u>. The words "execution," "signed," "signature," and words of like import in any Lender Assignment and Assumption shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Laws, including Parts 2 and 3 of the *Personal Information Protection and Electronic Documents Act* (Canada), the *Electronic Commerce Act*, 2000 (Ontario) and other similar federal or provincial laws based on the *Uniform Electronic Commerce Act*, as the case may be.

<u>ARTICLE 26 - TREATMENT OF</u> CERTAIN INFORMATION: CONFIDENTIALITY

26.01 Treatment of Certain Information: Confidentiality

(1) Each of the Administrative Agent and the Lenders agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and its Affiliates' respective partners, directors, officers, employees, agents, advisors and representatives (to the extent necessary to administer or enforce this Agreement and the other Loan Documents) (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and will be bound and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority having jurisdiction over it (including any self-regulatory authority), (c) to the extent required by Applicable Laws or other legal process, (d) to any other party hereto, (e) to the extent reasonable, in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or (ii) any actual or prospective counterparty (or its advisors) to any swap, derivative, credit-linked note or similar transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (x) becomes publicly

available other than as a result of a breach of this Section or (y) becomes available to the Administrative Agent or any Lender on a non-confidential basis from a source other than an Obligor.

- (2) For purposes of this Section, "Information" means all information relating to the Borrower or any of its Affiliates or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a non-confidential basis prior to such receipt. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information. In addition, the Administrative Agent may disclose to any agency or organization that assigns standard identification numbers to loan facilities such basic information describing the facilities provided hereunder as is necessary to assign unique identifiers (and, if requested, supply a copy of this Agreement), it being understood that the Person to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to make available to the public only such Information as such person normally makes available in the course of its business of assigning identification numbers.
- (3) In addition, and notwithstanding anything herein to the contrary, the Administrative Agent may provide information concerning the Borrower and the Credit Facility established herein to recognized trade publishers of information for general circulation in the loan market.

ARTICLE 27 - GENERAL

27.01 Addresses, Etc. for Notices

The mailing addresses and addresses for electronic communications for the purposes of notices and other communications to the Obligors, the Administrative Agent and the Lenders are set out on the signature pages of this Agreement.

27.02 Effect of Assignment

For greater certainty, an assignment by any Lender, in accordance with the terms hereof, of its rights hereunder will not constitute a repayment, discharge, rescission, extinguishment or novation of any extension of credit by such Lender under this Agreement or interest therein, and the obligations so assigned will continue to be the same obligations and not new obligations.

27.03 Benefit of the Agreement

This Agreement shall enure to the benefit of and be binding upon the Borrower, each other Obligor, the Administrative Agent and each Lender and their respective successors and assigns.

27.04 No Other Duties, Etc.

Anything herein to the contrary notwithstanding, none of the bookrunners, arrangers or holders of similar titles, if any, specified in this Agreement shall have any powers, duties or responsibilities under any of the Loan Documents, except in its capacity, as applicable, as a Lender hereunder.

27.05 Survival

The provisions of Article 15 and Article 16 will survive the repayment of all Loans, whether on account of principal, interest or fees, and the termination of this Agreement, unless a specific release of such provisions by the Lenders is delivered to Borrower. Notwithstanding the foregoing, the Obligors shall be released from the environmental indemnity in Article 15 following the termination of the Loan Documents and payments of the Obligations hereunder unless the Administrative Agent was a mortgagee in possession of the Property or has commenced to enforce its rights under the Loan Documents following an Event of Default.

27.06 <u>Severability and Conflict</u>

If any provision of this Agreement is determined by any court of competent jurisdiction to be illegal or unenforceable, that provision will be severed from this Agreement and the remaining provisions will continue in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any of the parties. In the event of any conflict or inconsistency between this Agreement and any other Loan Document, the provisions of this Agreement shall prevail, to the extent of any such conflict or inconsistency (provided, for greater certainty, that the existence of additional rights or remedies in favour of the Administrative Agent or a Lender in such other Loan Documents shall not constitute a conflict or inconsistency).

27.07 Further Assurances

Each Obligor and the Administrative Agent will promptly cure any default by it in the execution and delivery of this Agreement, the Loan Documents or of any of the agreements provided for hereunder to which it is a party. Each Obligor, at its expense, will promptly execute and deliver to the Administrative Agent or a Lender, upon request by the Administrative Agent or a Lender, all such other and further documents, agreements, opinions, certificates and instruments in compliance with, or for the accomplishment of the covenants and agreements of such Obligor hereunder or more fully to state the obligations of such Obligor as set forth herein or to make any recording, file any notice or obtain any consent, all as may be reasonably necessary or appropriate in connection therewith.

Time of the Essence

Time shall be of the essence of the Loan Documents.

[Signature pages follow]

S-1 of 3

IN WITNESS WHEREOF the parties have executed this Agreement.

BORROWER:	HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, in its capacity as general partner for and on behalf of HBC YSS1 LIMITED PARTNERSHIP, as Borrower
401 Bay Street, Suite 2302 Toronto, ON, M5H 2Y4	By: Name: Title:
Attention: SVP, Real Estate E-Mail:	By: Name: Title: We have the authority to bind the above.
NOMINEE:	HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, as Nominee
401 Bay Street, Suite 2302 Toronto, ON, M5H 2Y4	By: Name: Title:
Attention: SVP, Real Estate E-Mail:	By: Name: Title: We have the authority to bind the above

GUARANTOR:	RIOCAN REAL ESTATE INVESTMENT TRUST as Guarantor		
2300 Yonge Street, Suite 500, PO Box 2386 Toronto, ON M4P 1E4	By:Name:Title:		
Attention: E-Mail:	By:		

S-3 of 3

ROYAL BANK OF CANADA ADMINISTRATIVE AGENT: as Administrative Agent Royal Bank of Canada By: Agency Services Group Name: 155 Wellington Street West, 8th Floor Title: Toronto, Ontario, M5V 3H1 By: Attention: Drake Guo Name: Email: drake.guo@rbccm.com Title: I/We have the authority to bind the above. **LENDERS:** ROYAL BANK OF CANADA, as a Lender Royal Bank of Canada By: Global Loan Administration Name: 155 Wellington Street West, 8th Floor Title: Toronto, Ontario, M5V 3H1 By: Attention: Jason Viaene Name: E-Mail: Jason. Viaene@rbccm.com Title:

I/We have the authority to bind the above.

Schedule A Lenders and Commitments

Lender and Lending Office	Credit Facility Commitment
Royal Bank of Canada Global Loan Administration 155 Wellington Street West, 8th Floor Toronto, Ontario M5V 3H1 Attention: Jason Viaene E-Mail: Jason.Viaene@rbccm.com	Cdn. \$75,000,000.00
Total	Cdn. \$75,000,000.00

Schedule B

Legal Description of Lands

[NTD: Fogler to advise.]

Schedule C

Certificate re. Identity

- TO: ROYAL BANK OF CANADA, as Administrative Agent and the Lenders, together with their successors and assigns
- RE: Credit agreement dated January 26, 2024 among, *inter alios*, HBC YSS1 Limited Partnership (the "Borrower"), Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI, as nominee, RioCan Real Estate Investment Trust, as guarantor, and Royal Bank of Canada (the "Administrative Agent") as administrative agent for the lenders party thereto from time to time (the "Lenders"), as amended from time to time (the "Credit Agreement")

All terms not otherwise defined herein shall have the meanings ascribed thereto in the Credit Agreement.

In the matter of the Credit Facility established by the Credit Agreement and Loan Documents and other documentation attendant thereto, I, ●, a lawyer and partner of ●, have agreed to act on behalf of the Administrative Agent in order that the Administrative Agent and the Lenders may comply with their identification obligations under federal *Proceeds of Crime (Money Laundering) and Terrorist Financing Regulations*.

license of •, an officer of •. The	nd attest that I personally referred to the original driver's e reference number of the above identity document is
The	e place of issuance of the above identity document is
Ontario. •'s date of birth is	. A true copy of this identity document,
both front and back, is affixed hereto a	
an officer of •. The referen	test that I personally referred to the original passport of •, ace number of the above identity document is e place of issuance of the above identity document is
	ation pages of this identity document is affixed hereto at
CERTIFIED AND DATED as of	, 2024.
	Name: •

EXHIBIT A

True copy of identity documents

See attached.

Schedule 1.01(A)

Compliance Certificate

TO: ROYAL BANK OF CANADA, as Administrative Agent and the Lenders, together with their successors and assigns

RE: Credit agreement dated January 26, 2024 among, inter alios, HBC YSS1 Limited Partnership (the "Borrower"), Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI, as nominee, RioCan Real Estate Investment Trust, as guarantor, and Royal Bank of Canada (the "Administrative Agent") as administrative agent for the lenders party thereto from time to time (the "Lenders"), as amended from time to time (the "Credit Agreement")

FROM: RIOCAN REAL ESTATE INVESTMENT TRUST (the "Guarantor")

DATE: •

This Compliance Certificate is delivered to you pursuant to Section 9.02(3) of the Credit Agreement. All terms used in this Compliance Certificate that are defined in the Credit Agreement have the same meanings herein.

I, *[name]*, the *[title]* of the Guarantor, certify for and on behalf of the Guarantor, and not in my personal capacity and without personal liability, that:

- 1. Examinations. I have made or caused to be made such examinations or investigations as are, in my opinion, necessary to furnish this Compliance Certificate and I have furnished this Compliance Certificate with the intent that it may be relied upon by the Administrative Agent and the Lenders as a basis for determining compliance by the Obligors with their respective covenants and obligations under the Credit Agreement. I have read and am familiar with the Credit Agreement including, in particular, the definitions of the various financial terms used in the Credit Agreement, the representations and warranties and covenants contained in Article 8 and Article 9 and the Events of Default described in Article 12 of the Credit Agreement.
- 2. Representations and Warranties All of the representations and warranties of the Obligors contained in Section 8.01 of the Credit Agreement are true and correct on and as of the date hereof as though made on and as of the date hereof, subject to changes thereto:
 - (a) given to the Administrative Agent by the Obligors and accepted in writing by the Administrative Agent, and
 - (b) expressly contemplated by the terms of the Credit Agreement and disclosed to the Administrative Agent in writing.

- 3. <u>Terms, Covenants and Conditions</u> All of the terms, covenants and conditions of the Credit Agreement and each of the other Loan Documents to be performed or complied with by the Obligors at or prior to the date hereof have been performed or complied with.
- 4. <u>Default</u> No Default or Event of Default has occurred and is continuing on the date hereof.
- 5. <u>Financial Statements</u> Attached hereto are the financial statements of the most recent date referred to in Section [9.02(2)/9.02(1)] of the Credit Agreement (such date being the last day of each Fiscal Quarter or the last day of each Fiscal Year, as applicable, being the "Reference Date"). Such financial statements when read with the notes thereto and the reconciliation accompanying such financial statements, represents fairly the consolidated financial position of the Guarantor as of the date of such statements and for the reporting period included in such statements, and such financial statements, and all calculations of financial covenants and financial ratios and presentation of financial information in this Compliance Certificate and the Appendices hereto have been prepared in accordance with GAAP.

6. <u>Financial Covenant Compliance</u>

(a) Total Indebtedness Ratio

The Guarantor is in compliance with the maximum Total Indebtedness Ratio requirements of Section 9.05(a) of Credit Agreement as demonstrated in the calculations set forth in Appendix B to this Compliance Certificate. The ratio of Consolidated Indebtedness to Consolidated Gross Book Value as of the Reference Date is ____, which does not exceed 60% percent.

(b) Secured Indebtedness Ratio

The Guarantor is in compliance with the maximum Secured Indebtedness Ratio requirements of Section 9.05(b) of Credit Agreement as demonstrated in the calculations set forth in Appendix C to this Compliance Certificate. The ratio of Consolidated Secured Indebtedness to Consolidated Gross Book Value as of the Reference Date is , which does not exceed 40% percent.

(c) Debt Service Coverage Ratio

The Guarantor is in compliance with the minimum Debt Service Coverage Ratio requirements of Section 9.05(c) of the Credit Agreement as demonstrated in the calculations set forth in Appendix D to this Compliance Certificate. The ratio of Consolidated EBITDA to Debt Service as of the Reference Date is ______, which is not less than 1.50x.

(d) <u>Minimum Unitholders' Equity</u>

The Guarantor is in compliance with the Minimum Unitholders' Equity requirements of Section 9.05(d) of the Credit Agreement as demonstrated in the

	calculations set forth in Appendix E to this Compliance Certificate. Unitholders' Equity as of the Reference Date is, which is not less than the sum of (i) Cdn. \$5,000,000,000 and (ii) 75% of the net proceeds received in connection with any equity offerings made by the Guarantor after the Closing Date.
(e)	Unencumbered Property Assets Ratio
	The Guarantor is in compliance with the maximum Unencumbered Property Assets Value Ratio requirements of Section 9.05(e) of Credit Agreement as demonstrated in the calculations set forth in Appendix F to this Compliance Certificate. The ratio of Unencumbered Property Assets Value to Consolidated Unsecured Indebtedness as of the Reference Date is, which does not exceed 1.50x.
DATED o	day of, 202●.
	Name:
	Title:

Schedule 1.01(B)

Rollover Notice

TO: ROYAL BANK OF CANADA, as Administrative Agtogether with their successors and assigns RE: Credit agreement dated January 26, 2024 among, in Limited Partnership (the "Borrower"), Hudson's Compagnie De La Baie D'Hudson SRI, as nominee, Investment Trust, as guarantor, and Royal Bat "Administrative Agent") as administrative agent for the from time to time (the "Lenders"), as amended from time Agreement")		ROYAL BANK OF CANADA, as Administrative Agent and the Lenders, together with their successors and assigns			
		Bay Company ULC e, RioCan Real Estate ank of Canada (the ne lenders party thereto			
FROM	И:	HBC YSS1 LIMITED PARTN	ERSHIP	(the "Borrowe	er")
DATI	Ε:	•			
1.	Agree	Rollover Notice is delivered to ement. All terms used in this ement have the same meanings here	Rollover		. /
2.	The E	Borrower hereby requests a Rollov	er as follo	ws:	
	(a)T	Type and amount of each Loan to b	oe rolled o	ver, and matur	ity date:
				Amount	Maturity Date
	(i)	Term CORRA Loan:	Cdn. \$		
	(ii)	Daily Compounded CORRA Loan:	Cdn.		
	(iii)	Term SOFR Loan:	U.S. \$	_	
	(b) T	Type, amount and new maturity da	te of each	Loan resulting	from Rollover:
				Amount	Maturity Date
	(i)	Term CORRA Loan:	Cdn. \$		
	(ii)	Daily Compounded CORRA Loan:	Cdn.		

(iii)	Term SOFR Loan:	U.S. \$	

- 3. Representations and Warranties All of the representations and warranties of the Obligors contained in Section 8.01 of the Credit Agreement are true and correct in all material respects on and as of the date hereof as though made on and as of the date hereof, subject to changes thereto:
 - (a) given to the Administrative Agent by the Borrower and accepted in writing by the Administrative Agent; and
 - (b) expressly contemplated by the terms of the Credit Agreement and disclosed to the Administrative Agent in writing.
- 4. All of the conditions precedent to the Advance requested hereby that have not been properly waived in writing by or on behalf of the Administrative Agent have been satisfied.
- 5. No Default or Event of Default has occurred and is continuing or will have occurred and be continuing on the Drawdown Date, or will result from the Loan requested hereby.

HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, in its capacity as general partner for and on behalf of HBC YSS1 LIMITED PARTNERSHIP

By:	
	Name:
	Title:
By:	
	Name:
	Title:
I/We	have the authority to bind the above.

Schedule 1.01(C)

Conversion Notice

то:		ROYAL BANK OF CANADA, as Administrative Agent and the Lenders, together with their successors and assigns		
RE:		Credit agreement dated January 26, 2024 among, inter alios, HBC YSS1 Limited Partnership (the "Borrower"), Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI, as nominee, RioCan Real Estate Investment Trust, as guarantor, and Royal Bank of Canada (the "Administrative Agent") as administrative agent for the lenders party thereto from time to time (the "Lenders"), as amended from time to time (the "Credit Agreement")		
FROM	/I:	HBC YSS1 LIMITED PARTNE	RSHIP ((the "Borrower")
DATE	E:	•		
1.	Agree		nversion	rsuant to Section 2.04(1) of the Credit Notice that are defined in the Credit
2.	The B	orrower hereby requests a Convers	ion as fol	llows:
(a) Type and amount of each Loan to be converted:				
				Amount
	(i)	Prime Rate Loan:	Cdn. \$	
	(ii)	Term CORRA Loan:	Cdn. \$	
	(iii)	Daily Compounded CORRA Loan:	Cdn. \$	
	(iv)	Term SOFR Loan:	U.S. \$	
	(b) T	ype and amount of each Loan resul	ting fron	n Conversion:
				Amount
	(i)	Prime Rate Loan:	Cdn. \$	

	(ii)	Term CORRA Loan:	Cdn. \$
	(iii)	Daily Compounded CORRA Loan:	A Cdn. \$
	(iv)	Term SOFR Loan:	Cdn. \$
	Loan,		le to such CORRA Loan or Term SOFR I the date on which the Interest Period is
3.	contair	ned in Section 8.01 of the Cred	the representations and warranties of the Obligors lit Agreement are true and correct in all material though made on and as of the date hereof, subject
	(a)	given to the Administrative Age Administrative Agent; and	ent by the Borrower and accepted in writing by the
	(b)	expressly contemplated by the t Administrative Agent in writing	terms of the Credit Agreement and disclosed to the g.
4.		ly waived in writing by or on	e Advance requested hereby that have not been behalf of the Administrative Agent have been
5.			curred and is continuing or will have occurred and or will result from the Loan requested hereby.
			HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, in its capacity as general partner for and on behalf of HBC YSS1 LIMITED PARTNERSHIP
			By:
			Name: Title:
			By:
			Name: Title:

I/We have the authority to bind the above.

Schedule 1.01(D)

Drawdown Notice

TO: ROYAL BANK OF CANADA, as Administrative Agent and the Lenders, together with their successors and assigns

RE: Credit agreement dated January 26, 2024 among, inter alios, HBC YSS1 Limited Partnership (the "Borrower"), Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI, as nominee, RioCan Real Estate Investment Trust, as guarantor, and Royal Bank of Canada (the "Administrative Agent") as administrative agent for the lenders party thereto from time to time (the "Lenders"), as amended from time to time (the "Credit Agreement")

FROM: HBC YSS1 LIMITED PARTNERSHIP (the "Borrower")

DATE: •

- 1. This Drawdown Notice is delivered to you pursuant to Section 2.04 of the Credit Agreement. All terms used in this Drawdown Notice that are defined in the Credit Agreement have the same meanings herein.
- 2. The Borrower hereby requests the following Loan:
 - (a) Drawdown Date: •
 - (b) Amount of Loan

Prime Rate Loan:

Cdn.

\$

- (c) Payment Instructions: In accordance with Direction re Funds dated as of the date hereof and delivered to the Administrative Agent.
- 3. <u>Representations and Warranties</u> All of the representations and warranties of the Obligors contained in Section 8.01 of the Credit Agreement are true and correct in all material respects on and as of the date hereof as though made on and as of the date hereof, subject to changes thereto:
 - (a) given to the Administrative Agent by the Borrower and accepted in writing by the Administrative Agent; and

- (b) expressly contemplated by the terms of the Credit Agreement and disclosed to the Administrative Agent in writing.
- 4. All of the conditions precedent to the Advance requested hereby that have not been properly waived in writing by or on behalf of the Administrative Agent have been satisfied.
- 5. No Default or Event of Default has occurred and is continuing or will have occurred and be continuing on the Drawdown Date, or will result from the Loan requested hereby.

HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, in its capacity as general partner for and on behalf of HBC YSS1 LIMITED PARTNERSHIP

By:	
	Name:
	Title:
By:	
	Name:
	Title:
I/We	have the authority to bind the above.

Schedule 1.01(E)

Material Licences and Material Property Agreements

Material Property Agreements

- 1. Store Ground Lease.
- 2. Store Sublease.
- 3. Limited Partnership Agreement.

Material Licenses

Nil.

Schedule 1.01(F)

Additional Permitted Encumbrances

1. [NTD: Fogler to advise.]

Schedule 1.01(I)

Repayment Notice

TO:	ROYAL BANK OF CANADA, as Administrative Agent and the Lenders,
	together with their successors and assigns

RE: Credit agreement dated January 26, 2024 among, inter alios, HBC YSS1 Limited Partnership (the "Borrower"), Hudson's Bay Company ULC Compagnie De La Baie D'Hudson SRI, as nominee, RioCan Real Estate Investment Trust, as guarantor, and Royal Bank of Canada (the "Administrative Agent") as administrative agent for the lenders party thereto from time to time (the "Lenders"), as amended from time to time (the "Credit Agreement")

FROM: HBC YSS1 LIMITED PARTNERSHIP (the "Borrower")

DATE: •

- 1. This Repayment Notice is delivered to you pursuant to Section 6.04 of the Credit Agreement. All terms used in this Repayment Notice that are defined in the Credit Agreement have the same meanings herein.
- 2. The Borrower hereby gives you notice that it intends to repay Cdn. \$● under the Credit Facility on ●.
- 3. The amount of such repayment will, subject to the provisions of the Credit Agreement, be used to repay Loans of the following type:

<u>Loan Type</u> <u>Amount</u>

•

HUDSON'S BAY COMPANY ULC COMPAGNIE DE LA BAIE D'HUDSON SRI, in its capacity as general partner for and on behalf of HBC YSS1 LIMITED PARTNERSHIP

By:		
	Name:	
	Title:	
By:		
	Name:	
	Title:	

I/We have the authority to bind the above.

Schedule 8.01(14)

Ownership Structure

[NTD: Borrower to provide.]

Schedule D

Applicable Margins

Applicable Margins:

Applicable Margins are based on the External Credit Ratings. Currently, only DBRS and S&P provide an External Credit Rating for the Guarantor. Should the Guarantor at any time have two or more ratings, the higher of the ratings shall be applied for the purposes of the Applicable Margins, unless the External Credit Ratings are more than a one-notch differential, in which case the rating which is one tier below the higher of the ratings will be taken for the purposes of the Applicable Margins.

DBRS Rating	CORRA / Term SOFR Margin (bps)	Prime Rate Margin (bps)
≥ A (L)	100	0
BBB (H)	120	20
≤ BBB	145	45

Schedule E

Benchmark Replacement Provisions (SOFR)

- (1) Notwithstanding anything to the contrary herein or in any other Loan Document, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior to any setting of the then-current Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (a) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document and (y) if a Benchmark Replacement is determined in accordance with clause (b) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of any Benchmark setting at or after 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Borrower.
- (2) The Administrative Agent will promptly notify the Borrower of (i) the implementation of any Benchmark Replacement, and (ii) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. The Administrative Agent will promptly notify the Borrower of the removal or reinstatement of any tenor of a Benchmark pursuant to Section (3) of this Schedule E. Any determination, decision or election that may be made by the Administrative Agent pursuant to this Schedule E, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Schedule E.
- (3) Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its discretion, acting reasonably, or (B) the administrator of such Benchmark or the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative or in compliance with or aligned with the International Organization of Securities Commissions (IOSCO) Principles for Financial Benchmarks (the "IOSCO Principles"), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable, non-representative, non-compliant or non-aligned tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service

for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative or in compliance with or aligned with the IOSCO Principles for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

- (4) Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any request for an Conversion to or Rollover of, a Term SOFR Loan to be made, converted or rolled over during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for Conversion to Prime Rate Loans in the Equivalent Amount in Canadian Dollars of the amount so requested.
- (5) In connection with the use, administration, adoption or implementation of a Benchmark Replacement, the Administrative Agent will have the right to make such Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (6) Solely as used in this Schedule E:

"Available Tenor" means, as of any date of determination and with respect to the then current Benchmark, as applicable, any tenor for such Benchmark or payment period for interest calculated with reference to such Benchmark, as applicable, that is or may be used for determining the length of an Interest Period pursuant to this Agreement as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then removed from the definition of "Interest Period" pursuant to Section (3) of this Schedule E.

"Benchmark" means, initially, Adjusted Term SOFR; <u>provided</u> that if a Benchmark Transition Event has occurred with respect to Adjusted Term SOFR or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section (1) of this Schedule E.

"Benchmark Replacement" means, for any Available Tenor, the sum of: (a) the alternate benchmark rate that has been selected by the Lender and the Borrower as the replacement for the then-current Benchmark for the applicable Corresponding Tenor giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for U.S. Dollar-denominated syndicated credit facilities at such time in the United States, and (b) the related Benchmark Replacement Adjustment. If the Benchmark Replacement as determined pursuant to the above would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the thencurrent Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to (i) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body, and/or (ii) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for U.S. Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means, with respect to any Benchmark, the earliest to occur of the following events with respect to the then-current Benchmark:

- (A) in the case of clause (A) or (B) of the definition of "Benchmark Transition Event," the later of (a) the date of the public statement or publication of information referenced therein, and (b) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (B) in the case of clause (C) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by or on behalf of the administrator of such Benchmark (or such component thereof) or the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative or non-compliant with or non-aligned with the IOSCO Principles; <u>provided</u>, that such non-representativeness, non-compliance or non-alignment will be determined by reference to the most recent statement or publication referenced in such clause (C) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date;

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (A) or (B) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

- "Benchmark Transition Event" means, with respect to any Benchmark, the occurrence of one or more of the following events with respect to such then-current Benchmark:
- (A) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely, <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);

- (B) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Board of Governors of the Federal Reserve System, the Federal Reserve Bank of New York, the Term SOFR Administrator, the SOFR Administrator, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), in each case, which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (C) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) or the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are no longer, or as of a specified future date will no longer be, representative or in compliance with or aligned with the IOSCO Principles.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means the period (if any) (x) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with this Schedule E, and (y) ending at the time that a Benchmark Replacement has replaced such then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with this Schedule E.

"Conforming Changes" means, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Business Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of this Schedule E and other technical, administrative or operational matters) that the Administrative Agent decides, acting reasonably, may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of

administration as the Administrative Agent decides, acting reasonably, is necessary in connection with the administration of this Agreement and the other Loan Documents).

"Corresponding Tenor" means, with respect to any Available Tenor, as applicable, either a tenor (including overnight) or an interest payment period, having approximately the same length (disregarding business day adjustment) as such Available Tenor.

"Relevant Governmental Body" means the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or any successor thereto.

"Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.